



AIRBORNE RADAR APPROACH FLIGHT

Report No. FAA-RD-80-60/

TEST EVALUATING VARIOUS

TRACK ORIENTATION TECHNIQUES

L.D. King





June 1980

Final Report

Document is available to the U.S. public through the National Technical Information Service, Springfield, Virginia 22161.

Prepared for

U.S. DEPARTMENT OF TRANSPORTATION FEDERAL AVIATION ADMINISTRATION Systems Research & Development Service Washington, D.C. 20590

80 3 29

079

0 08842 \sim AD

DC FILE COPY

NOTICE

•

This document is disseminated under the sponsorship of the Department of Transportation in the interest of information exchange. The United States Government assumes no liability for its contents or use thereof.

1.000

Technical Report Documentation Page 3. Recipient's Catalog No. 1. Report No 2. Government Accession No. FAA-RD-80-60 tD-A088 426 Title and Subtitle 5. Report Doin June 1980 Airborne Radar Approach Flight Test Evaluating 6. Performing Organization Code Various Track Orientation Techniques . B. Performing Organization Report No. -Author's) 1111. Larry D./King **1**) 10. Work Unit No. (TRAIS) Performing Organization Name and Address Systems Control, Inc. (Vt) 1 87244 11. Contract or Grant No. 1801 Page Mill Road DOT-FA79WA-4293 Palo Alto, California 94304 13. Type of Report and Period Covered 12. Spansoring Agency Name and Address Department of Transportation Final Report. Federal Aviation Administration 14. Sponsoring Agency Code Systems Research and Development Service Washington, D.C. 20591 15. Supplementary Notes Champlain Technology Industries, A Division of Systems Control, Inc. (Vt) A subsidiary of Systems Control, Palo Alto, California 16. Abstract This comprehensive report presents the results of a flight test experiment of an Airborne Radar Approach (ARA) System utilizing various track orientation techniques and operational modes. The tests were performed in the immediate area of NAFEC in Atlantic City, N.J. The test environment involved the airport terminal area and offshore sites. The test aircraft was a NASA CH53A helicopter manufactured by Sikorsky Aircraft and currently based at NAFEC. The test period was from January 1979 to February 1979 and from June 1979 to August 1979. Flight tests for ARA accuracy and procedures development were performed in six distinct operational modes. These were as follows: beacon w/cursor, multiple beacon, skin paint, skin paint w/cursor, combined and beacon-only modes. The specific program objectives can be summarized as follows: 1) to evaluate the ability of the radar operator to guide an aircraft along a predetermined path using various track orientation techniques, namely: the cursor and multiple beacon techniques; 2) to assist the FAA in developing and certifying standard ARA procedures and weather minimums; 3) to define and quantify specific ARA system functions and characteristics for use in a Minimum Operational Performance Standards (MOPS) document. The primary conclusions of this flight test experiment were: the ARA system performed satisfactorily from both an accuracy and an operational viewpoint in all six operational modes, the ARA performance utilizing the track orientation techniques showed marked decrease in the overall Total System Cross Track (TSCT) and Flight Technical Error (FTE) quantities; the cursor and multiple beacon techniques also eliminated the tendency to "home" to station; in the skin paint mode distinguishing offshore targets such as lighthouses from ships and other surface objects is virtually impossible. 18. Distribution Statement 17. Key Words Airborne Radar Approach (ARA) This document is available to the public Beacon w/cursor through the National Technical Information Service, Springfield, Virginia 22161 Skin Paint Multiple Beacon MOPS 21. No. of Puges | 22. Price 20. Security Classif. (of this page) 19. Security Classif. (of this report) 181 Unclassified Unclassified Form DOT F 1700.7 (8-72) Reproduction of completed page authorized

	Approximate Con	iversions to Metric	Measures		9 9	£7		Approximate Conve	ersions from Metr	ic Measures	
Symbel	When You Know	Muitiply by	To Find	Symbol	1 8 1.1.1.1.1.1.1 4.0000009393 64:	22 112	Symbol	When You Knew	Multiply by LENGTH	Ta find	Symbol
		LENGTH			. l.ı. ata	50		ļ		1	
	1				 	61	a ES	multimeters Centemeters	0.04	arches	5 9
ş	.n.hes	5° č.	contractors.	Ë	. J.		É	meters	3.3	feet	E #
r	feet	30	Centimeters	E.	1910 '1' 7	មរ	ε.	meters	1.1	×1115	Ŗ
¥ 6	rards	சு ம ப் ச	meters	ε.	1.) 1.)		Ĺ	No 10 meters	0.b	TILES	Ê
		0	K - : Ometer S	L.	.].1 . 	²¹			ÅRFA		
	1	AREA			h. 10141	91		ļ	71177		
, e		u u		~		, . 	cm,	square centimeters	0.16	Square inchris	٦ų.
1	Saudra andres	C.0	square centimeters	۲		s1	بر	square meters	1.2	SQuare arts	۲đ
Ĵ.	Square vards	5.8 9.5	square meters	Ŀ~Ŀ	1. 1.	•	, Lu	square kilometers	9.4	Square - 1es	~ŧ
4	Square miles	2.6	Square hilometers	ν." 1	; 	+I	Ьà	hectares -10,000 m ⁻	2 5	SáuJP	
	ACLES	6.4	heclares	Ъđ	.1.	įŧ					
	~	AASS (weight)) 5	1		-	MASS (weight)	1	
	s a Julia El S	2			1" ' 1" '	7 1	σ	grams	0.035	ounces	20
		90	S.19	σ.	5) '	· 1	, Pa	k lograms	2.2	spunda	ŧ
0	chort tons		tompes	ç.	ن ال ال	ł	·	tonnes 1000 kg.	1.1	short tons	
	2.000 · 5			-	* * * *	0					
	1	VOLUME			 11	1 ·			VOLUME		
					ין 1141ק יןי 	6				1	
2.1	succus et	۰. ۱	r	ł	: /// 1	1	Ē	multiliters	0 03	fiuld ounces	1 02
1040	tablespoons frind familier	n .	T. Fulters	ī,	; [] []			liters liters	2.1	punts	E 8
:	SQUE	52 0.24	1, 2011					i there	52 C		÷ 8
ñ	D.ats	0 47	litters		: (111) (111)	L	۳.	cubic meters	35	cubic feet	i~=
5	quart >	ũ.95	2) thes		11	•	,	Cubic meters	1 3	cubic yards	, b
V.D.	10-11-01	3.8	· ters	- '	₩ '	•					
. 1	1941 D.CL.	5570 24 0	Cubic meters	 L	t: ' 2	١ç		TEMI	PERATIRE (avar)	-	
:										-1	
	TEMF	ERATURE (exact)			. 1 .	•	c	C etsus terrestature	9 5 then add 32	f shrenheit terration	-
u	f greeter	1 i i i i	(j.e.151-42)		- 11 11	£		in the sector of	37 600		
	، وستر المهدية: ١١ م	subtracting 32	terperature		· -> -	. 1		23	98 £	ц. к.	
					ж, Ч		4	· ·	8C 25	50 20C	
ę		•				۱ 	4	╋╸╋╺╌╸┝╸┍┶ ╕	20 1 20		ω
					11 1' 	•••1	°		37		

METRIC CONVERSION FACTORS

ï,

and the second

٠

y

					Aangertion For		٦
			TABLE	OF CONTENTS		 	•
Table List List	of Co of Fig of Tab	ntents jures les			A		i iv vii
SECTI	ON					- · · · · · · · · · · · · · · · · · · ·	PAGE
1.0	EXECU	TIVE SU	MMARY				1-1
	1.1 1.2 1.3 1.4	BRIEF EQUIP SUMMAR METHOD RESULT	DESCRIPTION O MENT UTILIZED Y OF TEST OBJ OF APPROACH S AND CONCLUS	F TESTS PERFO ECTIVES IONS	ORMED AND		1-1 1-1 1-2 1-6
2.0	INTRO	DUCTION					2-1
	2.1 2.2 2.3 2.4	BACKGR ARA TE PURPOS ORGANI	OUND ST CONCEPTS E OF THE TEST ZATION OF THE	S REPORT			2-1 2-3 2-4 2-5
3.0	DETAI	LED TEC	HNICAL OBJECT	IVES			3-1
4.0	DESCR	IPTION	OF THE ARA EX	PERIMENT DESI	IGN		4-1
	4.1	EQUIPM	ENT SUMMARY				4-1
		4.1.1 4.1.2 4.1.3 4.1.4	The Bendix R The RCA Prim The Transpon Flight Test I	DR-1400A Rada us-50 Radar der Beacons Helicopter	ar		4-1 4-7 4-10 4-11
	4.2	TEST P	ROFILES AND P	ROCEDURES			4-11
		4.2.1 4.2.2 4.2.3	General Pilo Bendix RDR-14 RCA Primus-50	t Procedures 400A Multiple 0 Combined ar	e Beacon Testin nd Beacon-Only	9	4-15 4-17
		4.2.4	Mode Testin Bendix RDR-14	g 400A Skin Pai	int and Skin Pa	int	4-19
		4.2.5	W/Cursor Tes Bendix RDR-14	sting 400A Sinale F	Beacon With Cur	sor	4-21
			Testing		ula este an on our		4-23
	4.3 4.4	SUBJEC DATA A	T PILOT EXPERI CQUISITION AND	IENCE AND TRA D PROCLSSING	AINING		4-27 4-28

i

A.

Sect	ion		TABLE OF CONTENTS - Continued	Page
		4.4.1 4.4.2 4.4.3 4.4.4 4.4.5 4.4.6	Airborne Instrumentation Ground Reference Data Manually Recorded Flight Logs Data Processing Data Processing Facilities Data Digitizing	4-28 4-28 4-29 4-29 4-32 4-34
	4.5	DATA A	NALYSIS PROCEDURES	4-34
		4.5.1	Navigation Error Analysis	4-34
5.0	ANAL	SIS OF	RESULTS	5-1
	5.1	AIRBOR	NE RADAR AS AN APPROACH AID	5-1
		5.1.1 5.1.2 5.1.3	Pilot/Copilot Procedures and Workload Overall Accuracy Assessment ATC Integration	5-2 5-3 5-10
	5.2	ANALYS	IS OF PILOT PROCEDURES	5-13
		5.2.1 5.2.2	Pilot/Copilot Workload Pilot/Copilot Blunders	5-13 5-19
	5.3	Detail	ed Accuracy Data	5-19
		5.3.1 5.3.2 5.3.3 5.3.4 5.3.5	Offshore Site: Beacon-Only and Skin Paint Comparison Offshore Site: Performance Enhancement in the Skin Paint Mode Using the Cursor Technique Airport Site: Performance in The Beacon Mode Using the Cursor Technique Airport Site: Performance Enhancement in The Beacon Mode Using the Multiple Beacon Technique Offshore Site: Primus-50 Combined Mode Versus the Beacon-Only Mode	5-20 5-35 5-46 5-66 5-77
	5.4	OPERAT	IONAL EVALUATION OF THE ARA CONCEPT	5-94
		5.4.1	Landside ATC Integration 5.4.1.1 Airport Terminal Area	5-95 5-96
		5.4.2	Offshore ATC Procedures	5-105
6.0	SUMMA	RY OF A	RA PERFORMANCE	6-1
	6.1	TECHN1	CAL PERFORMANCE	6-1
		6.1.1 6.1.2 6.1.3	Range Performance Bearing Accuracy Display Readability Display Resolution	6-1 6-2 6-2 6-2

ii

TABLE OF CONTENTS - Continued

Sect	tion			Page
	6.2	OPERAT	IONAL PERFORMANCE	6-3
		6.2.1	Beacon/Ground Clutter Discrimination	6-3
		6.2.2	Cursor Interpretation	6-3 6-4
	l	6.2.4	Performance in The Skin Paint and Skin Paint W/Cursor Modes	6-4
		6.2.5	Performance in The Single Beacon With Cursor Mode	6-4
	(5.2.6	Performance in The Multiple Beacon Mode RCA Primus-50 Combined and Beacon-Only	6-4
			Mode Performance	6-5
7.0	CONCLUS	SIONS		7-1
REFE	RENCES			R-1
BIBL	IOGRAPHY			B-1

LIST OF FIGURES

Figur	<u>e</u>	Page
1.1	Overall ARA Flight Test Program Summary	1-4
4.1	Bendix RDR-1400A Front Panel Bendix RDR-1400A Front Panel With Cursor Display	4-2 4-5
4.3	Heading Cursor Technique (Off-Course)	4-6
4.4	Heading Cursor Technique (On-Course)	4-6
4.5	RCA Primus-50 Front Panel	4-8
4.6	The Transponder Beacon	4-12
4.7	NASA Sikorsky CH53A Test Helicopter	4-13
4.8	Airborne Radar Approach Geometries	4-18
4.9	Bendix RDR-1400A Multiple Beacon Direct Straight Procedures	4-20
4.10	RCA Primus-50 Offshore Direct Straight Procedures	4-22
4.11	Bendix RDR-1400A Offshore Direct Straight Procedures	4-24
4.12	Profile 1: Direct Straight Airport Site	4-25
4.13	Profile 2: Overhead Straight Airport Site	4-26
4.14	Time Correlation Correction Plots	4-33
5.1	ARA Offshore Skin Paint Mode Approach Total System Cross	
	Track Error	5-27
5.2	ARA Offshore Site TSCT and FTE Plot: Direct Straight	
	Approach Flown 17 July 1979	5-29
5.3	ARA Offshore Site TSCT and FTE Plot: Direct Straight	
_ .	Approach Flown 16 July 1979	5-30
5.4	ARA Offshore Single Beacon Approach Total System Cross	ر د ما
- -	TRACK Error	5-31
5.5	ARA UTTSNORE SKIN Paint Mode Flight Technical Error	5-32
5.0	ARA OTTShore Single Bedcon Mode Filght Technical Error	5-33
5.7	ARA ULTShore Skill Paint House Alrourne System Error	5-36
5.0	ARA ULTSHUPE SHIGLE DEdCON NODE AIRDORNE System LITON Pondiy DDP 14004 Offebana Sita Skin Daint Mode Histograms	5-30
5.9	Rendix RDR 1400A Offshore Site Skin Paint Mode Histograms	5-38
5.10	APA Offehana Skin Paint With Curson Mode Total System Cross	5-50
5.11	Track Error	5-43
5 12	ARA Offshore Skin Paint With Cursor Mode Elight Technical	0 10
J.12	Frror	5-44
5.13	ARA Offshore Site ISCI and FIE Plot: Direct Straight	
5.10	Approach Flown 27 July 1979	5-45
5.14	ARA Offshore Skin Paint With Cursor Airborne System Error	5-47
5.15	Bendix RDR-1400A Skin Paint With Cursor Mode Histograms	5-48
5.16	ARA Airport Single Beacon With Cursor Mode Total System	
	Cross Track Error	5-57
5.17	Initial Segment of An Overhead Straight Approach Flown on	
	3 November 1978	5-58
5.18	Intentional Offset Approach Flown on 6 August 1979	5-59
5.19	ARA Airport Single Beacon Approach Total System Cross	
	Track Error	5-61
5.20	ARA Airport Single Beacon With Cursor Approach Flight	~
	Technical Error	5-62

LIST OF FIGURES - Continued

Figur	re	Page
5.21 5.22	ARA Single Beacon Approach Flight Technical Error ARA Airport Single Beacon With Cursor Approach Airborne	5-63
5.23 5.24 5.25 5.26	ARA Airport Single Beacon Approach Airborne System Error Bendix RDR-1400A Beacon With Cursor Mode Histograms Bendix RDR-1400A Airport Site Single Beacon Mode Histograms ARA Airport Multiple Beacon Mode Total System Cross Track	5-64 5-65 5-67 5-68
5.27 5.28 5.29 5.30	Error ARA Airport Multiple Beacon Mode Flight Technical Error ARA Airport Multiple Beacon Mode Airborne System Error Bendix RDR-1400A Multiple Beacon Mode Histograms ARA Offshore RCA Primus-50 Combined Mode Total System	5-74 5-75 5-76 5-78
5.31	Cross Track Error ARA Offshore RCA Primus-50 Combined Mode Flight Technical	5-82
5.32	Error ARA Offshore RCA Primus-50 Combined Mode Airborne System	5-84
5.33	Error ARA Offshore RCA Primus-50 Beacon-Only Mode Total System	5-85
5.34	ARA Offshore RCA Primus-50 Beacon-Only Mode Flight	5-89
5.35	ARA Offshore RCA Primus-50 Beacon-Only Mode Airborne System Error	5-90
5.36 5.37	RCA Primus-50 Combined Mode Histograms RCA Primus-50 Beacon-Only Mode Histograms	5-92 5-93
5.38	Clearance Airspace ARA Airport Single Beacon With Cursor Approaches Related	5-98
5.40	to Obstacle Clearance Airspace ARA Airport Multiple Beacon Approaches Related to	5-99
5.41	Obstacle Clearance Airspace Airport Site Single Beacon Mode: Mean, Mean Plus or Minus	5-100
5.42	2-Sigma Values Bendix RDR-1400A Beacon With Cursor: Mean, Mean Plus or	5-102
5.43	Minus 2-Sigma Values Bendix RDR-1400A Multiple Beacon: Mean, Mean Plus or Minus 2 Sigma Values	5-103
5.44	ARA Offshore Single Beacon Approaches Related to Obstacle Clearance Airspace	5-104
5.45	ARA Offshore Skin Paint Approaches Related to Obstacles Clearance Airspace	5-107
5.46	ARA Offshore Skin Paint With Cursor Approaches Related to Obstacle Clearance Airspace	5-108
5.47	ARA Uttshore Primus-50 Combined Mode Approaches Related to Obstacle Clearance Airspace ARA Offshore Primus 50 Reacon_Only Approaches Related to	5-109
5.48 5.40	Obstacle Clearance Airspace Bendix RDR-1400A Offshore Site Single Beacon Mode: Mean.	5-110
J. 72	Mean Plus or Minus 2-Sigma Values	5-112

i

Samler Little Martin

LIST OF FIGURES - Continued

Figur	<u>e</u>	Page
5.50	Bendix RDR-1400A Skin Paint Mode: Mean, Mean Plus or Minus 2-Sigma Values	5-113
5.51	Bendix RDR-1400A Skin Paint With Cursor: Mean, Mean Plus or Minus 2-Sigma Values	5-114
5.52	RCA Primus-50 Combined Mode: Mean, Plus or Minus 2-Sigma Values	5-115
5.53	RCA Primus-50 Beacon-Only Mode: Mean, Mean Plus or Minus 2-Sigma Values	5-116

LIST OF TABLES

Tabl	<u>e</u>	Page
4.1 4.2 4.3	Bendix RDR-1400A Multiple Beacon Flight Test Matrix RCA Primus-50 Flight Test Matrix Bendix RDR-1400A Skin Paint and Skin Paint W/Cursor	4-14 4-14
4 4	Flight Test Matrix Bendix RDR-14004 Single Beacon W/Cursor Elight	4-15
	Test Matrix	4-16
4.5 4.6	Flight Experience of Subject Pilots in Hours Data Processing and Analysis Outline	4-27 4-35
5.1	Overall Bendix RDR-1400A Offshore ARA Performance	
5.2	Summary ARA Offshore TSCT Linear and Angular Errors as a	5-4
5.3	ARA Offshore FTE Linear and Angular Errors as a	5-5
5.4	Overall Bendix RDR-1400A Airport ARA Performance	5-6
5.5	ARA Airport TSCT Linear and Angular Errors as a	5-7
5.6	ARA Airport FTE Linear and Angular Errors as a Function	5-8
r 7	of Along Track Distance	5-8
5.7 5.8	ARA Offshore RCA Primus-50 Linear and Angular Errors as	5-9
59	a runction of Along frack Distance Pilot/Copilot Workload Ratings	5-10
5.10	NAFEC ARA Bendix RDR-1400A Skin Paint Offshore Approaches	2-10
c 11	Error Analysis Log and Statistical Summary	5-21
5.11	NAFEC ARA Bendix RDR-1400A Beacon Mode Offshore Approaches	r 00
5.12	Bendix RDR-1400A Skin Paint Mode One Nautical Mile	5-22
	Aggregate Data	5-24
5.13	Bendix RDR-1400A Offshore Site Beacon Mode One	
5 14	NAUTICAL MILE Aggregate Data NAFFC ARA Bendix RDR-1400A Skin Paint With Curson Offshore	5-25
••••	Approaches Error Analysis Log and Statistical Summary	5-39
5.15	Bendix RDR-1400A Skin Paint With Cursor One Nautical Mile	
5 16	Aggregate Data NAFEC ADA Bondiy DDD 1400A Boncon With Currey Aignort	5-41
5.10	Approaches Error Analysis Log and Statistical Summary	5-49
5.17	NAFEC ARA Bendix RDR-1400A Beacon Mode Airport Approaches	5 15
5 10	Error Analysis Log and Statistical Summary	5-51
5.18	Bendix RDR-1400A Beacon With Cursor One Nautical Mile	5 5 2
5.19	Bendix RDR-1400A Airport Site Beacon Mode One Nautical Mile	2-23
	Aggregate Data	5-54
5.20	Beacon With Cursor Letdown Error Quantities	5-56
5.21	NAFEC ARA Bendix RDR-1400A Multiple Beacon Airport	_
	Approaches Error Analysis Log and Statistical Summary	5-69

LIST OF TABLES - Continued

<u>Table</u>		Page
5.22	Bendix RDR-1400A Multiple Beacon One Nautical Mile	
	Aggregate Data	5-71
5.23	Multiple Beacon Letdown Error Quantities	5-72
5.24	NAFEC ARA RCA Primus-50 Combined Mode Offshore Approaches	
	Error Analysis Log and Statistical Summary	5-79
5.25	RCA Primus-50 Combined Mode One Nautical Mile	
	Aggregate Data	5-81
5.26	NAFEC ARA RCA Primus-5 Beacon-Only Mode Offshore Approaches	
	Error Analysis Log and Statistical Summary	5-86
5.27	RCA Primus-50 Beacon-Only Mode One Nautical Mile	
	Aggregate Data	5-87
5.28	RCA Primus-50 Beacon Return Area Statistics	5-117
6.1	Airborne Radar Approach System Bearing Accuracy Data	6-2

viii

EXECUTIVE SUMMARY

An overview of the analysis of the performance of airborne radar as an approach aid utilizing various track orientation techniques and operational modes is presented in this executive summary. In order to attain the proper perspective and to understand the impact of these results it is necessary to briefly review the experiment design, the equipment used and the test objectives. This section begins with a review of these important issues and then summarizes the overall method of approach used to answer operational, functional and accuracy questions.

1.1 BRIEF DESCRIPTION OF TESTS PERFORMED AND EQUIPMENT UTILIZED

The Airborne Radar Approach (ARA) System Flight Test Experiment was initiated by the Systems Research and Development Service of the Federal Aviation Administration (FAA). The flight tests were performed by the Approach and Landing Branch of the FAA's National Aviation Facilities Experimental Center (NAFEC). The tests were supported by Systems Control, Inc. (Vt.) in the areas of test planning, data collection/reduction and final report preparation.

The ARA tests were performed utilizing the Bendix RDR-1400A and RCA Primus-50 radar systems. The test vehicle was a CH53A helicopter manufactured by Sikorsky Aircraft and currently based at NAFEC. Four test pilots were provided by NAFEC as subjects for this experiment. The ARA test flights were performed in the general area of NAFEC. Test airspace environments included both the airport terminal area and offshore sites. Flight tests for ARA accuracy and procedures development were performed in the beacon with cursor, multiple beacon, skin paint, skin paint with cursor, combined and beacon-only modes. The testing period was from January 1979 to February 1979 and from June 1979 to August 1979.

1.2 SUMMARY OF TEST OBJECTIVES

The ARA flight test experiment was designed to obtained both quantitative and qualitative data in the areas of system accuracy, ARA procedures, ARA functional requirements and ATC operational integration problems.

1.0

Specific program objectives can be summarized as follows:

- To evaluate the ability of the radar operator to guide an aircraft along a predetermined path using various track orientation techniques, namely: the cursor and multiple beacon techniques.
- To assist the FAA and the user community in developing and certifying standard ARA procedures, associated weather minimums and obstacle clearance requirements.
- To define and quantify specific ARA system functions and characteristics for use in a Minimum Operational Performance Standards document.

1.3 METHOD OF APPROACH

The basic ARA test program consisted of approach testing in six major test configurations. These were defined as, Beacon with Cursor Tests. Multiple Beacon Tests, Skin Paint Tests, Skin Paint with Cursor Tests, Combined Mode Tests and Beacon-only Tests. Twenty-five flights with 88 approaches were flown in the Airborne Radar approach experiment. Of these twenty-five flights, four were performed using the modified Bendix radar system in the beacon with cursor mode, ten flights were performed in the multiple beacon mode, six were accomplished in the skin paint and skin paint with cursor modes, and the remaining five flights utilized the RCA Primus-50 radar system in the combined and beacon-only modes.

The single beacon with cursor tests were conducted at the airport site utilizing the modified Bendix RDR-1400A Radar system. The modified radar system generated an additional azimuth line or cursor on the radar screen. The cursor, which represents course error, represents the intended approach course selectable on the Horizontal Situation Indicator (HSI). Two different types of approaches (direct and overhead straight) were conducted during this phase of testing so as to provide a reasonable data base in order that cursor aided approaches could be compared to non-cursor aided approaches. The multiple beacon testing concentrated on using two independent ground beacons to establish visual reference indications of the desired final approach course on the airborne radar display. The skin paint and skin paint with cursor portion of the testing utilized the Bendix radar system. The effort in this portion of the testing was to use a prominent surface object (Brandywine Lighthouse) to simulate Airborne Radar Approaches to offshore sites with and without the aid of a cursor. The RCA Primus-50 testing was conducted at the offshore site in the combined and beacon-only modes. The purpose of these tests was to determine the operational viability of the combined beacon/ground mapping mode in a offshore environment. Quantitative data was obtained for all of the above mentioned test areas.

Four subject pilots* were used for the Airborne Radar Approach flight test program. All four pilots alternated as pilot and copilot during the test program. The subject pilots had previous experience flying ARA approaches. This experience was obtained from the single beacon approach testing conducted at NAFEC during the test period from October 1978 to December 1978 (Reference 1). In any case the proficiency in using the airborne radar came about through actual operational use.

There were two pilots per crew, with the copilot being the only crew member hooded. For safety reasons, the pilot was not hooded, but was instructed to fly only those course headings indicated by the copilot It was also the pilot's responsibility to handle all communications.

Two separate testing sites were used for the ARA experiment. These were: airport testing conducted in the NAFEC terminal area and offshore testing conducted in Delaware Bay using Brandywine Lighthouse. In addition, two different types of approaches were conducted at the airport site (direct straight and overhead straight) while only one approach type (direct straight) was utilized at the offshore site. Pilot procedures and profiles were generated in advance in order to insure a well disciplined test environment.

Figure 1.1 summarizes the overall ARA flight test program. This figure shows that a total of eighty-eight approaches were flown: 14 in

<u>/NOTE</u>/ *Due to the retirement of one subject pilot after the multiple beacon testing the three remaining pilots completed the flight test program.



Figure 1.1 Overall ARA Flight Test Program Summary

n

The second s

1-4

the beacon with cursor mode, 39 in the multiple beacon mode*, 9 in the skin paint mode, 9 in the skin paint with cursor mode, 11 in the combined mode and 6 in the beacon-only mode. The dashed portion of Figure 1.1 provides the breakdown of flight testing conducted in the single beacon mode (Reference 1). These data were included in this report so that comparisons could be made with the cursor aided, multiple beacon and skin paint approaches. Figure 1.1 provides the detailed breakdown of the number and type of approaches flown in each specific mode of operation. In addition, Figure 1.1 shows the experiment balance between interexperiment variables for each mode (e.g. the balance between skin paint and skin paint with cursor approaches). The final point of interest presented in the figure is the number and type of each approach flown in the airport terminal and offshore areas. As shown in the figure, three approach procedures (direct straight, overhead straight and overhead offset) were utilized during the previous single beacon testing, while only two approach procedures (direct straight and overhead straight) were utilized during the track orientation technique testing. In addition, a fourth procedure (intentional offset) was utilized at the airport site in the beacon with cursor mode. The intentional offset procedure was used to intentionally misplace the aircraft off course at a position the aircraft had acquired in previous single beacon testing. This procedure was utilized to determine the ability of the radar operator to acquire and track the intended course using the cursor technique.

The direct straight was used when the winds were favorable for landing upwind in the approach direction and obstruction clearances permitted descent to minimums. The overhead straight was used when the winds favored an opposite direction when approaching the landing site. The procedures used for flying both profiles were identical, with one exception. On the overhead approaches it was necessary for the copilot to call out when directly overhead the target so that the pilot could start an outbound timing of 25 minutes. At the end of 25 minutes a

/NOTE/ *Twenty-four approaches were non-recoverable from the film interpretation standpoint because the beacon return was splayed across the entire azimuth of the screen. This splaying occured because the Bendix Radar's STC circuit was misadjusted. procedure turn was executed and the Intermediate Approach Course was acquired. The accuracy with which each of these approach procedures was executed was determined in detail. These results are presented and discussed in-depth in Section 5.3 for each of the test areas: airport and offshore. The dicussion in Section 5.3 also presents an in-depth look at the various operational modes and track orientation techniques tested.

A more detailed description of specific flight profiles, the equipment used, the test procedures, the subject pilots and the data collection/reduction/analysis procedures is presented in Sections 4.1, 4.2, 4.3, 4.4 and 4.5.

1.4 RESULTS AND CONCLUSIONS

This section presents a generalized summary or overview of the primary results and conclusions which may be found in Sections 5.0, 6.0 and 7.0. A detailed analysis and expanded discussion of each of the major test areas and operational modes indicated in Figure 1.1 is contained in Section 5.0. A comprehensive discussion of pilot/copilot procedural operations using ARA is contained in Section 5.2. Section 6.0 is structured to present the data applicable to specific MOPS questions raised by the Radio Technical Commission for Aeronautics (RTCA) Special Committee (SC-133) on ARA. Finally, Section 7.0 presents a more detailed and expanded discussion of the qualitative conclusions regarding ARA as a non-precision approach technique utilizing various track orientation techniques and operational modes. The following subsection will provide a summary of results and conclusions obtained during the flight test program.

- The modified Bendix RDR-1400A Airborne Radar Approach System tested performed satisfactorily from both an accuracy and an operational viewpoint in the single beacon.with cursor mode.
 - The error quantity used to measure this performance was Total System Cross Track (TSCT) error. A comparison of summary data between the single beacon and single beacon with cursor approaches over the length of the approach courses for the airport environment is as follows:

TSCT Operational mode $\pm 1\sigma$ 1.36 nm Single Beacon .57 nm Single Beacon W/Cursor The results indicated a large reduction in the TSCT and FTE values for the cursor-aided approaches. The FTE values determined during flight test at the airport site were as follows. FTE **Operational Mode** tlσ 1.44 nm Single Beacon Single Beacon W/Cursor .76 nm The ARA lateral track keeping accuracy (TSCT) was well within specified obstacle clearance airspace limits, ±4 nm established by the RTCA SC-133 MOPS. The cross track Airborne System Error (ASE) and the Along Track Error (ATE) for the Bendix radar system were consistently small and showed no dependency on distance from the target. 2) ARA performance in the multiple beacon mode showed basically the same results obtained in the beacon with cursor mode. That is, the track orientation afforded by the multiple beacon mode reduced the overall TSCT and FTE values for the approaches flown at the airport site. These values were as follows: TSCT FTE **Operational Mode** tlσ t]σ 1.36 nm 1.44 nm Single Beacon .60 nm .70 nm Multiple Beacon 3) ARA performance was quantified in both the skin paint mode and the skin paint with cursor mode approach testing conducted at the offshore site. The results indicated three distinct items.

If the target is positively identified, ARA approaches in the skin paint mode can be executed as accurately as single beacon approaches. The following TSCT and FTE results support this fact. TSCT FTE Operational Mode ±lσ ±lσ .89 nm 1.02 nm Single Beacon .34 nm .64 nm Skin Paint

- Offshore targets such as ships provide bright returns but are not distinguishable from the intended target (an oil rig).
- The cursor-aided approaches flown in the skin paint mode indicate an improvement in the FTE quantities. These FTE quantities are improved because of the track guidance offered by the cursor. The TSCT and FTE values were as follows:

TSCT FTE Operational Mode 10 +10 .34 nm .64 nm Skin Paint .57 nm .48 nm Skin Paint W/Cursor

4) Technically and operationally the RCA Primus-50 performed well in the combined and beacon-only modes. System performance results (ASE and ATE) indicate that the system provided accurate guidance to the target. The TSCT and FTE values calculated were as follows:

> TSCT FTE Operational Mode +1σ +1.: .40 nm .64 nm Combined .42 nm .69 nm Beacon-only

- Operationally the combined mode offers one serious problem. Because of the large beacon return displayed, surface objects in the immediate area surrounding the intended target are blocked out, therefore offering limited obstacle clearance information.
- 5) ARA System functions and characteristics were investigated to satisfy the requirements of RTCA's SC-133. These were in the areas of technical performance and operational performance. A brief summary of the details discussed in Section 6.0 is provided in the following list.

TECHNICAL PERFORMANCE

- A. RANGE PERFORMANCE
 - Single Beacon 21 nm at 1000 feet altitude with the beacon at ground level.
 - 35 nm at 1000 feet altitude with the beacon at 30 feet above water level.
 - (2) Skin Paint 20 nm at 1000 feet with relatively low background clutter. (Brandywine lighthouse was utilized as the target.)

B. BEARING ACCURACY

- (1) Bendix RDR-1400A At 5 nm from the target one-Radar System sigma numbers range in value from ±3.1° to ±5.4°. At 10 nm the one-sigma values were determined to be ±2.2° to ±3.8°.
- (2) RCA Primus-50 At 5 nm and 10 nm from the target the one-sigma values were determined to be $\pm 2.0^{\circ}$ and $\pm 3.5^{\circ}$ respectively.
- C. DISPLAY READABILITY Not a specific test variable. Qualitative observations indicated that the readability was adequate except in direct sunlight.
- D. DISPLAY RESOLUTION Not a specific test variable. Observations and calculations showed the radar displays tested had adequate resolution. In addition, the displayed size of the beacon return did not adversly affect the pilot's ability to conduct the approach.

OPERATIONAL PERFORMANCE

- A. BEACON/GROUND CLUTTER DISCRIMINATION The combined beacon/ground mapping mode was found to be operationally feasible, but because of the large displayed beacon return obstacles within the immediate area of the intended target were blocked out.
- B. OFFSHORE TARGET DISCRIMINATION It was discovered during the offshore skin paint testing that while executing approaches to the lighthouse it was often times difficult to distinguish between ships and the lighthouse.
- C. CURSOR INTERPRETATION The use of an electronically generated course direction cursor, which obtains its inputs directly from HSI course selection proved very effective. Results indicate a marked reduction in the TSCT and FTE quantities.
- D. PERFORMANCE IN THE SKIN PAINT AND SKIN PAINT W/CURSOR MODES - The only serious problem encountered was that of positive target identification.
- E. PERFORMANCE IN THE SINGLE BEACON W/CURSOR MODE Results indicate that the tendency to home to the station is eliminated with the cursor technique. Also, the cursor decreases the level of mental workload required to fly the approach.
- F. PERFORMANCE IN THE MULTIPLE BEACON MODE Results indicate that by utilizing two longitudinally separated beacons track guidance is improved. It was also discovered that the level of mental workload is increased because the track angle error is not directly displayed on the radar.
- G. RCA PRIMUS-50 COMBINED AND BEACON-ONLY MODE PERFORMANCE -Operationally the RCA Primus-50 performed well in both modes of operation. The only major problem encountered was the large displayed beacon return (See A).

INTRODUCTION

This report was prepared to summarize the results of the Airborne Radar Approach (ARA) flight tests. These tests were performed by Approach and Landing Branch (ANA 110) of the Federal Aviation Administration's National Aviation Facilities Experimental Center (NAFEC). The tests were supported by Systems Control, Inc. (Vt.) [SCI (Vt.)] in the areas of test planning, data collection/reduction and final report preparation. The test vehicle was a CH53A helicopter manufactured by Sikorsky Aircraft. Flight tests for ARA accuracy and procedures development were performed in two distinct operational environments. These were the airport environment and the offshore environment. The airport ARA tests were performed at NAFEC and the offshore tests were conducted nearby in Delaware Bay. The testing period was from January 1979 to February 1979 and from June 1979 to August 1979. The primary reasons for the ARA flight test program were: 1) to evaluate the ARA concept both quantitatively and qualitatively in the areas of accuracy and flight procedures involving various track orientation techniques and modes of operation; 2) to provide empirical inputs for the "Minimum Operational Performance Standard" being generated by the Radio Technical Commission for Aeronautics (RTCA) Special Committee 133 (SC-133).

2.1 BACKGROUND

Continued expansion in the application of helicopters to the accomplishment of civilian oriented tasks depends to a significant extent on the capabilities of the aircroft and the navigation systems in order to operate in all weather conditions. Much of the future growth of the helicopter market will be in applications that involve the transport of people in such areas as offshore oil support, in corporate transport, and eventually in scheduled transportation. The effectiveness of the helicopter in these missions depends on its ability to circumvent the time delays of other modes of transportation. If weather results in a significant number of cancellations and delays, the helicopter's effectiveness is lost. A particularly interesting facet of this general problem area arises from the needs of helicopter operators to fly in adverse weather in remote areas. This type of mission generates a

2.0

requirement for a self-contained helicopter instrument approach system for landing on oil rigs and other landing areas remote from conventional navigational aids. Such a system would also benefit the corporate operator who desires insturment approach minimums equivalent to conventional non-precision approach procedures at a variety of sites, many of which may be of an ad hoc nature, but who would be unable to afford the time and expense necessary to achieve the installation of ground navigation aids.

Weather radar used in the mapping mode for IFR approaches offers a possible immediate low-cost solution. The application of airborne weather/mapping radar as an approach and landing aid has generally become known throughout the industry as an Airborne Radar Approach (ARA) System. This terminology will be used frequently in this document.

The major impetus for the ARA operational application has come from the Helicopter Association of America (HAA) in general, and its offshore energy exploration support members in particular. In addition to the basic requirements of the HAA to stimulate the development of helicopter IFR procedures and systems, particularly at sites where instrument approach procedures are unavailable, the necessity to provide approach capability to offshore oil rigs under Instrument Meteorological Conditions (IMC) is critical to their mission. The HAA, therefore, has consistently requested the FAA to develop standard operational procedures and equipment certification criteria as regards ARA systems and their operation in the IFR portion of the National Airspace System as one means of providing instrument approach and landing capability. Certain offshore helicopter operators have been granted approval for ARA approaches on a singular basis, but no general certification criteria currently exists within the FAA.

In recognition of the emerging need for some measure of equipment performance criteria, early in 1977 the Radio Technical Commision for Aeronautics (RTCA) constituted a Special Committee (SC-133) for the purpose of developing a Minimum Operational Performance Standards (MOPS) for ARA systems for helicopters. This MOPS document will contain both operational and technical performance criteria which might ultimately be



used for FAA certification purposes. At least two requests have been made of the FAA by RTCA, in behalf of SC-133, which contains a postulated operational scenario and initial technical performance specifications. However, no substantive technical data was previously available on many of the critical issues concerning the ARA system application.

2.2 ARA TEST CONCEPTS

Many test programs are limited by the practical considerations of time and money. In the case of the ARA testing these were additional considerations which further limited the investigation. It is therefore necessary to identify in some detail what this ARA evaluation does and does not cover.

Most of the following discussion concerning the ARA flight test limitation emanate from considerations of the aircraft used as the test vehicle. The Sikorsky CH53A has sufficient passenger and payload capacity for experimental test purposes, however, due to limited fuel capacity, it has an effective flight endurance of approximately 1-3/4 hours. For this reason it was decided to limit the offshore portion of this evaluation to the Brandywine Lighthouse located in Delaware Bay. Brandywine Lighthouse was located within the useable operating radius of the aircraft while the existing oil rigs are located 60 miles east of Atlantic City. Section 4.0 identifies the specifics of the test design. Approximately 20 hours of the 30 hour flight test program were assigned to offshore testing. The remainder of the testing was dedicated to the airport beacon with cursor testing.

In order to investigate a spectrum of target signatures and various track orientation techniques it was decided to investigate radar performance against: a) skin paint targets over water, b) skin paint targets utilizing the cursor technique, c) beacon and skin paint targets over water (combined mode), d) beacon targets over water, e) beacon targets over land utilizing the cursor technique, and f) multiple beacon targets over land.

Two other issues relating to the scope of the current effort should be discussed. First, there were two airborne radar systems available for test, the RCA Primus-50 and the Bendix RDR-1400A. Both systems have both primary ground mapping and beacon modes. In addition, the PCA unit has a combined beacon/ground mapping mode (which was included as a requirement in the ARA MOPS generated by RTCA SC-133). Since the combined beacon/ ground mapping mode was a requirement in the ARA MOPS, six of the thirty hours of the flight test program were dedicated to the RCA Primus-50 combined mode testing. Also the Primus-50 was tested in the beacon-only mode. The skin paint and cursor testing utilized the Bendix RDR-1400A radar system. For the cursor testing the Bendix radar system was modified to electronically display a cursor which indicates track angle error utilizing synchro inputs from the Horizontal Situation Indicator (HSI).

The second issue relates to the subject of the pilot population sample used for these tests. Radar display interpretation and pilot steering techniques form a major portion of this investigation. Although a wide variety of subject pilots is usually a goal in the design of such flight experiments, due to the small number of flight hours available for these tests, the number of pilots used was limited to three. During the single and multiple beacon testing a fourth pilot was utilized but because of his retirement only three qualified CH-53A pilots remained. The subject of pilot performance variability should properly be studied in a more comprehensive and dedicated experiment.

2.3 PURPOSE OF THE TESTS

Simply stated, the purpose of this ARA test program was twofold. First, to acquire a statistically significant data base, concerning operational procedures utilizing various track orientation techniques and overall ARA system performance that will assist the FAA and the airspace users alike in developing and certifying standard approach procedures and associated weather minimums through the application or modification of TERPS criteria. Second, to quantify specific ARA system performance parameters for use by RIFA 50-100 and the fea in specifying ARA required technical performance. The state there alon objectives is expanded and discussed in depth constants.

2.4 ORGANIZATION OF THE REPORT

The results of the ARA flight test program are presented in the remainder of this report. Section 4.0 provides a detailed equipment summary, a flight test description, a review of test profile designs, data acquisition procedures and data reduction/analysis techniques. Sections 5.0 presents and documents the specific results obtained in five major areas:

- 1) Airborne Radar as an Approach Aid
- 2) Analysis of Pilot Procedures
- 3) Detailed Accuracy Data
- 4) Operational Evaluation of the ARA Concept

Section 6.0 presents a summary of the technical and operational performance of the ARA System. Finally, Section 7.0 presents major qualitative conclusions as they relate to the stated program objectives from Section 3.0

DETAILED TECHNICAL OBJECTIVES

The general objectives of the ARA track orientation concepts were to evaluate and/or establish basic ARA operating procedures and overall system performance so that a direct quantitive comparison could be established between the raw radar return testing and the cursor aided or multiple beacon testing. The results of this program are therefore applicable to the FAA, RTCA and the user community (HAA). For purposes of discussion these test objectives have been grouped into two categories, namely Technical and Operational. Subsequent to this discussion, a correlation will be presented between the stated test objectives and the specific test results obtained (Section 5.0).

И

A. <u>Technical Performance Objectives</u>

.....

3.0

- Range Performance -- To establish the maximum and minimum radar ranges at which beacon and skin paint targets, respectively, can be acquired, identified and tracked. SC-133 has specified a minimum range requirement of at least 25 nm in clear weather and 15 nm with 4 mm/hr/nm of intervening precipitation. Minimum range is specified at 1000 ft.
- Bearing Accuracy -- To determine, for the system tested, the accuracy in bearing with which a target can be displayed; SC-133 requires ±3°.
- 3) Display Readability -- To validate the specified display readability. SC-133 currently requires that the display be functionally readable when viewed under conditions of 20,000 lux impinging upon the display face.

- 4) Impact of Antenna Stabilization -- To validate the requirement of SC-133 for antenna stabilization up to a vector sum of +30° for combined roll, pitch and yaw and to evaluate the impact of values in excess of +30° on display discrimination.
- 5) Display Resolution -- To assess display resolution requirements. Although this parameter is an inherent system design characteristic, it is considered desirable to obtain data on subjectively viewed display resolution for comparison with the requirements of SC-133.

B. Operational Performance Objectives

- Beacon/Ground Clutter Discrimination -- To evaluate the operational viability of the combined beacon/ ground mapping mode of operation as currently required by SC-133.
- 2) Offshore Target Discrimination -- To evaluate the ability of the ARA system to acquire and identify offshore targets in a variety of back scatter conditions as influenced by sea state and water depth parameters. While not an explicit test variable, careful note should be made of such conditions during each test in order to establish any possible correlation.
- 3) Cursor Interpretation -- To evaluate the ability of the radar operator to duide an aircraft along a predetermined path using an electronically generated cursor which obtains its input directly from HSI course selection. This modification should provide the operator with orientation guidance relating to aircraft position and heading, desired course, and target location.

- Beacon Proximity -- To evaluate the effect of range between multiple beacons such that individual targets can be acquired and identified.
- 5) Lateral Cross Track Error and Flight Technical Error --To establish statistically significant values for lateral cross track error for the overall ARA system and lateral flight technical error for the pilot/ operator under actual operational approach conditions for each of the basic modes of operation (beacon, beacon w/cursor, multiple beacon, skin paint, combined and skin paint w/cursor). These values quantify the ability of the pilot to utilize the ARA system to maintain a desired lateral ground track.
- 6) Longitudinal Along Track Error and Letdown Error --To establish statistically significant values of longitudinal along track error for the overall ARA system and the along track flight technical error (Letdown Error) for the pilot/operator. These values quantify the ability of the pilot to utilize the ARA system to define and identify a step-down fix and/or a missed approach point in order to execute a nonprecision approach vertical profile.
- 7) Pilot/Operator Procedures, Workload and Blunder Performance -- To establish quantitative measures, whenever possible, of pilot performance factors such as operational procedures, comparative workload and blunders as related to the different ARA operating modes (beacon-only, beacon w/cursor, multiple beacon, skin paint, skin paint w/cursor and combined).

Section 4.0 which follows, describes the ARA test plan which was configured to meet these stated objectives. Following Section 4.0, a correlation between test objectives and specific flight test results is presented.

4.0 DESCRIPTION OF THE ARA EXPERIMENT DESIGN

This section describes the equipment, test profiles, procedures, subject pilots and data requirements necessary for the Airborne Radar Approach flight test program. The experiment was designed to test two different types of airborne weather/ground mapping radars as an approach aid to landing using six different modes. These are as follows: beacon, beacon with cursor, multiple beacon, skin paint, skin paint with cursor, and combined. Two distinct environments were included in the testing to determine the capability of the airborne radar to aid the pilot in making a safe approach where other navigational aids are not available. These were as follows: airport and offshore site. At the airport site two different approach profiles were utilized, while at the offshore site only one approach profile was used.

4.1 EQUIPMENT SUMMARY

This section describes the equipment used in the Airborne Radar Approach flight test program at NAFEC in Atlantic City, New Jersey.

4.1.1 The Bendix RDR-1400A Radar

The Bendix RDR-1400A Radar system tested was a multi-mode, 10 KW X-band airborne radar modified to electronically produce a radar cursor display of track angle error. The system provides up to three air-tosurface search and detection modes plus the usual weather avoidance mode. It also contains the additional capability of a transponder beacon mode. The search modes provide both a ground mapping function along with the ability to detect and display prominent surface objects. The beacon mode is a special function used to interrogate and receive pulses from a ground based transponder(s) within line-of-sight range. The bearing and distance of the beacon target is then displayed on the CRT (cathode ray tube) free of any ground clutter.

The RDR-1400A uses digital techniques to continuously display a reliable return from significant weather or terrain. The display features an alphanumeric read-out directly on the screen depicting the selected mode, range, and range intervals. System checkout, either in the air or on the ground, is a straightforward procedure.

The entire of the second receiver-transformed in a second mounted in a second mounted in a second mounted in a second mounted in this second accessibility of the second mounted second mounted mounte

mits: enne tenne tenne tenstrument teasy enrols. Alt ene front

4

The mode selector switch located on the display unit offers six distinct display capabilities.

- a) SRCH 1 Mode normally used for over water search. This mode optimizes point targets within a sea clutter background. It is generally used for mapping ground targets or surface craft at short range. This mode optimizes short range resolution and clutter rejection.
- b) SRCH 2 Principal use for this mode is high resolution at all ranges. SRCH 2 offers no clutter rejection so it is generally use for ground mapping only. This mode offers precision ground mapping over many types of terrain.
- c) SRCH Mode normally used for the mapping of oil slicks. SRCH 3 offers long range mapping and/or maximum clutter returns.
- d) WX Weather mode: in this mode the receiver is optimized for weather detection. It provides early warning of bad weather and possible storm activity enroute.
- e) WXA This mode is the same as WX with one exception. When operating in the WXA mode the display flashes contoured areas to alert the pilot to clouds with high rainfall rates.
- f) BCN This mode has the capability to interrogate beacon transponders which receive a frequency of 9375 MHz and transmit back at 9310 MHz. The BCN mode allows the pilot to navigate to a predetermined target or landing site, while continuously displaying both range and bearing to the target. This mode

eliminates any background clutter and displays only the beacon or beacons within the field of view. Depending on aircraft position and altitude, the beacon target(s) can be received at considerable distances.

The RDR-1400A also offers an indicator test pattern. When the mode selector switch is in the TEST position the pilot then has the ability to determine if the radar is operating, either in the air or on the ground. The range selector switch offers range selection from 240 nautical miles to as close as 2.5 nautical miles, full scale, with range marks varying according to different scales selected. The tilt control adjusts the tilt of the antenna in relation to the longitudinal axis of the aircraft to allow best indicator presentation. Range of tilt control adjustment is -15 degrees. The receiver gain is adjustable for the search and beacon modes only.

In the weather moder the definite preset, therefore the gain control has no effect. The Scan Stab belector switch offers the opportunity to select one of two antenna scan angles. The 120° STAB position places the antenna in a 120 scan mode, (SO) each side of the aircraft longitudinal axis. The 40° SIAP position places the antenna in a 40° scan mode, (20° each side of the aircraft longitudinal axis.

The modified radar system cenerated an additional azimuth line or cursor shown in Figure 4.2 as a stashed line. The angular difference displayed between the lorg class control content the roturn is equal to track angle error. The control control content the roturn is equal to track angle error. The control control content on the Portrontal Situation Indicator control cland the control content carse error signal from the HSI is fed into the roturn evolves and and second and restore the subset. The cursor therefore represent to the additional electric to the aircraft heading which is begin evolves on the optimate line on the radar screen. When the converse control to the intended target the aircraft is on the decoded control the transfer of the manavered onto the desired control control control to the intended target the aircraft is on the decode control to the transfer of the summer does not split the material control control of the contour does not split the material control of the section.

. .



Figure 4.2 Bendix RDR-1400A Front Panel With Cursor Display

í

ŧ
The following Figures (4.3 & 4.4) shows typical on-course and off-course indications.







Figure 4.4 - Healing Curson Technique (Un-Course)

4.0

A LAN

Two rules of thumb can be emphasized when using this technique. Both apply to a no-wind situation, but with application of proper drift angle logic, will apply to any situation:

- The target return should be kept between the error cursor and the O° azimuth mark. This will insure interception of the desired final approach course prior to the missed approach point. The greater the angular distance between the return and the O° azimuth mark the sooner the approach course will be intercepted.
- Turning the direction corresponding to the direction from the error cursor to the target return will insure the proper direction of turn for course correction.

4.1.2 The RCA Primus-50 Radar

The RCA Primus-50 Radar system tested was a multi-mode, X-Band airborne radar. The system provided three distinct modes of operation. They are as follows: beacon-only, weather avoidance, and combined mode. The beacon-only mode offers the cabability of being able to interrogate, within line-of-sight range, a ground based transponder(s) beacon, as in the Bendix RDR-1400A. The combined or both mode is a special function used to interrogate and receive pulses from a ground based transponder(s) while at the same time displaying surrounding ground mapped returns. The bearing and distance of the target is then displayed on the CRT along with other surrounding skin paint targets.

The RCA Primus-50 uses digital techniques to continuously display a reliable return from significant weather or terrain. The radar display itself offers no alphanumeric read-out, therefore it is necessary to read selector switches to determine selected mode, range, and range intervals.

As in the Bendix RDR-1400A the entire RCA Primus-50 system consists of 3 separate units: receiver-transmitter, display indicator, and antenna. The antenna used was a gyro stabilized twelve inch slotted flat plate antenna mounted in a radome directly on the nose of the aircraft. The display indicator was mounted in the same location as



The capacity of the contract o

:

Bendix RDR-1400A the BCN mode allows the pilot to fly to a predetermined target or landing site, while continuously displaying both range and bearing to the target, free of any background clutter.

- b) RAD ~ Weather Mode: This mode offers an optimized means for weather detection. It provides early warning of bad weather and possible storm activity enroute.
- c) BOTH Combined Mode: This mode offers the capability of receiving and displaying both primary targets and the transponder beacon return at the same time. To make the beacon return distinguishable from the primary returns the system flashes the beacon display block on and off at one second intervals.

2) Pushbutton Switches

- a) OFF Pushbutton switch used to turn system off.
- b) STBY Pushbutton switch used to turn radar on. Standby is useful for keeping the radar in the ready state while taxiing, loading, etc.
- c) NORM Normal Switch used to activate the beacon and combined mode function or for routine weather mapping in the RAD mode.
- d) CTR Contour switch enhances contours of high rainfall rate in the weather mode.
- e) CYC Pushbutton switch used to select CYC dual operation which cause displayed contourable targets to flash on and off at 0.5 - second intervals.
- f) AZIMUTH Azimuth pushbutto: places azimuth lines every 15°.
- g) MAP Pushbutton switch that enables ground-mapping.
 This function enhances background clutter.

3) Rotary Controls

- a) RNG Rotary range selector switch. This switch offers the operator range selection from as far out as 150 nautical miles to as close as 2 nautical miles, full scale, with range marks varying according to different scales selected. Included in this switch is the TEST mode which offers straightforward system checkout.
- b) TILT This control enables the pilot to select desired angle of beam tilt with relation to the earth's plane. Control index references increments of tilt in degrees from 0 to 15 degrees up and down. The tilt is additive to any elevation correction by the stabilization circuits.
- c) GAIN This rotary switch is used to adjust the sensitivity of the receiver. It offers a PRESET position used normally for weather alert and adjustable gain levels used in the beacon-only and combined modes. This switch acts as a dual function switch allowing the operator to adjust the desired CRT intensity.
- d) STAB/SWEEP STAB is used to turn the antenna stabilization function on or off. SWEEP allows the selection of two distinct full scale sweep rates of 60 and 120 degrees. The FRZ (freeze) Mode, when enabled, freezes the display in its last updated position.

4.1.3 The Transponder Beacons

The transponder beacons used during the Airborne Radar Approach testing were manufactured by the Motorola Co., Model SST-181 X-1. The transponder beacons operate at a receive frequency of 9375 MHz and a transmit frequency of 9310 MHz. They have a power output of 400 watts. The transponder beacons were powered by a series of twelve volt lead-acid batteries mounted on a small cart. Figure 4.6 shows the transponder beacon mounted in a portable, water-tight case.

4.1.4 Flight Test Helicopter

The aircraft utilized for the Airborne Radar Approach (ARA) flight test program was a NASA Sikorsky CH53A helicopter (N-39) as shown in Figure 4.7. This type of aircraft normally cruises at 140 kts and is primarily used by NAFEC and NASA for flight test purposes. The test aircraft was based out of NAFEC in Atlantic City, New Jersey.

The CH53A helicopter is a fully operational IFR aircraft. Both the pilot and copilot have full sets of operating flight controls and instruments. The center console houses control heads for single UHF and HF, and DUAL VHF communications systems, along with DUAL Collins NCS-31A RNAV systems. The front panel contained various instruments including a flight director, HSI, RMI, Radar and Barometric altimeters.

4.2 TEST PROFILES AND PROCEDURES

Twenty-five flights were flown in the Airborne Radar Approach (ARA) flight test program at NAFEC in Atlantic City, N.J. The twenty-five flights flown involved utilizing two distinct radar systems, various track orientation techniques and operational modes. Of the twenty-five flights, ten were performed in the multiple beacon mode using two longitudinally spaced beacons during the testing period from 1 January 1979 to 2 February 1979 with one additional flight accomplished on 7 August 1979. Table 4.1 presents the Bendix RDR-1400A multiple beacon flight test matrix, showing test location, beacon spacing and total number of approaches flown. Five of the twenty-five flights flown utilized the RCA Primus-50 radar system. The testing using the Primus-50 involved Brandywine lighthouse (offshore site) where two operational modes were tested during the period from 26 June 1979 to 29 June 1979. Table 4.2 presents the RCA Primus-50 flight test matrix, showing operational mode and number of approaches. During the period from 16 July 1979 to 30 July 1979 six of the twenty-five flights were accomplished in the skin paint and skin-paint w/cursor modes. Table 4.3 presents the Bendix RDR-1400A skin paint and skin paint w/cursor flight test matrix, showing which





FLIGHT	DATE	LOCATION AIRPORT SITE	BEACON SPACING	NUMB APPR 25 NM	ER OF OACHES 10 NM	PILOT	COPILOT
1	1/19/79	Rwy 13	10,400'	1	3	A	В
2	1/25/79 am	Rwy 13	10,400'	1	2	A	С
3	1/25/79 pm	Rwy 13	10,400'	1	3	с	D
4	1/26/79 am	Rwy 13	5,000'	1	3	D	A
5	1/26/79 pm	Rwy 13	5,000'	1	3	С	D
6	1/31/79 am	Rwy 13	5,000'	7	3	В	С
7	1/31/79 pm	Rwy 13	15,000'	1	3	A	В
8*	2/5/79 am	Rwy 13	10,400'	1	3	A	D
9*	2/5/79 pm	Rwy 13	15,000'	1	3	D	В
10	8/7/79	Rwy 13	10,400'	1	2	С	D
			Total	10	29	A-5	B-4
						C-5	D-6

Table 4.1 Bendix RDR-1400A Multiple Bracon Flight Test Matrix

*Note: A 3dB attenuator was implemented on the near beacon to reduce splaying.

FLIGHT	DATE	LOCATION	NUMBER OF OPERATING APPROACHES MODE 25 NM 10 NM		PILOT	COPILOT	
1	6/26/79	Offshore	Combined	1	2	D	С
2	6/27/79 am	Offshore	Combined	1	2	С	В
3	6/27/79 pm	Offshore	Combined	1	2	В	D
4*	6/29/79 am	Offshore	Combined/ Beacon-Only	۱	3	С	В
5	6/29/79 am	Offshore	Beacon-Only	۱	3	В	D
		Total	Combined	4	7	A-0	B-4
			Beacon-Only	1	5	C-3	D-3

Table 4.2 RCA Primus-50 Flight Test Matrix

*Note: Two approaches were flown in the combined mode and two in the beacon-only mode.

flights were flown utilizing the new cursor technique, location, and total number of approaches. The remaining four flights were flown at the airport site using the modified Bendix radar system in the beacon with cursor mode. These four flights were accomplished during the period from 2 August 1979 to 6 August 1979. Table 4.4 presents the Bendix RDR-1400A single beacon w/cursor flight test matrix, showing type of profile flown, location, and total number of approaches.

FLIGHT	DATE	LOCATION	OPERATING MODE	NUMBER OF APPROACHES 25 NM 10 NM		PILOT	COPILOT
1	7/16/79	Offshore	W/O Cursor	1	2	С	D
2	7/17/79 am	Offshore	W/O Cursor	٦	2	D	С
3	7/17/79 pm	Offshore	W/O Cursor	1	2	С	D
4	7/19/79	Offshore	W/Cursor	1	2	D	С
5	7/27/79	Offshore	W/Cursor	1	2	D	С
6	7/30/79	Offshore	W/Cursor	1	2	С	D
		Total	W/O Cursor	3	6	A-0	B-0
			W/Cursor	3	6	C-6	D-6

Table 4.3 Bendix RDR-1400A Skin Paint and Skin Paint W/Cursor Flight Test Matrix

4.2.1 General Pilot Procedures

The pilot procedures which were used took into account airspace requirements, obstruction clearance and noise abatement considerations. During the tests the copilot was hooded and the pilot was unhooded. Prior to take-off, the pilot input RNAV waypoint coordinates to aid in getting established on the desired approach course. Upon completion of the procedure turn inbound and just prior to reaching the Initial Approach Fix (IAF) inbound, the copilot took over navigation using the airborne radar. It was the copilot's responsibility to call out heading (e.g., turn to heading 180°) and/or heading changes so that the pilot

Table 4.4 Bendix RDR-1400A Single Beacon W/Cursor Flight Test Matrix

	DATE FLOWN				8/2/79 am	8/2/79 am	8/2/79 pm	8/6/79	
	COPILOT				Q	U	G	J	B-0 D-4
	PILOT				ى د	۵	U	Q	A-0 C-4
	NUMBER OF APPROACHES				4	3	4	m	14
	INTENTIONAL OFFSET	25 5 25 5	26 08					× ××	2 1
AIRPORT SITE	OVERHEAD STRAIGHT	25 5 25 5	26 08			x x x	×× ×		1 2 3
	DIRECT STRAIGHT	25 5 25 5	26 08		× × × ×		×		2 1 1 1
LOCATION	PROCEDURE	Segment Length nm	Runway	Flight	F =	(51	m	- 1	-Cta]

•

} : |-†

ğ

5

could maintain the desired track. It was also the copilot's responsibility to call out range from the target at one mile intervals and altitudes depending on which profile was flown. On reaching the Initial Approach Fix, the copilot called out range and altitude so that the pilot could start a descent and cross the Final Approach Fix (FAF) not less than 500' Above Ground Level (AGL). After crossing the IAF and stabilizing, a specified rate of descent was established and the airspeed was decreased to 90 knots. On crossing the Final Approach Fix (FAF), a descent was established so that the Minimum Descent Altitude (MDA) would be reached at no less than one half $(\frac{1}{2})$ mile from the landing area. Again, after the rate of descent was established, the airspeed was then decreased to 50 knots and held there until either the landing area was in sight or a missed approach was executed. If the landing area was not in sight, the copilot called out "missed approach", at which time the missed approach procedures were executed. The missed approach executed was indicated on the approach plate used. It consisted of a climbing right or left turn to 1000' to intercept the Initial Approach Fix. The pilot's other duties included radio communications and watching out for traffic. It should be noted that even though the pilot flew unhooded he was instructed to fly only those headings and heading changes indicated by the copilot. Figure 4.8 presents the Airborne Radar Approach geometries for both the airport and offshore sitës.

4.2.2 Bendix RDR-1400A Multiple Beacon Testing

The effort in this portion of the testing was concentrated on using two independent ground beacons to establish visual reference indications of the desired final approach course on the airborne radar display. It is the functional equivalent of the multiple-reflector technique used in the ground mapping mode. Ten flights were accomplished in the multiple beacon testing and of these ten flights six were non-recoverable from the film interpretation standpoint because of beacon splaying. Although not known during the testing, it was later discovered that the STC circuit in the Bendix radar system was improperly adjusted. The testing accomplished after the adjustments were made verified this fact to some extent, but it was still found necessary to constantly adjust the gain control throughout the approach.



A THE R



The double beacon technique has applications primarily to landside sites only. Therefore, the airport site was utilized as the testing site. Runway 13 was picked because it allowed large longitudinal distances for beacon placement. During the double beacon testing, beacon separation was included as a controlled variable, with one beacon permanently positioned at the target landing zone (threshold of runway 13) and the other positioned in line with the final approach course (runway centerline) at varying distances beyond the first.

The procedures utilized were identical to those used during the single beacon approach testing from October 1978 to December 1978. Pilot procedures, as described in Section 4.2.1, and profiles were generated in advance to offer the test program a controlled environment. The test profile flown was the direct straight profile as shown in Figure 4.9. This direct straight procedure was utilized quite extensively during the single beacon approach testing (Reference 1) conducted at the airport, remote and offshore sites. Normally the direct straight procedure is used when the winds are favorable for landing upwind in the approach direction.

4.2.3 RCA Primus-50 Combined and Beacon-Only Mode Testing

Five data flights were flown at the offshore site in the combined and beacon-only modes using the RCA Primus-50 radar system. All of the data flights flown were of acceptable quality with the exception of the first where it was found necessary to omit the first two approaches from the data base due to instrumentation difficulties. The combined mode feature of the Primus-50 offers the capability of mapping both beacon and ground returns simultaneously. This mode of operation was included as a requirement in the draft ARA MOPS (Minimum Operational Performance Standard) generated by RTCA SC-133. This series of flight tests supplied data to formulate a combined and beacon-only mode data base suitable for comparison to that established during the extensive testing performed in the single beacon mode using the Bendix System from October 1978 to December 1978.

The procedures for the Primus-50 testing utilized the direct straight procedure at the offshore site. The offshore site was chosen because



Figure 4.9 Bendix RDR-1400A Multiple Beacon Direct Straight Procedure

it offered sufficient skin paint targets necessary to determine the operational feasibility of the combined mode. Figure 4.10 presents the approach plate used during the Primus-50 testing. As before, the direct straight procedure was utilized for the descent profile. RNAV waypoint coordinates were established to aid the pilot in getting established on the proper approach course. One difference should be noted between the multiple beacon procedure (Figure 4.9) and the Primus-50 procedure. Instead of using the 10 nm IAF as the RNAV waypoint, as was the case in the multiple beacon tests, the RNAV waypoint was established as a fix 12 nm from the intended target. This extra 2 nm gave the pilot time to get established on the final approach course before handing off navigation to the copilot.

4.2.4 Bendix RDR-1400A Skin Paint and Skin Paint W/Cursor Testing

The effort in this portion of the testing was to use a prominent surface object (Brandywine Lighthouse) to simulate Airborne Radar Approaches to offshore sites with and without the aid of a cursor. The testing performed included six data flights and all were considered to be of very acceptable quality. During previous skin paint testing, local shipping occasionally created confusion on the part of the operator. The previous tests performed also offered no quantitative data so that the overall accuracy could be determined. These six flights flown from 16 July 1979 to 30 July 1979 offered a data base large enough to compare single beacon versus skin paint results at offshore sites and to also determine to what degree the pilot is aided by the use of a cursor in such an environment. As mentioned earlier the modified Bendix radar system offers the presentation of a cursor display electronically generated from the OBS setting on the HSI. The purpose of the cursor is to reduce the "homing" effect caused by the lack of track orientation information presented on the radar screen. In most of the skin paint testing the SRCH 2 mode was used because of its high resolution at all ranges in small sea states.

The same basic procedures were utilized in the skin paint testing that were used in the Primus-50 testing. The direct straight procedures were used and also the same waypoints were utilized for track acquisition.



Figure 4.10 RCA Primus-50 Offshore Direct Straight Procedures

Figure 4.11 presents the approach procedures utilized during the Bendix RDR-1400A offshore skin paint and skin paint with cursor testing. The same procedures were used so that a direct correlation of results could be made possible.

4.2.5 Bendix RDR-1400A Single Beacon With Cursor Testing

The single beacon with cursor testing consisted of four flights, with a total of fourteen approaches. Of these fourteen approaches two were not recoverable due to film processing difficulties. Two different types of approaches were conducted at the airport site providing a reasonable data base so that cursor aided approaches could be compared to non-cursor aided approaches. As in the other areas of testing, pilot procedures and profiles were generated in advance in order to insure a well disciplined test environment.

The procedures used during this phase of testing were identical to those described in Section 4.2.1. Figures 4.12 and 4.13 show the direct straight and overhead straight approach profiles flown with their associated plan views. The procedures used for flying the two profiles were identical, with one exception. On the overhead approaches it was necessary for the copilot to call out when directly overhead the target so that the pilot could start an outbound timing of 2½ minutes. At the end of 2½ minutes, a procedure turn was executed and the Intermediate Approach Course was acquired. The overhead offset procedure was not utilized during this phase of testing because of the limited number of flights. The inclusion of the overhead offset approaches would have increased the data base to an unacceptable level with the available sampling.

During the single beacon with cursor testing a new test variable was introduced. It involved intentionally offsetting the aircraft from the beginning of the approach, based on specific instances of aircraft placement discovered during the analysis of the data collected during the single beacon approach testing. The purpose of the intentional offset procedure was to determine the effectiveness the cursor technique. That is, can the radar operator acquire the intended course



1

Figure 4.11 - Bendix RDR-1400A Offshore Direct Straight Procedures

4-24



MISSED APPROACH:

Figure 4.12 Profile 1: Direct Straight Airport Site



: 1

.



26.21 11 12

using the cursor and track it inbound without homing to the target. Due to film difficulties only one of the intentional offset approaches was recoverable. The tracking plot for the earlier approach selected as a comparison for the intentional offset profile can be seen later in Section 5.0.

Only one beacon transmitter was installed for this test series. The beacon transmitter was located at the threshold of runway 26 regardless of approach direction.

4.3 SUBJECT PILOT EXPERIENCE AND TRAINING

Four subject pilots were used for the Airborne Radar Approach flight test program using various track orientation techniques. All four pilots alternated as pilot and copilot during part of the test program. Due to the retirement of one subject after the completion of the multiple beacon testing the three remaining subject pilots completed the test program. All pilots involved were FAA personnel resident at NAFEC in Atlantic City, N.J.

Proficiency in using the airborne radar came about through actual operational use. This operational use was accomplished during the skin paint and single beacon approach testing from July 1978 to December 1978. There were two pilots per crew with the copilot being the only crew member hooded. The pilot was not hooded for safety reasons, but was instructed to fly only those course headings indicated by the copilot. It was also the pilot's responsibility to handle all communications. A summary of each pilot's experience level is presented in Table 4.5.

Subject Pilot	Total	Rotary Wing	Fixed Wing	CH53A
А	11,500	4,500	7,000	55
В	16,010	415	15,595	125
С	17,000	1,500	15,500	160
D	17,175	1,800	15,375	115

Table 4.5 Flight Experience of Subject Pilots In Hours

4.4 DATA ACQUISITION AND PROCESSING

This section describes the methods used to instrument, record, recover, and process the flight test data for the Airborne Radar Approach testing.

4.4.1 Airborne Instrumentation

The airborne instrumentation consisted of electronically recording selected parameters using a Litton LTN-51 Inertial Navigation System (INS) interfaced with a Kennedy magnetic tape recording system on the multiple beacon program and a Norden MZ-RX11-DB Dual floppy disc drive recording system on the later cursor aided approach testing. The following parameters were recorded: time, latitude, longitude, ground speed, true heading, and track angle. In addition to recording inertial position data, the radar screen itself was photographed along with a digital display of time and aircraft magnetic heading. The film data provided the simultaneous recording of many parameters such as: time, heading, beacon position relative to zero azimuth, scale, sweep, tilt, and gain. The optimum photographic recording rate was determined to one frame every two seconds. Other rates were tried but were found to be inadequate due to the limited amount of film available on each cartridge, or due to insufficient data recording frequency.

4.4.2 Ground Reference Data

The ground reference data was obtained using the NAFEC "Extended Area Instrumentation Radar" (EAIR). The EAIR radar was utilized as the indicator of actual aircraft position, by detecting and recording (real time) the azimuth, elevation and range of the test aircraft. EAIR is a precision, C-band tracking radar which provides the slant range, azimuth angle and elevation angle of an aircraft within a range of 100 nautical miles when operating in the skin tracking mode, with a maximum distance of 190 nautical miles when operated in the beacon tracking mode. (All of the ARA test flights were tracked in beacon tracking mode). The slant range obtained by the FAIR facility is accurate within 20 yards and the azimuth angle and elevation angle are accurate within 0.011 degrees. For example, at 50 miles the accuracy would be 20 yards in range and 20 yards in azimuth and elevation. The radar antenna can track a target 360° in azimuth and from 0° to +89° in elevation. The antenna can be directed as low as minus one and one-half degrees in elevation.

4.4.3 Manually Recorded Flight Logs

During all flights a trained cockpit observer monitored and kept an accurate log of routine and special events that occured during the flight. The observer was responsible for documenting the crew workload and performance. The flight logs recorded by the observer were a major source of data acquisition from which flight test results could be operationally evaluated. The following is a summary of the flight test data recorded by the observer during each flight.

- 1) Procedural Errors
- 2) Elapsed Time
- 3) Altitude
- 4) Airspeed
- 5) Aircraft Heading
- 6) Radar Approach Distance
- 7) Radar Mode
- 8) Radar Range Scale
- 9) Radar Gain Position
- 10) Radar Tilt and Stabilization
- 11) Pilot Workload

4.4.4 Data Processing

The airborne and ground-derived position tracking data were used to determine the capability of an airborne radar approach procedure to guide a helicopter along a predetermined approach path to a target using various types of track orientation techniques. To logically evaluate this capability, the following basic group of measures were computed for each test approach:

 Helicopter deviation from the intended track (the track to be defined by an inbound bearing to the runway threshold or landing zone).

- Radar sensor error in both the along track and cross track directions.
- Flight technical (FTE) which is a measure of pilotage error in the cross track direction at all ranges.
- 4) Letdown error (LDE) which is a measure of pilotage error in the along track direction at step down fixes. LDE quantifies the pilot's ability to identify the step down fix from the radar display.

The raw flight test data was reduced according to the following steps:

- A projected preview of each film was performed to identify the targets, before digitization of the photographic data.
- 2) The relevant photographic data was then recovered by projecting the data on a digitizer tablet which was interfaced with a computer. In addition, data read from each frame such as time and heading was also inputted. The overall return dimensions were recovered wherever possible, since return size and shape played an important role in pilot orientation during each approach.
- While digitizing, computer routines were used to convert digitized points to radar range and azimuth coordinates.
- 4) The file was then transmitted via dataphone to the time sharing system, where NAFEC EAIR and INS data tapes had also been sent for processing.

5) The airborne data was then merged with the tracking information to produce a complete data file from which navigation error measures were derived. EAIR tracking was the primary source of ground truth data, but on some approaches to the offshore site EAIR tracking was lost due to low altitude, in which case the INS had to be used. In those cases it was necessary to perform a three-way merge. When the three-way merge was complete, EAIR tracking dropout times were matched with the original EAIR tracking and INS printouts. Using these dropout times, the corresponding EAIR latitudes and longitudes were noted. In order to remove the effects of INS drift, differences in latitudes and longitudes were computed by substracting the EAIR tracking values from the INS values.

 Δ lat = lat (EAIR) - lat (INS) $\Delta 1 \text{on} = 1 \text{on} (EAIR) - 1 \text{on} (INS)$ These values were then supplied to the error analysis program, which used the three-way merge as the input file. This program then sequenced from EAIR to INS as the position standard at the times manually arrived at earlier. The program also makes use of the lat/lon correction factors, and interpolates linearly between the two corrections to yield an INS correction factor for each data point. When the merge was complete, navigation errors were computed. These were as follows: Total system cross track error, flight technical error, airborne system cross track error and airborne system

along track error. The precise definition of these error quantities is specified in Section 4.5.1. The statistical treatment of these error quantities will be discussed in Section 5.

6) Due to abnormal hardware difficulties on a few approaches a time correlation problem was encountered between the tracking data and the airborne data. To resolve this problem it was necessary to derive track angle along the approach path from the EAIR tracking data. Then a plot was generated of track angle versus time for both the EAIR tracking and INS data. When the plots were completed the data was correlated and the time correction factor noted. This time correction factor was then added or subtracted to the airborne data depending on the direction of the time shift. Figure 4.14 presents typical plots from which time correlation factors were obtained.

Afterwards, the data was then merged with the ground reference data.

Observer log data as well as pilot and copilot workload ratings were also evaluated for each approach.

4.4.5 Data Processing Facilities

Data processing was accomplished using a combination of a dedicated microcomputer system resident at the SCI (Vt.) facility, and a remote time-sharing system. Data was recovered using a Summagraphics digitizer tablet interfaced with a 48K Byte North Star microcomputer system. The digitized data, along with the parametric data read from each frame (time, heading, scale factor, sweep angle) keyed in from a CRT terminal, was stored on disk. Once complete files of data for each test were assembled, they were transmitted by direct computer data interchange to the CDC Cybernet system. The Cybernet was then used to load the INS and EAIR tapes, perform the merge step, and then perform the error measure derivation and statistical analysis steps.







4.4.6 Data Digitizing

To automate data recovery and reduce both the manual effort and the inherent potential for error, a digitizer tablet was used. Interfaced with a computer and using X-Y coordinates, the tablet allows direct entry of a broad range of data types (graphs, plans, maps, photographs, etc.) with a high degree of resolution. This technique was used with an image of each frame of film photographed in the ARA tests projected directly onto the tablet itself. -

Exact registration with the tablet coordinate system was not necessary, and the problems associated with scale maintenance were eliminated, since the computer algorithm makes the necessary scale and registration adjustments frame by frame. For instance, several reference points on the CRT image (e.g., range marks) were first digitized by touching the tablet stylus to those several points in a pre-determined order. The computer then calculated scale and registration factors for that frame. The operator could then digitize the endpoints of the radar target, resulting in accurate measures of target azimuth, range and size.

4.5 DATA ANALYSIS PROCEDURES

Based on the defined intended track, the actual range/azimuth and cross track error were computed, along with airborne system error (radar and heading sensor error), and flight technical error (FIE). The production of the measures permitted a statistical analysis of each approach segment. An overall review and statistical aggregation was the result of this data processing: also plots were generated depicting the same information represented in the statistical analysis. The sample size for these quantities was determined by the number of film data points collected. It should be noted though that some data presented in Section 5.0 includes a count of the number of film data points collected and the number of approach segments flown. An outline of the data included in this report is presented in Table 4.6.

4.5.1 Navigation Error Analysis

Three measures of navigation error are desired: total systemetror (as measured by the CAIR and INS tracking systems), flight

Table 4.6 Data Processing and Analysis Outline

MEASURE: TOTAL SYSTEM ERROR (along track and cross track)

Source

- EAIR Tracking Radar Data
 INS Platform Data
 Desired Track Parameters

Presentation

Plots of Actual Track vs Desired Track
 Histogram of Total System Error

Statistical Analyses

- Mean
 StanLard Deviation

Applications

- TERPS Protected Airspace Comparison of Enhancement Nodes
- MEASURE: LETDOWN ERROR (along track)

Source

- .
- EAIR Tracking Radar Data INS Platform Data Airborne Radar Distance to Landing Zone Step-down Fix Distance Aircraft Altitude Time Synchronization

Statistical Analyses

Mean
 Standard Deviation

Applications

- TERPS Fix Displacement Error
- MEASURE: FLIGHT TECHNICAL ERROR (cross track)

Source

- EAIR Tracking Radar Data ISS Platform Data Airborne Radar Data
- - range azimuth
- Aircraft Heading Aircraft Altitude Time Synchronization

Presentation

Plots of FTE as a Function of Range to Landing Zone
 Histogram of Flight Technical Error

Statistical Analyses

- Nean - Standard Deviation

Application

- Certification Frror Budget

 NEASURE: AIRBORNE SENSOR ERROR (along track and cross track) Source

Total System Error
 Flight Technical Error

Presentation

- Plots of Sensor Error vs Range to Landing Zone Histogram of Sensor Error

Statistical Analyses

Mean
 Standard Deviation

Applications

- Certification Error Budget
- MEASURE: BEACON PROXIMITY DATA (range and assmuch)

Source

•

- Airborne Range to Landing Zone
 Reacon Presentation on Radar Display (target size)

Presentation

Beacon Image Size vs Range to Landing Zone (sensitive to gain setting)

Applications

- Operational Evaluation of Enhanced Targets Beacon Presentation Certification Data Display Resolution for SC-133

MEASURE: AIRBORNE RADAR RANGE DATA

Source

- Airborne Radar Presentation

Presentation

- Maximum and Minimum Airborne Range vs Site (dependent upon gain setting)

Application

- SC-133 Requirement and FAA Certification Specification
- MEASURE: PILOT PROCEDURES AND BLUNDER ERROR DATA

Source

- Tracking Radar Data
 Observer Log

Presentation

- Blunder Type vs Operational Procedure Count of Procedural Error and Blunder by Type

Application

- Workload Assessment

- Portional Procedures Evaluation TERPS Protected Airspace Requirements FAA and SC-133 Specific Radar Design Requirements

technical error, and airborne system (radar and heading combination) error. These quantities were calculated from the measured parameters in the following manner:

TOTAL SYSTEM ERROR -- Total system error is the deviation of the aircraft from desired track (in the cross track direction) as measured by the tracking system (EAIR). After appropriate coordinate conversions, actual aircraft range (r_0) and bearing to the target (θ_0) were calculated given target coordinates, intended track bearing (θ_t) and aircraft position. Total System Cross Track Error (TSCT) and Total Along Track Distance (TATD) were calculated as follows:

TSCT =
$$r_0$$
 SIN $(\theta_0 - \theta_t)$
TATD = r_0 COS $(\theta_0 - \theta_t)$

FLIGHT TECHNICAL ERROR -- FTE is defined as the <u>indicated</u> deviation from desired course in nautical miles. The radar does not directly display this value, and so the pilot must deduce the indication from available parameters: radar range, (r_i) and azimuth (θ_i) to the target (landing zone) and aircraft heading (e_h) , plus knowledge of desired inbound track bearing (e_t) . Since the antenna boresight is in line with aircraft heading, indicated target azimuth (θ_i) is actually derived as being

$h_i = e_h + \Delta e_i$

where Δe_i is the indicated azimuth deviation from the centerline of the display (see following figure). The pilot tries to navigate the aircraft such that indicated target azimuth (\triangle_i) is equal to desired tracking bearing (\triangle_i) .



When this is true the aircraft is on course. Under a zero-wind condition, if the pilot successfully acquires and flies the intended course, then $\Delta \theta_i$ will go to zero. In the presence of a cross wind, $\Delta \theta_i$ would stabilize on some finite value of crab angle if the course is properly tracked. When θ_i is not equal to θ_t , an off-course condition is indicated. The value of the FTE indication is

also, $FTE = r_i SIN (\theta_i - \theta_t) (left is +)$ $ATD = r_i COS (\theta_i - \theta_t) (TO is +)$

where ATD is along track distance to the target AIRBORNE SYSTEM ERROR -- Cross track and along track components of airborne system navigation error were computed from knowledge of position calculated from tracking data and the indicated position of the radar/ heading combination. These errors, ASE (Airborne System Error), and ATE (Nav System Along Track Error were calculated as follows:

> ASE = TSCT-FTEATE = ATD-TATD

An additional error component is along track pilotage error experienced when a step down fix is crossed. Letdown Error (LDE) is defined as the difference between ATD when descent is actually initiated and the nominal along track distance of the step down fix as charted.

الله ا

-

「「「「「

ANALYSIS OF RESULTS

The purpose of this section is to provide detailed insight into the flight test results and data analysis for the Airborne Radar approach flight test program utilizing various track orientation techniques. The details presented in this section present the results of a comprehensive review of the specific data collected during the flight test program.

This section is divided into four subsections of data analysis to assist in the understanding and interpretation of the primary results. These categories are:

- 5.1 AIRBORNE RADAR AS AN APPROACH AID
- 5.2 ANALYSIS OF PILOT PROCEDURES
- 5.3 DETAILED ACCURACY DATA

5.0

5.4 OPERATION EVALUATION OF THE ARA CONCEPT

Due to the fact that two test areas were involved — Airport and Offshore — and that six radar operating modes were tested — multiple beacon, beacon with cursor, skin paint, skin paint with cursor, combined, and beacon-only — and that three types of data are of interest accuracy, functional and procedures — this results discussion begins in Section 5.1 with a brief summary of the major findings. This summary of results is then followed and substantiated by more detailed tabular and graphica' data analysis. This data analysis has been categorized as shown above into Sections 5.2, 5.3 and 5.4 for ease of reference. Also references will be made to an earlier ARA report (October 1979) written entitled "Airborne Radar Approach System Flight Test Experiment" (Reference 1). This report will be utilized so that comparisons can be made with the results obtained during the track orientation technique flight test program.

5.1 AIRBORNE RADAR AS AN APPROACH AID

The results presented in the following discussion have been compiled from the overall analysis of ARA as an approach aid utilizing various operational modes and track orientation techniques. The purpose of this section is to highlight the data and the quantitative results which provide the most significant impact on the qualitative

conclusions reached. To this end, Section 5.1.1 discusses primary results developed in detail in Section 5.2. Similarly, Section 5.1.2 summarizes Section 5.3 etc.

5.1.1 Pilot/Copilot Procedures and Workload

Pilot ARA procedures were developed which resulted in safe approaches with acceptable cockpit workload. These procedures were compatible with current ATC operational constraints in the airport and offshore areas. The profiles and procedures utilized during the track orientation testing were virtually identical to those utilized during previous testing (Reference 1). The conceptual aspects of the flight test included utilizing one of two basic flight profiles to perform the approach. These were the direct straight and overhead straight profiles. The emperical aspects of the ARA approach procedure were in the area of pilot/copilot responsibility assignments. The crew coordination procedures combined with the flight profile procedures resulted in a calm and coordinated cockpit environment.

The pilot/copilot workload involved in flying an Airborne Radar Approach is quite heavy. Based on qualitative observations experienced during the single beacon testing, the Airborne Radar Approach was concluded definitely to be a two pilot operation. The copilot's workload consisted primarily of three distinct items:

- Radar target interperation in relation to intended course (radar inteperation included track orientation utilizing the cursor and multiple beacon techniques).
- Operation of radar display controls
- Range and heading call-outs to the pilot

The copilot's mental workload was reduced by the use of various track orientation techniques. The skin paint with cursor, beacon with cursor, and multiple beacon approaches offered the copilot with more positive course guidance. During the course of the flight test program the pilots and copilots rated themselves on the level of mental and physical effort involved in flying each Airborne Radar Approach. The responses indicated that the approaches utilizing some type of track orientation technique offered only a moderate workload while workload ratings for other types of approaches ranged from moderate to very high.

Another high workload situation arose for the copilot in the skin paint mode. Because of the lack of positive target identification available in this mode of operation copilot workload was high. Numerous surrounding surface objects made positive navigation difficult.

The pilot's workload was quite different from that of the copilot. It was the pilot's sole responsibility to fly the aircraft and handle all Air Traffic Control (ATC) communications. As mentioned in Section 4.2.1 the copilot was hooded during the approaches while the pilot remained unhooded. The pilot was unhooded so that he could be aware of traffic in the surrounding area, especially in the airport terminal area. The pilot's workload mainly stemmed from two items: pilot/copilot and pilot/ATC communications, and aircraft handling. Handling of the helicopter became a problem because of the slow airspeed (50 knots) in high crosswind conditions.

It was concluded from the flight tests that some type of track orientation device is required to effectively fly an ARA approach. The track guidance offered by such devices, aids in track keeping therefore reducing the level of mental workload. This reduction in mental workload allows the pilot and copilot to concentrate more on the safety of the helicopter.

5.1.2 Overall Accuracy Assessment

The detailed ARA accuracy data is presented and analyzed in Section 5.3. This section summarizes ARA accuracy using two levels of analysis. First, overall ARA accuracy statistics are presented for the four primary error measures. These measures are Total System Cross Track Error (TSCT), Flight Technical Error (FTE). Airborne System Error (ASE) and Along Track Error (ATE). These error quantities have been defined previously in Section 4.5. Second, error statistics are analyzed with respect to along track distance from the target landing zone. The one-sigma error quantities presented are quantified in linear and angular terms at one, three and five nautical mile intervals from the target landing zone.
Table 5.1 summarizes the mean and one-sigma ARA error quantities obtained during the flight test program. The data in the table includes all offshore approaches flown with the Bendix RDR-1400A radar system regardless of the approach procedure. Section 5.3 discusses the details of long and short approach segments.

 Table 5.1
 Overall Bendix RDR-1400A Offshore ARA Performance Summary

	ERROR MAGNITUDES							
ERROR OUANTITY	Single Beacon*		Skin	Paint	Skin Paint W/Cursor			
, qc/(((****))) [1_	$\overline{\chi}(nm)$	+lə (nm)	χ(nm)	ിം (nm)	<u>x(nm)</u>	+1.5 (nm)		
TSCT	2293	.8873	.2570	. 3400	.1940	.5744		
FTE	0273	1.0185	.0615	.6350	0357	. 4778		
ASE	2021	.5353	. 1955	.6264	.2297	.5811		
ATE	.0033	.2383	.0345	.2567	.0340	. 3430		

ŧj.

*These results were obtained from Reference 1 (Section 5.3.3).

The results presented in Table 5.1 include three different operational modes. They are: single beacon, skin paint and skin paint with cursor. The results obtained in the single beacon and the skin paint mode are not significantly different. The TSCI mean values are similiar in magnitude. That is, the single beacon TSCT mean value is -.2293 nm and the skin paint TSCT mean value is .2570 nm. The one sigma single beacon TSCI value shown in Table 5.1 is .8873 nm and the skin paint one sigma value is .3400 nm. The fTE values for both areas are smaller than the ISCT values. Table 5.1 also indicates that the TSCI values are similiar in magnitude for both areas (single beacon and skin paint). The single beacon FTE mean is .0273 nm, while the skin paint fTE mean is .0615 nm. The system errors (i.e. ASE and ATE) are small for both areas of testing. The small mean and one-sigma ASE and ATE quantities inticates that the airborne radar is both reliable and repeatable.

Table 5.1 also presents the skin paint with curson approach testing results obtained at the offshore site. The field and FTE values obtained for the curson albed approaches induction a cest, then over the ISCL and

 $\gamma \to \bar{Q}$

FTE values obtained for the non-cursor aided approaches. For example, the skin paint with cursor TSCT mean presented in Table 5.1 is .1940 nm and the FTE mean is -.0357 nm. The one-sigma skin paint with cursor TSCT and FTE values are .5744 nm and .4778 nm, respectively. These data show a decrease of .23 nm in the one-sigma TSCT value over the skin paint without cursor TSCT value. The FTE one-sigma value indicates virtually the same decrease. Again, the small ASE and ATE values prove that the system performs well and is quite consistent.

ş

Tables 5.2 and 5.3 presents the performance of the ARA as a function of distance from the target. The tables illustrate the behavior of both linear and angular errors within the five mile distance-to-go area. Table 5.2 presents the offshore TSCT linear and angular error quantities for three different areas: single beacon, skin paint and skin paint with cursor. Table 5.3 presents the offshore FTE quantities for the same three operational areas. The most obvious fact observable in Table 5.2 is that the TSCT single beacon linear errors were acceptably small $(\pm .30 \text{ to } \pm .83 \text{ nm})$ in the 1-5 nm along track region. The skin paint and skin paint with cursor TSCT linear errors were also acceptably small $(\pm .18 \text{ to } \pm .51 \text{ nm})$. The angular TSCT quantities presented in Table 5.2 were somewhat larger due to the proximity to the target, these small cross track errors produce large $(\pm 4.5^{\circ} \text{ to } \pm 16.5^{\circ})$ angular errors. The table indicates that the larger angular errors are in the one mile distance to the target region. Even though the angular TSCT (one-sigma) errors are fairly larger they could still be included in a ±30° cone with its origin at the MAP.

DISTANCE	ONE SIGMA TSCT ERROR QUANTITIES						
TO	Singl	e Beacon	Skin	Paint	Skin Pain	t W/Cursor	
TARGET	nm	Deg	nm	Deg	nm	Deg	
l nm	±.30	±16.5	±.18	10.4	±.16	18.9	
3 n m	±.66	±12.3	+.41	+ 7.9	±.36	+6.9	
5 nm	·.83	± 9.4	±.40	+ 4.5	•.51	+5.8	

Table 5.2 ARA Offshore TSCT Linear and Angular Errors as a Function of Along Track Distance

DISTANCE	ONE SIGMA FTE ERROR QUANTITIES						
TO	Single Beacon		Skin Paint		Skin Paint W/Cursor		
		veg	110	Deg	rini	Deg	
1 nm	+.26	-14.5	+.25	14.0	+.15	+8.4	
3 nm	+.64	+12.1	+.48	+ 9.1	+.08	+1.5	
5 nm	+.84	± 9.5	+.45	+ 5.1	+.25	+2.8	

Table 5.3 ARA Offshore FTE Linear and Angular Errors as a Function of Along Track Distance

The FTE linear and angular values presented in Table 5.3 also show relatively small linear errors. They range in value for the single beacon mode from 4.26 nm at 1 nm to 4.84 at 5 nm. Other linear FTE quantities presented in the table range in value from 4.15 nm to 4.48 nm. The angular FTE values are similar in magnitude to those seen in the TSCT table (i.e. +1.5° to 414.5°). The skin paint with cursor quantities in Table 5.3 show a marked decrease over those skin paint only values presented.

The overall assessment of the ARA operational performance in the offshore environment utilizing the Bendix Radar System was that it was quite acceptable. The skin paint phase of the tesing proved acceptable, but on two occasions the wrong target was tracked down to minimums. This lack of positive intended target identification could cause serious problems as regards airspace requirements. The stated RTCA SC-133 MOPS -4.0 nm airspace requirements are quite adequate in the offshore environment if the proper target can be identified in a repeatable manner. The skin paint with cursor aided approaches showed a decrease in the TSCT and FTE linear and angular error quantities at all ranges. The cursor technique provides the operator with a type of omnidirectional course guidance not offered by the single radar return.

Table 5.4 summarizes the overall Bendix RDR-1400A airport site ARA performance. As in Table 5.1 the table presents four basic error quantities (TSCT, FTE, ASE and ATE) for three specific test areas (single beacon, single beacon with cursor and multiple beacon). Inspection of Table 5.4 chows that the mean and one-store USCL and FTE

ERROR QUANTITY	ERROR MAGNITUDES							
	Single Beacon		Beacon W/Cursor		Multiple Beacon			
	x̄(nm)	±lσ(nm)	x(nm)	$\pm l\sigma(nm)$	x(nm)	±10(nm)		
TSCT	.5725	1.3593	.1520	.5676	2434	. 5986		
FTE	.6279	1.4361	.2574	.7623	0232	.6995		
ASE	0554	.4715	1054	.4863	2202	.3861		
ATE	.1168	.1804	.1970	.1825	.0975	.1416		

Table 5.4 Overall Bendix RDR-1400A Airport ARA Performance Summary

*These results were obtained from Reference 1 (Section 5.3.1).

quantities are considerably larger than those TSCT and FTE quantities calculated for the beacon with cursor and multiple beacon flight test areas. The single beacon TSCT mean presented in Table 5.4 showed a value of .5725 nm with a one-sigma value of 1.3593 nm. The beacon with cursor and multiple beacon TSCT means (.1520 nm and -.2434 nm) showed over a fifty present reduction in magnitude over the single beacon quantities. The FTE quantities indicated virtually the same results. That is, with the use of a particular track orientation technique the FTE values are reduced approximately fifty percent in both the mean and one-sigma values. The ASE and ATE quantities again are small for all of the above mentioned test areas. Typically the ASE values are never greater than -.22 nm and the ATE values are usually less than .20 nm.

Table 5.5 summarizes the airport site TSCT linear and angular errors as a function of distance from the target landing zone. The ARA error magnitudes expressed in this form further substantiate that the approaches flown in the beacon with cursor and multiple beacon modes were more accurate. That is, the TSCT one-sigma data in the single beacon mode was approximately \pm .65 nm while the beacon with cursor data was approximately \pm .25 nm. At one nautical mile the cursor aided approaches showed a marked decrease in the one-sigma angular quantities with the single beacon quantity being \pm 34.8 degrees and the beacon with cursor being \pm 10.6 degrees. The other angular quantities presented in Table 5.5 (at 3 & 5 nm) indicate virtually the same large reduction.

DISTANCE		ONE	SIGMA TSCT ERROR QUANTITIES					
TO	Single	Beacon	Beacon I	N/Cursor	Multipl	e Beacon		
BEACON	nm	Deg	nın	Deg	nm	Deg		
1 nm	+.70	± 34. 8	+.19	±10.6	+.23	+12.8		
3 nm	+.63	±11.9	+.25	+ 4.7	+.40	+ 7.7		
5 nm	+.65	+ 7.4	+ . 31	+ 3.6	+.40	+ 4.5		

Table 5.5ARA Airport TSCT Linear and Angular Errors as a
Function of Along Track Distance

Table 5.6 presents the airport site FTE linear and angular errors as a function of distance to the beacon. The FTE values presented in Table 5.6 are very similar in magnitude to those TSCT values presented in Table 5.5. The single beacon FTE linear one-sigma values range from \cdot .77 nm at 1 nm to \cdot .65 nm at 5 nm. The FTE one-sigma linear values for the beacon with cursor approaches show a marked decrease at 1 nm (\cdot .19 nm) and at 5 nm (\cdot .43 nm). The multiple beacon FTE values are similar in magnitude to those beacon with cursor values presented in Table 5.6 at all range intervals. The angular quantities are again large for the single beacon approaches and comparatively smaller for the beacon with cursor and multiple beacon approaches.

Table 5.6	ARA Airport FTE Linear and Angular Errors as a
	Function of Along (rack Distance

DISTANCE		ONE SIGMA FTE ERROR QUANTIFIES							
TO	Single	Beacon	Leacon	W/Cursor	Multiple Beacon				
BEACON	ทท	Deg	na	Dea I	nm	Deg			
1 nm	+ .77	+37.7	· . 19	10.3	4.25	+14.2			
3 mm	+.67	12.7	1.19 - 19	5.5. ji	· . 41	- 7.9			
5 800	1.66	• 7.1	· _ 47 3	4.9	· . 4·5	$1 \frac{1}{2} \frac{2}{r}$			

In summary, the overall accuracy assessment of the ARA Bendix RDR-1400A airport data showed that the curson aided and multiple beacon approaches offered a decrease in the ISCT and FIE values over the single beacon approaches. The data indicates that the track orientation techniques

virtually eliminated the tendency to "home" to the station. As shown in the skin paint with cursor data the cursor offers an omnidirectional capability. The multiple beacon method offers a different type of track guidance but the track angle error is not directly displayed on the radar screen. The intended course line must be formed as an image by the pilot between the centers of the two displayed beacons. The data indicates that the primary airspace requirements (\pm 4.0 nm at the IAF and \pm 1.7 nm at the MAP) established by the RTCA SC-133 MOPS are more than satisfied by the beacon with cursor and multiple beacon approaches.

The offshore RCA Primus-50 testing was conducted using Brandywine Lighthouse located in Delaware Bay. The tests were performed using the combined and beacon-only modes of operation. Table 5.7 summarizes the overall TSCT, FTE, ASE and ATE errors measured during the offshore tests. Inspection of Table 5.7 results in several conclusions. First, the TSCT one-sigma quantities are very similiar in magnitude between the two different operational modes. The same fact is evident for the FTE values. The combined TSCT mean value is .1561 nm while the beacon-only TSCT mean is .2680 nm. The combined mode FTE is very small (-.0039 nm) and the beacon-only mode value is quite large in comparison (.5587 nm). The ASE values for both test areas are small while the ATE values are somewhat large for this particular radar system. The large errors are possibly the result of the large size of the displayed beacon.

FRROR	ERROR MAGNITUDES							
QUANTITY	Combined Mode		Beacon-o	nly Mode				
	x (nm)	+1σ(n m)	× (nm)	tlo(nm)				
7007	.1561	.4044	.2680	. 4231				
FTE	0039	.6448	. 5587	.6935				
ASE	. 1600	.6360	2908	. 4218				
ATE	. 3587	.2816	. 3955	. 3071				

Table 5.7 Overall RCA Primus-50 Offshore ARA Performance Summary

Table 5.8 illustrates the behavior of both angular and linear errors for the RCA Primus-50 radar system within the five mile distanceto-go area. The table indicates that the linear TSCT and FTE quantities are relatively small at all range intervals. Typical linear combined mode TSCT values range in magnitude from \cdot .20 nm at 1 nm to \cdot .46 nm at

5 nm. The beacon-only mode TSCT linear values are slightly smaller (i.e. \pm .09 to \pm .40 nm). The FTE linear values for both modes of operation range in value from \pm .23 nm to \pm .51 nm. The angular quantities are somewhat large for both operational modes. The combined mode TSCT and FTE linear and angular quantities range in value from \pm 5.3 degrees to \pm 12.9 degrees while the beacon-only mode values are slightly smaller (\pm 4.5° to \pm 8.6°). Outside of five nautical miles the results presented in Section 5.3.5 indicate virtually the same results obtained within five (5) miles.

		TE ERROR QUANTITIES						
DISTANCE		Combine	d Mode			Beacon	only M	ode
TO		TSCT FTE				CT	F	TE
BEACON	ทก	Deg	nm	Deg	nm	Deg	ntu	Deg
] nm	·.20	·11.4	+.23	12.9	+.09	+4.9	+.15	·8.6
3 n m	<u>+</u> .45	· 8.4	+.48	• 9.1	.44	18.3	•.42	·7.9
5 nm	•.46	· 5.3	•.51	+ 5.8	•.40	•4.5	•.43	•4.9

Table 5.8ARA Offshore RCA Primus-50 Linear and Angular Errors
as a Function of Along Track Distance

The RCA Primus-50 radar system performed well during the flight test program. Operationally there is only one significant problem evident from the test results. In the combined mode, because of the large displayed beacon size, surface objects in the immediate area of the intended target are blocked out by the beacon return. This fact will be discussed in further detail in Section 5.1.3.

Data presented in Section 5.3 offers some very interesting conclusions. First, the track orientation techniques evaluated (cursor and rultiple beacon) indicate a significant reduction in the overall TEET and FTE values. Second, without some means of positive target identification in the skin paint mode navigation down to minimums could be dangerous. Trind, if in the skin paint mode, the target is correctly identified then results indicate that this particular mode of operation is just as accurate as single beacon. Fourth, the combined mode of operation must provide a smaller beacon return so that surrounding surface objects can be displayed.

5.1.3 ATC Integration

An operational evaluation of the ARA concept in today's ATC environment was also performed. Section 5.4 discusses this analysis in detail for airport and offshore sites ATC procedures. A summary of this analysis is presented in this section.

The net result of the operational evaluation of the ARA concept in the ATC environment was that it is a practical and viable solution to providing non-precision approach capabilities where other navigation aids are unavailable. This conclusion applies to both approach regions investigated. Although current ARA ground and airborne equipment performed acceptably during these tests, operational utilization in the day-to-day ATC environment would benefit from several ARA system enhancements. First, the ground based equipment could be improved in both signal strength and reliability. Second, more advanced radar features are highly desirable to reduce crew workload and improve the safety of the ARA concept. These improvements are in the areas of automatic gain control and tilt control, variable gain beacons and improved display characteristics. Third, formal crew training procedures and requirements must be developed. With these modifications, the ARA system can provide ATC compatible performance which exceeds the experimental performance at airports and offshore sites.

In the airport area, an ARA approach closely parallels the standard NDB non-precision approach technique in both workload and accuracy. In fact, the approach flown using the track orientation techniques more than satisfied the present day NDB airspace requirements. The experimental airport data (which included single beacon, beacon with cursor and multiple beacon approaches) showed lateral accuracy well within current landside lateral obstacle clearance minima. However, current ARA lateral obstacle clearance limits established by RTCA SC-133 MOPS exceed NDB and other non-precision approach values. Therefore, unless some type of track orientation technique is implemented, either using the cursor technique or multiple beacon technique, minimums higher than present day non-precision approaches might need to be implemented for ARA. Graphic illustrations are presented in Section 5.4 which demonstrate the behavior of ATE, ASE, TSCT and FTE with respect to specific limits. Basically, the ATE and ASE errors are independent of range from the target. On the other hand the single beacon TSCT and FTE errors exhibit the "homing" characteristics previously discussed while the beacon with cursor and multiple beacon TSCT and FTE errors

do not. From an ATC integration viewpoint, higher altitudes and larger lateral obstacle clearance minimums require more airspace. This could limit the areas where ARA would be usable or it could necessitate special approach procedures, such as the point-in-space approaches, to congested terminal areas where adequate approach airspace is not available to the active runway. Airport ARA approaches also require positive navigation when flying to the Initial Approach Fix. This could translate into a requirement for beacons on the ground or some additional airborne navigation equipment (such as the RNAV system used for these tests). Even with these additional navigation aids, multidirectional ARA procedures at airports will require careful planning and a high degree of pilot proficiency to achieve the desired accuracy.

In the offshore ARA tests, three operational ATC considerations are important. First, the ARA system and procedures must provide accurate and repeatable guidance to an IAF in the vicinity of the offshore target. Second, the ARA system must provide adequate guidance in a controlled descent to the specified minimums while providing adequate obstacle clearance. Third, the ARA procedures must provide a simple and safe missed approach procedure in the vicinity of a multitude of prominent surface objects. While the first condition was satisfied during the offshore testing the second and third conditions were not for two reasons. First, during the skin paint testing on two occasions the wrong target was tracked down to minimums. This indicates that without positive target identification, obstacle clearance and safe missed approach procedures cannot be assured. Second, because of the large beacon displayed in the combined mode, obstacle clearance in the immediate area of the intended target cannot be assured. Results presented in Section 5.4 indicate that on the 2.0 nm range selector setting a circle approximately 840 feet in radius is blocked out around the intended target by the large displayed return. Offshore the ARA system is a viable means of accurate navidation but because of the above mentioned items certain important problems must be corrected in certain operational modes before safe navigation can be conducted in a real environment.

5.2 ANALYSIS OF PILOT PROCEDURES

Pilot ARA procedures were developed which resulted in safe approaches with acceptable cockpit workload. These procedures were compatible with current ATC operational constraints in the airport and offshore areas. The pilots experienced a learning curve effect due to the previous single beacon approach testing conducted at the airport, remote and offshore sites (Reference 1). The profiles and procedures utilized during the track orientation testing were virtually identical to those utilized during previous testing. The conceptual aspects of the flight test included utilizing one of two basic flight profiles to perform the approach. These were the direct straight and overhead straight profiles. The use of these flight profiles provided approach procedures that were adaptable to existing meteorological conditions at the landing site.

The emperical aspects of the ARA approach procedures were in the area of pilot/copilot responsibility assignments. It was determined during the single beacon testing that the following ARA crew workload assignents were found acceptable:

- The pilot was given primary responsibility for flight control and safety of flight.
- The pilot was assigned radio communications duties.
- The copilot was solely in charge of ARA navigation.
- The copilot was responsible for interpreting the radar (cursor track angle error and multiple beacon orientation) and communicating required headings, heading changes, altitudes and airspeeds to the pilot.

The crew coordination procedures combined with the flight profile procedures resulted in a calm an ' coordinated cockpit environment. These procedures also insured maximum obstruction clearance while meeting the required test objectives. In summary, the flight procedures were simple, offered a considerable amount of versatility and integrated well with the operational ATC System.

5.2.1 Pilot/Copilot Workload

The pilot/copilot workload involved in flying an Airborne Radar Approach is quite heavy. Based on qualitative observations experienced during the single beacon testing, the Airborne Radar Approach was concluded

definitely to be a two pilot operation. The Airborne Radar System is a good approach aid, but for the pilot to interpret the given information, and to constantly adjust the radar display controls, requires considerably more effort than any other standard non-precision approach using conventional radio navigation aids. For this reason two track orientation techniques were conceived; single beacon and skin paint with cursor, and the multiple beacon technique. It was the intent of the test program to evaluate whether either of these track orientation techniques alleviated workload and improved track keeping. Probably the greatest workload aspect of the airborne radar tested were the constant adjustment required of the gain, tilt, and radar range controls. Gain adjustment was a problem during the multiple beacon testing because the Sensitivity Time Constant (STC) adjustment for this particular radar unit was not set properly. The radar was adjusted and operated adequately during the track orientation technique testing. There were, however, optimum tilt settings that were found to apply during the approaches, e.q., -2 degrees was found to be the optimum setting. Although the -2 degrees tilt setting worked for the entire approach, a constant tilt setting was not necessarily the best recommended procedure for flying the approach. Theoretically the tilt should be adjusted so that the target is scanned within the radar's vertical beam width.

The copilot's workload consisted primarily of three distinct items:

- Radar target interperation in relation to intended course (radar interperation included track orientation utilizing the cursor and multiple beacon techniques).
- Operation of radar display controls.
- Range and heading call-outs to the pilot.

Aside from the track orientation techniques utilized, the radar farget interpretation is strictly empirical for many reasons. First, on the Bendix Radar System with the STAB indicator on the 120 degree setting, there are only azimuth lines every 30 degrees displayed on the radar screen in the BCN mode. However, in the SCH modes azimuth lines are displayed every 15 degrees. Therefore, if the target lies between the azimuth lines the cobilot must be able to interpret bis position relative to zero azimuth accurately. On the Primus-50 radar if the azimuth button is activated, azimuth lines are generated every 15 degrees on the radar screen. Azimuth lines every 15 degrees reduces azimuth interpretation errors, but in the case of the Primus-50 radar system because of the large "blob" that is displayed as the beacon return, azimuth interpretation is again difficult. This large return also makes range interpretation extremely difficult. A more detailed description of the Primus-50 target size will be presented in Section 5.4. Second, the copilot must assume (regardless of the radar system) that the center of the beacon return displayed on the screen is the intended target. In the case of the Primus-50 radar return it must be assumed that the center of the edge closest to the apex is the intended target. The target width analysis presented in Section 5.0 of Reference 1 for the Bendix Radar System showed a mean value of 13.18 degrees and a one-sigma of 4.10 degrees. It is apparent that this large target width or target size would put a limitation on the pilot's interpretive judgement of actual position with relation to the intended course. Another cause for error in display interpretation is the slow update rate of the airborne radar. Five seconds is required for the Bendix system to sweep in one direction. Therefore, aircraft heading could change considerably while the displayed target appears to stay stationary. All of the above factors greatly increase the workload of the copilot, making his position as navigator a full time job.

In the skin paint mode target identification poses a distinct problem for the copilot. Because of numerous targets surrounding the intended target, positive target identification from twenty-five (25) nautical miles was virtually impossible and within ten (10) nautical miles was marginally acceptable. During the skin paint testing correct target identification was a "best guess" situation. Section 5.3.1 shows that on two occasions the copilot identified the wrong target and flew the approach either to a ship or a different lighthouse. On another occasion the copilot tracked the wrong target at the beginning of the approach, but then later identified the correct one and completed the approach. The lack of positive target identification and numerous surrounding targets in the landing area causes a large increase in workload for the copilot in the skin paint mode.

The combined mode which was tested using the RCA Primus-50 offers the ability to receive a ground based transponder and surrounding skin paint targets simultaneously. This mode offers positive target identification

while at the same time affording the pilot obstacle clearance; but because of the large displayed beacon size on the Primus Radar targets, immediately surrounding the beacon are not visible. In the combined mode the beacon target is easily discernable from other targets because it flashes on and off at one second intervals. Again, all of the above factors tend to increase the copilot's mental workload.

The purpose of the cursor and multiple beacon techniques were to aid the copilot in track orientation and to also alleviate the copilot's mental workload. With the cursor technique it was only necessary for the copilot to align the cursor with the center of the return. When this was accomplished the aircraft was on course with the correct amount of drift angle already implemented. The copilot's general reaction during the testing was that the cursor definitely aided in track orientation and that it also reduced the amount of mental workload involved. The multiple beacon testing offered different results as regards workload. Although the multiple beacon technique improved track orientation because of the two targets displayed, the mental workload was greater because the track angle error was not displayed directly on the screen. It was the copilot's responsibility to form an image of the "intended course line" between the targets.

Three other problems were also experienced during the multiple beacon testing. First, because the STC circuit was improperly adjusted on the Bendix Radar System it was necessary to constantly adjust the gain control. Even with constant adjustment, at large longitudinal spacings the beacon for which the gain was not adjusted either disappeared or was splayed across the entire azimuth of the screen. In either case the target was not useable for navigation. Second, at times one of the two ground based beacons was found to be inoperative. This caused a great deal of confusion for the copilot because without discrete positive identification for each particular beacon it was impossible to determine which beacon was the observed target. Third, when tracking the two beacons inbound a range scale change often resulted in the second beacon being beyond the screen display. When the second beacon was lost, obviously track orientation information was not available to the copilot. It is apparent that the multiple beacon testing induced a certain level of mental workload not previously encountered during other areas of testing.

The pilot's workload was quite different from that of the copilot. It was the pilot's sole responsibility to fly the aircraft and handle all Air Traffic Control (ATC) communications. As mentioned earlier the pilot was unhooded throughout the entire flight test. This was done for a particular reason: along with his other responsibilities the pilot needed to be aware and look out for all other traffic in the area. The observer and crew chief onboard the helicopter also aided in watching for traffic, but it was the major responsibility of the pilot to do so. Since the traffic around the NAFEC airport was fairly dense, many times there was a considerable amount of confusion in the pilot/copilot and pilot/ATC communications. Often times this delay in communications from the copilot to the pilot resulted in a deviation from intended course or a delayed correction to reacquire the intended course. Another pilot workload factor introduced during the flight occured on the final approach course. Because of the slow airspeed (50 knots), handling of the helicopter became a problem particularly in crosswind conditions.

Pilot and copilot workload rating sheets were given to each crew member at the end of every flight. The pilots rated themselves on the level of mental and physical effort applied in flying each Airborne Radar Approach. The copilots rated themselves on the level of mental and physical effort applied in navigating and vectoring each Airborne Radar Approach. Table 5.9 summarizes the responses which the pilot and copilot indicated on the workload rating sheets for all of the different flight test areas. The table shows that at no time during the testing did the pilot or copilot consider the workload to be low or very low. It is evident from Table 5.9 that the pilots considered the beacon with cursor approaches to have only a moderate workload, while during the single beacon testing the pilots considered the workload to be either high or very high on nine of the twenty flights. The single beacon approach testing shows slightly higher workload rating because during this phase of testing learning curve effects were still being encountered. Table 5.9 shows that in multiple beacon testing five of the flights were considered to be of moderate workload and three of the flights were considered to be a high workload situation. During the skin paint and skin paint with cursor testing all of the flights were considered to be

Table 5.9 Pilot/Copilot Workload Ratings

		1					[1
	mus-50 sting	Copilot	0	0	3	0	2
Pri	Pri Te	Pilot	0	0	4	l	0
	t W/Cursor	Copilot	0	0	2	Į	0
	Skin Pain	Pilot	0	0	3	0	0
	Paint	Copilot	0	0	3	0	0
TS	Skin	Pilot	0	0	Э	0	0
ER OF FLIGH	le Beacon	Corlot	0	0	2	2	ŋ
MUMB	Multip	Pilot	Q	C	£	-	0
	W/Cursor	Ccrilot	C	0	4	Ċ	C
	Bracon	25105	ر <u>،</u>	Ģ	শ্য	0	C
	Seacor +	Copilot	£ 1	5	۰.r	- C	
	Single	Pilot	Ģ	c-	7	(~)	
	t bû î kirdî. Libû û di		Nor See.	C.	V0.10+358		

See Reference 1 (Section 6.1) for details.

•

•

.

1

ļ

۲

3

of moderate workload with one exception. During the Primus-50 testing seven of the flights were considered to be of moderate workload and three of the flights were considered to be either a high or very high workload situation. Basically the pilots consider the Airborne Radar Approach to be either a moderate or high workload situation, with the track orientation techniques tending to decrease the level of mental workload involved. This fact is evident from Table 5.9.

5.2.2 Pilot/Copilot Blunders

The pilot/copilot blunders encountered during the test were few. One type of blunder encountered was the improper adjustment of display controls. The display control problems encountered were as follows: the scale being changed too soon leaving no target on the radar screen, improper gain adjustment often making the target disappear, and improper adjustment of the tilt control, making close-in navigation difficult. Basically the only other blunders experienced were during the skin paint testing. On two occasions the wrong target was identified and tracked down to approach minimums. On the first instance a lighthouse was tracked and on the second instance a ship was tracked. On the third occasion a ship was tracked in the beginning of the approach until positive target identification was made(for details see Section 5.3.1 and 5.3.2). The following section will describe in detail the accuracy data accumulated during the flight test program. The data is presented so that the reader can make a detailed comparison between various operational modes.

5.3 DETAILED ACCURACY DATA

The purpose of this subsection is to provide insight into the flight test results and data analysis for the Airborne Radar Approach flight test program. The areas of interest involved the RCA Primus-50 and the Bendix RDR-1400A Airborne Radar Systems. The details presented in this subsection represent the results of a comprehensive review of the specific data collected during the Airborne Radar Approach flight test program using new track orientation techniques. The approach data collected involving new track orientation techniques will be compared with the appropriate areas where track orientation techniques were not utilized (Reference 1). The data is presented in four different forms. They are as follows:

- Statistical summary tables showing mean and standard deviation of four error quantities.
- Statistical summaries of data aggregated at one nautical mile intervals along the approach path.
- Plots of Total System Cross Track (TSCT), Flight Technical Error (FTE), and Airborne System Error (ASE), versus along track position of the helicopter.
- Histograms of TSCT, FTE, ASE and Along Track Error (ATE) quantities for each test area.

5.3.1 Offshore Site: Beacon-only and Skin Paint Comparison

This subsection provides a direct comparison between the Bendix RDR-1400A beacon-only and skin paint mode testing conducted at the offshore site. The test environment and approach profiles utilized remained the same offering a means by which a direct comparison could be made.

Table 5.10 summarizes the results of the Airborne Radar Approach testing conducted in the skin paint mode*. The error analysis log and statistical summary of error quantities in the table presents the mean values, standard deviations, number of data points and number of approach segments for four specific error quantities: ARA along track (ATE), ARA cross track (ASE), flight technical error (FTE), and total system cross track (TSCT).

Table 5.10 shows in the ARA ATE case that the calculated mean is .0345 nm and the sigma is .2567 nm for all of the approach segments. The results for the ARA ASE were a total mean value of .1955 nm, and a one-sigma value of .6264 nm. The values were obtained from a sample size of 466 data points or seven (7) approach segments.

Table 5.11 summarizes the results of the Airborne Radar Approach testing conducted at the offshore site in the single beacon mode. The data is presented in the same general format as the data in Table 5.10. The data for this table was collected during the period from October 1978 to December 1978 (see Reference 1 for details). The data was originally aggregated using three different types of approach segments: long, short

*Note: Two approaches were flown to the wrong target. These were not included in the data summaries.

ARA ATE	$\overline{\chi}$ (nm)	σ (nm)	Data Points	Approach Segments	
Long Short Total	.0706 .0225 .0345	.0965 .2901 .2567	116 350 466	1 6 7	
ARA ASE					
Long Short Total	.8343 0163 .1955	.7792 .3770 .6264	116 350 466	1 6 7	
FTE					
Long Short Total	4260 .2230 .0615	.7959 .4726 .6350	116 350 466	1 6 7	
TSCT					{
Long Short Total	.4083 .2068 .2570	.4084 .2982 .3400	116 350 466	1 6 7	
IDENTIFIER		True Hea	ding	Segment	
7/16/79 -2 7/16/79 -3		222 222		Short Short	
7/17/79 AM 7/17/79 AM	-2 -3	222 222		Short Short	•
7/17/79 PM 7/17/79 PM 7/17/79 PM	- 1 - 2 - 3	222 222 222		Long Short Short	

Table 5.10 NAFEC ARA Bendix RDR-1400A <u>Skin Paint</u> Offshore Approaches Error Analysis Log And Statistical Summary

5-21

in. 10

ARA ATE	$\overline{\chi}$ (nm)	σ (nm)	Data Points	Approach Segme	nts
Long Short Total	.0052 0115 .0033	.2502 .1014 .2383	978 123 1101	9 3 12	
ARA ASE					
Long Short Total	2235 0315 2021	.5588 .2235 .5353	978 123 1101	9 3 12	
FTE					
Long Short Total	0844 .4269 0273	1.0456 .6026 1.0185	978 123 1101	9 3 12	
TSCT					
Long Short Total	3079 .3954 2293	.8938 .5081 .8873	978 123 1101	9 3 12	
IDENTIFIER		True Head	ding	Segment	
11/14/78 -	3 Initial	150		Long	
11/15/78 - 11/15/78 -	1 2	222 222		Long Long	
12/12/78 - 12/12/78 - 12/12/78 -	3 4 Initial 4 Final	150 150 330		Long Long Short	
12/13/78 A 12/13/78 A 12/13/78 A 12/13/78 A 12/13/78 A 12/13/78 A	M-1 Initial M-1 Final M-2 Initial M-2 Final M-3 Initial	222 42 227 42 150		Long Short Long Short Long	
12/14/78 -	1	222		Long	

Table 5.11 NAFEC ARA Bendix RDR-1400A <u>Beacon Mode</u> Offshore Approaches Error Analysis Log And Statistical Summary

ALC: NO

and offset, but for purposes of comparison the data was recomputed to reflect only the long and short segments. Table 5.11 shows in the ARA ATE statistics that the total mean value is .0033 nm and the one-sigma value is .2383. The ARA ASE results show a mean value of -.2021 nm and a one-sigma of .5353. The ARA ATE results presented in Table 5.10, show a mean value of .1955 nm and a one-sigma of .6264 nm. In the beacon mode the ARA ATE mean is virtually zero, with the same holding true for the skin paint mode. These data in Table 5.11 were calculated from a sample size of 1101 data points or twelve (12) approach segments.

The Flight Technical Error (FTE) quantities indicated in Table 5.10 showed a mean value of .0615 and a one-sigma of .6350. The FTE values indicated in Table 5.11 for the beacon mode show a smaller mean value of -.0273 nm, with the one-sigma being slightly larger 1.0185 nm for all segments. The larger mean could be attributed to the lack of positive identification available in the skin paint mode as opposed to the beacon mode. This lack of positive identification makes target tracking difficult, with a large increase in pilot workload. The Total System Cross Track (TSCT) values indicated in Table 5.10 for the skin paint mode shows a mean value of .2570 nm and a one-sigma of .3400 nm. Table 5.11 shows a TSCT total mean of -.2293 and a one-sigma of .8873. While the skin paint mean is slightly larger and of a different sign than the beacon mode results, the TSCT one-sigma is smaller in the skin paint mode.

Tables 5.10 and 5.11 indicated that the approaches were aggregated according to two segment types: long and short. The long segments are generally approaches initiated at the 25 nm approach fix. The short segments are generally of two types, at the offshore site they are those approaches initiated at 10 nm while at the airport site they begin at 5 nm.

Tables 5.12 and 5.13 summarizes in statistical quantities the Airborne Radar Approach test data at one nautical mile intervals, starting at ten (10) nautical miles^{*}. The quantities in Table 5.12 reflect data collected in the skin paint mode and the data in Table 5.13 was obtained in the offshore single beacon mode testing (Reference 1). The quantities in both tables were calculated in both linear and angular terms, for both long and short segments. In Table 5.12 at one nautical mile

/NOTE/*In these tables and similiar ones to follow, a single data point was taken from each approach at one nautical mile increments.

Bendix RDR-1400A Skin Paint Mode One Nautical Mile Aggregate Data Table 5.12

2

	· · · · · · · · · · · · · · · · · · ·	LINEAR	ERRORS		UNC	MULAR ERITOR	I
	AFC	TSET	F FL	1951 1951	Tati	л. Н	A9E A9E
-	0.7.7.0.	tII0	-1001	2 7 7 7 7 °	021V	5.7160	-6.2712 MEAN
	•••• 2.16 • 	€4 50 75 75 75 75 75 75 75 75 75 75 75 75 75			<u> </u>	11.0115	
						4.001%	AUGH SVST
	· /- · · ·		105.	. 124 S	0363.4	11.5425	11-12 Oct - 21
					「白田の」や	7. George	-1.5150 MEAN
	∵1 0 ∀ *	<u>.</u> 117.	.45.2		7-8621	9.1410	010 1001 G
•		•	· · · · ·	010+••	C:012	4.2040	1.507× MCAN
	1 . <u>1</u> 71 . •	1.50.	. 4 .		C - 272.7	4.421	4.7511 010
•	· · · · ·		•	: Þ~Þ -	1.6665	906	5.474 MERN
	• • • •	•	.45.02	101 - X -	7 to v	1.1408	115 SZTO-1
	N		11 41 41 2	10,000	5 1 1 1 1 1	· 小小小	3.7070 MEAN
				1011.	5000 to	5.0247	7.3040 STD
	·115?		1.1.1.	.1111	1,9041	2.024	ANDEM SILS.
	0	· •	- 7 S C -	- 6 PQS	2.7100	0.100 °0	1.4 250 6 1D
•) () () () () ()	0 0 0 1 1 1	1.9610	た の で の	L. A SGL MEAN
	1010.		· 4 336	- 7 2 % S	0.000 S	4	115 807T S
•••	· · · · · ·			19 24	1.9207		I. CUSA MEAN
					1.7140	4.74	
1.		· · · ·	5. 19. 19. 19.				2.1001 M.C.M.
	-	0.12.1.1	•••			- 414	1 . C707 .

A NAME AND A DESCRIPTION OF A DESCRIPTIO

5-24

e de la compañía de la

Bendix RDR-1400A Offshore Site Beacon Mode One Nautical Mile Aggregate Data Table 5.13

	⊾4		LINEAR	EKRORS		NG	SULAR CREDR	· · · · · · · · · · · · · · · · · · ·
2::-		j Stri		11	700 V	· · ·	-	
			1000				• . 	
							•	
		• 0.1		1000 ·	、 11 (小) ・	1.1.1		
				100V.	41 11 14 14		•	
	<u></u>					27 14 14 14	1 1 	
		• •			- 1		1 · · ·	
	 	 - -		· · · · · · ·		- 	, , ,	
		• . • • • •			-	•••		
۲		•		•5 • - • - •	いつうせい		· · ·	
			00040. -					
į	۰. سر							NATE STORE
					N. () ₹(1 1	<: -4 00 -9 -9 -9 -9 -9 -9 -9 -9 -9 -9 -9 -9 -9		
• .			1961.			1. C. 'S		
		· • •	ایر میں میں ا		N			
		•		•				
		•		7 - 4 - 4 - 4				
		•		•				
				1		2.202.3		
-		14 A.A.A.A.A.A.A.A.A.A.A.A.A.A.A.A.A.A.A.		1				
			1 d	1 2320	N N	0,000,0		

5-25

1

عىلى <u>م</u>نىلى تە

TY CAA

the skin paint FTE mean is 5.7 degrees and the one-sigma is 14.0 degrees. The beacon mode FTE in Table 5.13 at one nautical mile shows a slightly smaller mean of -1.2 degrees while the one-sigmas are virtually identical, 14.5 degrees in the beacon mode. Other FTE quantities between one and four nautical miles reflect the same orders of magnitudes as numbers previously stated. The skin paint TSCT quantities in Table 5.12 show a mean angular value of -.65 degrees and a one-sigma of 10.4 degrees at one nautical mile. The offshore single beacon TSCT quantities in Table 5.13 show a mean angular quantity of -2.4 degrees and a one-sigma of 16.5 degrees at one nautical mile. The skin paint linear TSCT mean at one nautical mile is -.01 nm and the one-sigma is .18 nm. The skin paint linear TSCT mean quantities past 3 nm stay consistently the same (i.e., between .3 and .4 nm). The linear FTE quantities in the skin paint mode are good close in (between 1 and 2 nauticalmiles) but further out at 3 and 4 nautical miles the mean quantities are .40 and .30 nm respectively. The skin paint ATE quantities are small and consistent at all ranges of operation. The ARA Airborne System Errors (ASE) in the skin paint mode reflect linear quantities that are larger than the beacon mode quantities at most ranges. For example, at ten (10) nautical miles Table 5.12 shows a mean value of .54 nm while Table 5.13 shows a mean of -.26 nm. These larger quantities in the skin paint mode are likely attributable to the numerous targets present on the radar screen causing many targets to run together. This overlap of target display tends to cause confusion on the part of the operator, making track orientation difficult to obtain. This confusion on the part of the operator in the skin paint mode also lends itself to larger FTE quantities. For example, as shown earlier in the skin paint mode at 3 and 4 nautical miles the mean values were .40 and .30 nm respectively while in the beacon mode at 3 and 4 nautical miles the mean values are .07 and -.08 nm, respectively.

Figure 5.1 summarizes in graphical form the Total System Cross Track Error (TSCT) of all the approaches*flown at the offshore site in

/NOTE/*All of the TSCT, FTE and ASE aggregate plots supplied in Section 5.3 reflect the total number of approach segments flown. For example, the overhead straight approach contains two separate approach segments.





and a second second

1

5-27

the skin paint mode. The plot shows that when the target was positively identified the maximum deviation from intended course was only -1.2 nm within five nautical miles. This quantity is well within the required airspace limits of ±1.7 nm at the Missed Approach Point (MAP) set by the RTCA SC-133 MOPS. Outside of five nautical miles the maximum deviation from intended course was -.7 nm which is well within the +4 nm airspace requirements established by RTCA SC-133 MOPS. Figure 5.1 also shows that on two occasions the target was not identified properly and placed the aircraft -1.7 and -5.0 nm off-course at the MAP. Figures 5.2 and 5.3 are individual plots of the above mentioned approaches which reflect TSCT and FTE for each approach.

Figure 5.4 summarizes as in Figure 5.1 the Total System Cross Track Error (TSCT) of all the long and short approaches flown at the offshore site in the beacon mode. Although in some cases the maximum deviation from intended course is greater outside of five nautical miles than those seen in the skin paint mode, within five nautical miles all of the approaches were flown directly to the target because of positive identification. This concept of positive target identification is very important in a real environment where many oil rigs are situated in a cluster with ships moving all about.

Figure 5.5 is a plot of Flight Technical Error (FTE) vs. distance along the desired track in the skin paint mode. These data were collected for all of the approaches flown at the offshore site. These data show a maximum deviation of 4.5 nm at 17.5 nm along the desired track and also a maximum deviation of 2.4 nm at 2.5 nm. These large FTE quantities can be attributed to the pilot not identifing the target correctly and flying to something other then the intended larget (a ship and another lighthouse). This misidentification comes about because of numerous targets displayed on the radar screen due to heavy ship activity surrounding the intended target. Figure 5.6 presents a plot of FTE vs. along track distance for the single beacon approach testing. With the exception of one approach, which was flown on the wrong approach course, at most ranges the majority of the FTE quantities remain within an area of -2.5 nm.

Figure 5.7 is a plot of Airborne System Error (ASE) for the approaches flown at the offshore site in the skin paint mode. Outside of ten (10) nautical miles Figure 5.7 shows a maximum Airborne System Error (ASE)

Section and the second restor

 $S = \mathbb{C} S$

SYSTEMS C AIRBORNE JUN 80 L	ONTROL I RADAR AP D KING	NC (VT) PROACH	FLIGHT	TEST E	VALUATI	NG VAR	IOUS TR	F/G ACK OR- A-4293 NL	17/7 -ETC(U)
				1						
				,						
	SYSTEMS C ALRBORNE JUN 80 L	SYSTEMS CONTROL I AIRBORNE RADAR AP JUN BO L D KING	SYSTEMS CONTROL INC (VI) AIRBORNE RADAR APPROACH JUN 80 L D KING	SYSTEMS CONTROL INC (V1) PALO / AIRBORNE RADAR APPROACH FLIGHT JUN BO L D KING	SYSTEMS CONTROL INC (VT) PALO ALTO CA AIRBORNE RADAR APPROACH FLIGHT TEST E JUN 80 L D KING FAA-	SYSTEMS CONTROL INC (YT) PALO ALTO CA AIRBORNE RADAR APPROACH FLIGHT TEST EVALUATI JUN 80 L D KING FAA-RD-80-6	SYSTEMS CONTROL INC (VT) PALO ALTO CA AIRBORNE RADAR APPROACH FLIGHT TEST EVALUATING VAR JUN 80 L D KING FAA-RD-80-60 FAA-RD-80-60	SYSTEMS CONTROL INC (VT) PALD ALTO CA AIRBORNE RADAR APPROACH FLIGHT TEST EVALUATING VARIOUS TR DOT-FA79M FAA-RD-80-60	SYSTEMS CONTROL INC (V1) PALO ALTO CA AIRBORNE RADAR APPROACH FLIGHT TEST EVALUATING VARIOUS TACK OR DOT-FA79WA-4293 FAA-RD-80-60 Image: Control incomposition of the state of	SYSTEMS CONTROL INC (V1) PALO ALTO CA AIRBORNE RADAR APPROACH FLIGHT TEST EVALUATING VARIOUS TRACK OR-ETC(U DOT-FA79MA-4293 FAA-RD-80-60 NL FAA-RD-80-60 NL I I I I I I I I I I I I I I I I I I I

00717A1

ARA APPRDACH --- OFFSHORE SITE BENDIX RDR-1400A SKIN PAINT MODE

FLIGHT DATE: 7/17/79-1 AM STRAIGHT IN -- 222 DEG. TN



ARA Offshore Site TSCT And FTE Plot: Direct Straight Approach Flown 17 July 1979 Figure 5.2

and a second second second

A CALL A CALL AND A

:



5-30

Ŋ

.

ARA APPROACHES -- OFFSHORE SITE BENDIX RDR-1400A SINGLE BEACON MODE

12 APPROACHES AGGREGATE TSCT



ARA Offshore Single Beacon Approach Total System Cross Track Error Figure 5.4



ARA Offshore Skin Paint Mode Flight Technical Error

Figure 5.5

4

ARA AFPRCACHES --- OFFSHORE SITE BENDIX RDR-1402A SKIN PAINT MONE

J APPROACHES Aggregate fte

DISTANCE (NM) 0 ູ່ , 100 • ណុំ ل ارى ကို 4 ហ 5 My Hundy SIQ

ARA Offshore Single Beacon Mode Flight Technical Error

Figure 5.6

ARA APPROACHES -- OFFSHORE SITE BENDIX RDR-1400A SINGLE BEACON MODE

12 APPROACHES AGGREGATE FTE

ARA APPROACHES -- OFFSHORE SITE BENDIX RDR-1400A SKIN PAINT MODE

9 AFPROACHES AGGREGATE ASE . ທ



Figure 5.7 ARA Offshore Skin Paint Mode Airborne System Error

of -6.6 nm. Between ten (10) and five (5) nautical miles the plot indicates a maximum error of -6.5 nm. Within five (5) nautical miles the plot shows a maximum error of -5.5 nm. All of these large error quantities can be attributed to the approach where the pilot tracked the wrong target (another lighthouse) down to the MAP. Figure 5.8 is a plot of ASE for the single beacon mode. Quantities indicated between ten (10) and five (5) nautical miles show a maximum error of 1.8 nm. Within five (5) nautical miles the plot indicates a maximum error of only 1.0 nm. A CONTRACTOR OF A CONTRACTOR O

Figure 5.9 presents histograms of ARA TSCT, FTE, ASE and ATE for the offshore skin paint mode testing. These histograms represent the error quantity distributions for the four error quantities mentioned previously. The TSCT histogram shows that the quantities appear skewed to the left. The FTE histogram shows the quantities appear to be skewed slightly to the right. The ATE quantities appear skewed to the left, while the ASE distributions appear skewed to the left.

Figure 5.10 presents histograms of ARA TSCT, FTE, ASE and ATE for the offshore single beacon mode testing. The TSCT historgram shows a normal distribution between ± 1.6 nm cross track deviation, but the histogram also indicates a very scattered distribution to -4.0 nm. The FTE distribution basically indicates the same characteristics as the TSCT distribution. The ATE quantities appear skewed slightly to the left, while the ASE quantities also appear skewed slightly to the left.

5.3.2 Offshore Site: Performance Enhancement In The Skin Paint Mode Using The Cursor Technique

This subsection will provide the detailed data necessary to establish an increase in the track keeping abilities of the radar operator using the cursor technique in the skin paint mode. The purpose of the cursor technique was to improve track acquisition and track orientation so that the operator could fly along a predetermined path instead of "homing" to the station. This subsection will show a definite decrease in the TSCT and FTE error quantities on the cursor-aided approaches.

Table 5.14 summarizes the results of the ARA testing conducted in the skin paint with cursor mode. The statistical summary of error quantities in the table presents the mean values, standard deviations,

5-35

1



ARA Offshore Single Beacon Mode Airborne System Error

Figure 5.8

BENDIX RDR-1400A SINGLE BEACON MODE ARA APPROACHES -- OFFSHORE SITE

12 APPROACHES AGGREGATE ASE



ത FTE (CROSS-TRACK) A. DISTANCE UND DISTANCE (NMD ASE 7 NOTE: Data Derived From 12 Approach Segments Ņ ۲ ရာ ņ t Ť 888 100 - 898 894 - 688 883 891 ÷ . 8935 ġ . 266 8 883 STNIO9 90 . STNIOG NO . Ø ത . DISTANCE UND DISTANCE (NH) TSCT ATE 7 7 Ņ Y P **6**) 1 1 1 609 1 . . 189 8 - 889 . 885 89 50 00 00 Ş 8 8 BINIOS NO . STNIOS TO .

1

Bendix RDR-1400A Offshore Site Single Beacon Mode Histograms Figure 5.10



1
404 475	x (nm)	σ (nm)	Data Points	Approach Segments
ARA ATE				
Long Short Total	.1674 0380 .0340	.2883 .3490 .3430	154 285 439	2 6 8
ARA ASE				
Long Short Total	.7193 0348 .2297	.6650 .2869 .5811	154 285 439	2 6 8
FTE				
Long Short Total	0491 0285 0357	.7312 .2529 .4778	154 285 439	2 6 8
TSCT				
Long Short Total	.6702 0634 .1940	.6262 .3290 .5744	154 285 439	2 6 8
IDENTIFIER		True Head	ling	Segment
7/19/79 -1 7/19/79 -2 7/19/79 -3		222 222 222		Long Short Short
7/27/79 -2 7/27/79 -3		222 222		Short Short
7/30/79 -1 7/30/79 -2 7/30/79 -3		222 222 222		Long Short Short

Table 5.14 NAFEC ARA Bendix RDR-1400A <u>Skin Paint With Cursor</u> Offshore Approaches Error Analysis Log And Statistical Summary

5-39

Sector Const

. .

number of data points and number of approach segments for four error quantities: ATE, ASE, FTE and TSCT. Table 5.14 shows in the ARA ATE case that the mean value is .0340 nm and the one-sigma is .3430 nm for all of the data points collected. These values correlate very closely to those ARA ATE values indicated in Table 5.10 (Section 5.3.1). Table 5.10 presented an ARA ATE mean of .0345 nm and a one-sigma of .2567 nm. Table 5.14 shows a mean ARA ASE value of .2297 nm and a onesigma of .5811 nm for all of the approach segments. Once again the ARA ASE values from Table 5.10 are similar in magnitude to those seen for the skin paint with cursor testing (Table 5.14).

The Flight Technical Error (FTE) quantities indicated in Table 5.14 showed a mean value of -.0357 nm and a one-sigma of .4778 nm. The skin paint with cursor FTE is small both in the mean and one-sigma values compared to the non-cursor aided skin paint flights. Table 5.10 shows a mean FTE value of .0615 nm and a one-sigma value of .6350 nm. A reduction of .16 nm was achieved in the one-sigma numbers for all of the approach segments using the cursor technique. The Total System Cross Track (TSCT) error quantities presented in Table 5.14 showed a mean value of .1940 nm and a one-sigma of .5744 nm. These quantities indicate a decrease of .06 nm in the mean value for all of the approach segments (skin paint quantities presented in Table 5.10).

Table 5.15 summarizes in statistical quantities the Airborne Radar Approach Skin Paint with Cursor test data at one nautical mile intervals, starting at ten (10) nautical miles. The quantities are presented in both linear and angular terms. The FTE angular guantities at one nautical mile shows a mean of -4.9 degrees and a one-sigma of 8.4 degrees. The TSCT angular quantities at one nautical mile shows a mean of -3.2 degrees and a one-sigma of 8.9 degrees. Table 5.12 (Section 5.3.1) presented the one nautical mile statistical data for the skin paint mode. The FTE values presented in Table 5.12 shows a mean value of 5.7 degrees and a one-sigma of 14.0 degrees at one nautical mile. The cursor-aided approaches showed a marked decrease of 5.6 degrees in the one-sigma quantities and a decrease of .8 degrees in the mean value. The skin paint with cursor linear TSCT mean at one nautical mile is -.06 nm with a onesigma of .16 nm. The linear FTE quantities presented in Table 5.15 show a mean value of -.09 nm and a one-sigma of .15 nm at one nautical mile.

Bendix RDR-1400A Skin Paint With Cursor One Nautical Mile Aggregate Data Table 5.15

			LINEAR	ERRORS		ANC	BULAR ERROP		-1
Σ	e To	ATE	TSCT		NSE	1301	FTE	ASC ASC	
-	z ^t r	.3164	0220	0855			-4.8001	1.1150	
		1-1-1-V -	.1571	.1483	.1378	0- 17 10- 0- 0-	00°V-0		ацэ Э
64	¢.	• 1 2 2 4	.0020	0317	08.80	.0560	- N - 9000	5174-5 - C	MEAN
		.1450	.2151	.1214	.110.	C-81 "7	3.4722	ゆうごう ゆ	013 8
0	Ç.	0000 · ·	.0126	0244	0740	0440.	4650	1.1126	MEAN
		.4428	.3630	.0805	.0000	6.0004	1.5380	0.0010	57D
4	ņ	N010.	.1351.	. 05.3	0. 100 100	2.0780	7701	1.0000	MCJM
		.4,55	.4100			0.9694	រ ខេត្ត ខេត្ត	0.000 00 00	0 1 0
UT)		-,0035		1010.	.18°°∆	ာ ကို	0600 -	8.1712	MEDN
		0460	1074 1074	. 1400	4704	1204-0	1040 1040 1040	0. 1. 0. 2.	01-33 73
÷	Ģ	.02120.	1961.	.0716	.1040	1.8748	. 6340		MEAN
		.0310	. 5602	5024	. U	く) (4) (1) (1)	4.8130		11 11 10
~	Ċ	0380.	1000.	.1217	.1127		5966.	.9224	MEAN
		.0526	1699.	.5240	.7272	0.6629	4.2810	0000 0000	5 1 D
0)	e	0) () (.2710			1.9000		0.007	N N
		- 14 C C	. 6766	. 2965	.000.	4,804.1	2.1203	1. 1044	11-3 21-3
2	ŀ.	.0801	5100		0 0 0 0 0 0	े २०८ <i>०</i>	-2026	N. 400V	MEAN
		2007 1	1000	1313.	. 8050	0.102	1.0000	0017-0	erb
10	anje.			. : 1.1	0100 .	1011 1011		2.2416	12/12/12
		.1072	.7140	. Seol	.680.	4,0390	0.0000	0.8904	61D

The skin paint linear FTE values presented in Table 5.12 show a mean value of .10 nm and a one-sigma of .25 nm resulting in a decrease of .10 nm for the one-sigma values using the cursor aided techniques. Table 5.15 shows a mean TSCT of .00 nm and a one-sigma of .22 nm at two nautical miles. TSCT mean and one-sigma values indicated at two nautical miles for the skin paint mode testing are .14 nm and .34 nm, respectively. These values show a decrease of .14 nm in the mean value and a decrease of .12 nm in the one-sigma value. As before the ATE linear values indicated in Table 5.15 are consistently small for all ranges.

Figure 5.11 summarizes in graphical form the Total System Cross Track Error (TSCT) of all the approaches flown at the offshore site in the skin paint with cursor mode. The plot shows that outside of ten (10) nautical miles the maximum deviation from intended course is -1.2 nm. Between ten (10) and five (5) nautical miles the maximum error indicated is -1.6 nm and within five (5) nautical miles the maximum error shown is -.8 nm. The quantities between ten (10) and five (5) miles are well within the ± 4.0 nm route width established by RTCA SC-133. Within five (5) nautical miles the maximum value of .8 nm is within the ± 1.7 nm airspace requirement established by SC-133.

Figure 5.12 summarizes in plot form the FTE of all the approaches flown at the offshore site in the skin paint with cursor mode. The large quantities of 6.0 nm outside of fifteen (15) nautical miles stem from an approach where the wrong target was tracked at the beginning of the approach. Figure 5.13 presents a plot of TSCT and FTE vs. along track distance which shows the approach where the wrong target was tracked until approximately eighteen (18) nautical miles from the intended target. Between ten (10) and five (5) nautical miles the maximum deviation in error is -1.5 nm and within five (5) nautical miles the maximum deviation in error is -.5 nm. These values show marked improvement over those mentioned in Figure 5.5 (Section 5.3.1). With the exception of the two approaches flown to the wrong target the figure indicates a maximum deviation of -1.2 nm in the skin paint mode, which indicates a .7 nm improvement in the 5 nm area due to the cursor technique.





ARA Offshore Skin Paint With Cursor Mode Flight Technical Error

Figure 5.12

BENULX RES-14206 SKIN PAINT MUCHANCE MODE

ANA APPRCADHED -- GEFCHORE SITE

0072719







TSCT FTE

DASHED LINE:

SOLID LINE:



5-45

DISTANCE (NM)

8

ų

ų.

ហ៊ុ

លំ

. m

ហ

Figure 5.14 is a graphical representation of the offshore site skin paint with cursor Airborne System Error (ASE). Again because the wrong target was tracked on one approach outside of ten (10) nautical miles the maximum error quantity indicated is -6.1 nm. Between ten (10) and five (5) nautical miles the maximum error is -2.2 nm. Within five (5) nautical miles the maximum error shown is only -.7 nm. The Airborne System Error once more is consistently small, which indicates the airborne radar is reliable and offers good repeatability.

Figure 5.15 presents histograms of ARA TSCT, FTE, ASE and ATE for the skin paint with cursor approaches flown at the offshore site. Only data within 10 nm of the target is included. These histograms represent the error quantity distributions for the four ARA error quantities. The TSCT histogram shows that the quantities appear skewed to the right. The FTE histogram appears to be normal. The ASE distribution appears skewed slightly to the left while the ATE quantities appear skewed to the right.

5.3.3 Airport Site: Performance Enhancement In The Beacon Mode Using The Cursor Technique

The heading error cursor technique is based on a minor airborne system modification (Section 4.1.1). The technique works equally well in both the landside and offshore environments. As mentioned in Section 5.3.1 the skin paint cursor aided approaches showed a decrease in the Total System Cross Track errors (TSCT) and Flight Technical Error (FTE) quantities. The intent of this subsection is to show that a marked decrease is also present in cursor aided beacon mode TSCT and FTE quantities.

Table 5.16 summarizes the results of the Airborne Radar Approach testing conducted at the airport site in the beacon with cursor mode. The airport site utilized the NAFEC terminal area with the beacon placed on the threshold of runway 26. An extended centerline of the runway offered the approach course to the beacon. This table is similar to the ones presented in Sections 5.3.1 and 5.3.2, and summarizes the mean values, standard deviations and the number of data points in four specific areas.

Table 5.16 indicates a mean ARA ATE quantity of .1970 nm and a onesigma of .1825 nm. The results of the ARA ASE quantities shows a mean value of -.1054 nm and a one-sigma of .4863 nm. These data were calculated from a sample size of 952 data points or 17 approach segments.



_

თ ø, FTE (CROSS-TRACK) ณ ណ DISTANCE (NM) DISTANCE (NM) ASE 60 7 7 Data Derived From 8 Approach Segments សុ പ്പ ກ ່ ရာ • • • 4 188-1 5887 - 881 688-200-588 - 986 288-- 883 488-- 888 188 -STNIO9 30 . STNIOS 30 . თ ന ល ស NOTE: DISTANCE (NM) DISTANCE (NM) TSCT ATE 0 8 ÷ ---**-**ណ ដ ę က ၊ + + ∞ i -+ + N 1 6887 6887 188-- 983 1881 288 1881 288-- 888 588-986 +88 • OF POINTS STNION HO .

Figure 5.15 Bendix RDR-1400A Skin Paint With Cursor Mode Histograms

Same and the second

	x (nm)	σ (nm)	Data Points	Approach Segments
ARA ATE				
Long Short Total	.1724 .2271 .1970	.1548 .2077 .1825	524 428 952	5 12 17
ARA ASE				
Long Short Total	2079 .0200 1054	.6115 .2002 .4863	524 428 952	5 12 17
FTE				
Long Short Total	.4744 0083 .2574	.9320 .3187 .7623	524 428 952	5 12 17
TSCT				
Long Short Total	.2665 .0118 .1520	.7194 .2186 .5676	524 428 952	5 12 17
IDENTIFIER		RWY	True Heading	Segment
8/2/79 AM-1 8/2/79 AM-2 8/2/79 AM-3 8/2/79 AM-4		26 26 08 08	252 252 72 72	Long Short Long Short
8/2/79 AM-5 8/2/79 AM-5 8/2/79 AM-6 8/2/79 AM-6 8/2/79 AM-7	Initial Final Initial Final Final	26 08 26 08 26	252 72 252 72 252	Long Short Short Short Short
8/2/79 PM-1 8/2/79 PM-2 8/2/79 PM-2 8/2/79 PM-3 8/2/79 PM-3 8/2/79 PM-4 8/2/79 PM-4	Initial Final Initial Final Initial Final	26 26 08 08 26 08 26	252 252 72 72 252 72 252 72 252	Long Short Short Short Short Short Short
8/6/79 -1		08	72	Long

Table 5.16NAFEC ARA Bendix RDR-1400A Beacon With Cursor
Airport Approaches Error Analysis Log And
Statistical Summary

5-49

4.3 1

Table 5.17 presents the results of the ARA testing at the airport site in the single beacon mode (Section 5.3.1, Reference 1). The error analysis log and statistical summary reflect only the long and short segments so that a direct comparison could be established. The offset segments were omitted from the totals because there were no offset approaches flown during the beacon with cursor testing.

Table 5.17 shows a mean ARA ATE value of .1168 nm and a one-sigma of .1804 nm for all of the approach segments flown. The ARA ASE quantities shows a mean and one-sigma value of -.0554 and .4715 nm, respectively. These values compare quite favorably with the ATE and ASE values presented in Table 5.16. In fact, for the Bendix Radar System, in general regardless of the operational mode the Airborne System errors have been well within acceptable limits.

The FTE quantities indicated in Table 5.16 (beacon with cursor) shows a total mean value of .2574 nm and a one-sigma of .7623 nm. Table 5.17 shows that the total mean and one-sigma quantities are .6279 nm and 1.4361 nm. respectively. A reduction of .37 nm in the mean quantity and .67 nm in the one-sigma quantity is evident from these results using the cursor technique during the approach. The cursor technique offers the pilot an immediate indication of track angle error. With this the pilot can determine his drift angle and course orientation. The cursor eliminates the "second-guessing" involved in determining track orientation, therefore, reducing the tendency to "home" to the station. In effect it gives the radar "omnidirectional" capabilities.

The TSCT quantities presented in Table 5.16 showed a mean value of .1520 nm and a one-sigma of .5676 nm for all of the approach segments flown. Table 5.17 indicates a total mean TSCT quantity of .5725 nm and a one-sigma of 1.3593 nm. The Total System Cross Track errors were reduced by .42 nm in the mean value and by .79 nm in the one-sigma quantity. The procedure turn executed during the single beacon testing induced some initial errors (see Reference 1 for details), but the pilot still had a tendency to "home" to the station without the aid of the cursor. Later, in this subsection a case will be presented where the aircraft was intentionally offset to a predetermined position so as to duplicate the initial conditions given to the pilot in an earlier approach. A direct comparison of the approaches will show that with the aid of the cursor the pilot was able to acquire the intended track in a short period of time and fly directly to the beacon.

	x (nm)	σ (nm)	Data Points	Approach Segments
ARA ATE				
Long Short Total	.1209 .1131 .1168	.2349 .1112 .1804	263 294 557	4 7 11
ARA ASE				
Long Short Total	.0168 1200 0554	.6336 .2324 .4715	263 294 557	4 7 11
FTE				
Long Short Total	.8908 .3927 .6279	1.9144 .7203 1.4361	263 294 557	4 7 11
TSCT				
Long Short Total	.9076 .2727 .5725	1.8025 .6402 1.3593	263 294 557	4 7 11
IDENTIFIED		RWY	True Heading	Segment
11/03/78 -1 11/03/78 -3	Initial Initial	26 08	253 073	Long Long
12/13/78 PM- 12/13/78 PM- 12/13/78 PM- 12/13/78 PM- 12/13/78 PM- 12/13/78 PM- 12/13/78 PM-	Initial Final I Final Initial Initial Initial Initial	26 08 26 08 08 26 08 26	253 073 253 073 073 253 073	Long Short Short Short Long Short Short
12/14/78 -4 12/14/78 -5	Final	26 26	253 253	Short Short

Table 5.17 NAFEC ARA Bendix RDR-1400A <u>Beacon Mode</u> Airport Approaches Error Analysis Log And Statistical Summary

5-51

Sec.

Tables 5.18 and 5.19 summarizes in statistical quantities the Airborne Radar Approach test data at one nautical mile intervals, starting at ten (10) nautical miles. The quantities in Table 5.18 reflect data collected in the single beacon with cursor mode and the data in Table 5.19 shows data obtained in the airport single beacon mode testing. The quantities in both tables were collected in both linear and angular terms, for both long and short segments. The angular FTE beacon with cursor quantities presented in Table 5.18 showed a mean value of -2.3 degrees and a one-sigma of 10.9 degrees at one nautical mile. The angular mean and one-sigma TSCT beacon with cursor quantities at one nautical mile are -.33 degrees and 10.6 degrees, respectively. At one nautical mile in Table 5.19 the FTE single beacon mean is 4.3 degrees and the one-sigma value is 37.7 degrees. The large one-sigma FTE can be attributed to the fact that on three of the approaches the pilot missed the target by .8 of a nautical mile. The TSCT angular quantities in the single beacon mode at one nautical mile shows a mean value of 4.0 degrees and a onesigma of 34.8 degrees. The FTE and TSCT angular quantities indicated in Table 5.18 are considerably smaller than those presented in Table 5.19, between one and three nautical miles for both the mean and one-sigma values. The smaller angular quantities indicated in Table 5.18 once again prove that the cursor technique affords the pilot the ability to track to the beacon with good accuracy and repeatability.

The linear error quantities in Tables 5.18 shows a mean TSCT value of .00 nm and a one-sigma of .19 nm at one nautical mile. The FTE beacon with cursor linear quantities indicates a mean value of -.04 nm and a one-sigma of .19 nm, again at one nautical mile. Table 5.19 shows an increase in the TSCT and FTE linear quantities, where the TSCT mean and one-sigma values are .07 nm and .70 nm, respectively, and the FTE mean and one-sigma value are .08 nm and .77 nm, respectively. At seven (7) nautical miles Table 5.19 shows a mean TSCT linear quantity of .40 nm and a one-sigma of 1.26 nm. The FTE quantities at seven (7) nautical miles in Table 5.19 indicates a mean value of .50 nm and a one-sigma of 1.30 nm. Other linear TSCT and FTE beacon with cursor quantities presented in Table 5.18 show that the quantities are consistently small. The ATE and ASE quantities presented in both tables are consistently small at all range intervals.

Bendix RDR-1400A Beacon With Cursor One Nautical Mile Aggregate Data Table 5.18

	-		LINEAR	ERRORG			AU AR ERROR	I81
42	ं निम	ATE		5 F F			یں بر ا	4 - 4 -
	 •							
					-			NUBE CREAK
		•					しんしょう	7.0000 010
	<u>.</u>		.0110.	1040.1	0120.		C1991	
			C 1 - 1 - 1			1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1		
			1. J.C. 1. 1					
	•) (, , ,) (/ 		1.0.14	1000 ·	.1000 11201
,		-	•	1001.	.1194	1901 5	0 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	11.10.20 10.10 10.10
•.	· . 4	•				1.1451	76:27	INV DAL VEGE
			i) • •	• • • •	() () ()	C S	1701	
70	+-							
						-		CONTRACTOR -
		- - - -					70N W . 77	コード ひりょうしい
s.!	2				N, 14 19 19 10			
		114	्र 	0000 ·	.4146	7 1 00 - 0	1 6012	
^ .	V)			0'270				
		1000					- 804 I	
	1		· • • • •	•			1.2676	4.5500 STD
	•	•		.4000	1012.00			
				6.250	.5724	0 7. 2	1 V	
	•		.1.0:03					
			C L . 1 . 1					Tet 21 2 2 400 T
() 	5		•	•				
	-							
		/:01.		1873	1.0422	2 0 0 0	1.25	

1. N. W.

5-53

1.00

Bendix RDR-1400A Airport Site Beacon Mode One Nautical Mile Aggregate Data Table 5.19

31.6

			LINEAR	LKRORS ······		INU]	JULAR EFRUR	
۲ Z) Ú.	: 	1.151		Case Case		1	1999 - 1999 1997 - 1997 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1
			10.20.		-,0004			
		10011	3367.		.1007			
1	ہ . ہے	• 5 9-11 - 4 • 4		 	1200			
			P357.	00 10 20 10	. 102.	10°11°0-	です。 1991年 1991年	
ēŋ.			: 101.			•		
			0.620 *	.6745.	0.2	14 17 19 19		
	• • • •		1010 .	\$770.			-000 - -	
				.7172				
£ î			. 27.7 .			2000 		
			5385.	िल्ल मार्थ भ				
.'			.1.00			1.6000		
		Conta -	2120.	7130 · 5	1202.	0 0 0 0		
Ņ	, ¹		12764	. 1075		0:40.0	1.0425	
		•		1.0004		1977 - OF		
ф)			1,200.1	- 1770 ·	0110 .	大学の 1	.1140	
			. 2170	のとうとい				<pre>{</pre>
.`	÷.	 	, 1144	, CAS .		20 40 10 10 10 10 10 10 10 10 10 10 10 10 10		
				5 C C S 4				
_			. 4444	1000 E				
			1130Z."		+			

Table 5.20 summarizes the mean and one-sigma Letdown Error (LDE) quantities obtained during the single beacon with cursor approach testing conducted at the airport site. These values quantify the ability of the pilot to utilize the ARA system to define and identify a stepdown fix. Table 5.20 represents values sampled at the 5.0 nautical mile Initial Approach Fix (IAF) and at the 2.0 nautical mile Final Approach Fix (FAF). The profiles shown earlier in Section 4.2 indicate that at the IAF the pilot initiated a descent from 1000 feet to 500 feet and at the FAF he initiated a descent from 500 feet to 200 feet. The error quantities were determined by correlating the airborne radar's indicated approach fix position with time. Next, using the EAIR tracking data printout it was then determined when a descent was actually initiated and this time was noted. Then this time was correlated with the airborne data and the indicated beacon position noted. The difference between displayed distance and the prescribed letdown point was then computed. (Distance to the target was computed as positive). The mean value at the IAF shows that the pilot initiated his descent .09 nm before passing the fix. The one-sigma value at the IAF was .54 nm. The mean value at the FAF shows that the pilot started his descent .09 nm before passing the fix with a one-sigma value of .22 nm. It is interesting to note that the values are similiar in magnitude and of the same sign, indicating that at the IAF and FAF the pilot anticipated his position. The single beacon airport LDE quantities (Reference 1 - Section 5.3.1) showed that at the IAF the pilot started his descent .36 nm after passing the fix. At the FAF the pilot started a descent .18 nm after passing the fix in the airport single beacon testing. It should be noted that during the single beacon testing the FAF was at 1.5 nm instead of 2.0 nm. This change was implemented so that the pilot could have more time to establish his final airspeed (50 Kts.), altitude (200') and crab angle. Since during the cursor testing the pilot started his descent slightly before reaching the fix this indicates that the decrease in workload afforded by using the cursor allowed the pilot to concentrate more on his along track position and utilize the radar for more accurate approach fix identification.

Approach	Error Ma	ignitudes
Position	Mean nm	+]., nm
IAF (5.0 nm Fix)	0911	.5368
FAF (2.0 nm Fix)	0908	.2207

Table 5.20 Beacon With Cursor Letdown Error Quantities

Figure 5.16 summarizes in graphical form the Total System Error of all the approaches flown at the airport site in the beacon with cursor mode. The plot shows that outside of ten (10) nautical miles the maximum deviation from intended track is -4.2 nm. This approach has a large initial error because on this particular approach the aircraft was initially intentionally offset from the intended track. Figure 5.17 presents the approach which was flown in 3 November 1978 during the single beacon approach testing at NAFEC. The initial approach position of this approach was computed so that the aircraft could be vectored to this position by ATC. From this position the approach was initiated using the cursor technique, to prove that when the cursor is utilized the tendency to "home" to the station is eliminated. Figure 5.17 shows that without the aid of the cursor the pilot had difficulity in acquiring the intended track and tracking inbound along the desired course. Figure 5.18 shows that with the aid of the cursor the pilot quickly identified his relative course position and then flew directly towards the intended course and tracked the beacon inbound with good precision. The segments presented in Figure 5.17 and 5.18 are initial segments of the overhead straight procedure.

Between ten (10) and five (5) nautical miles Figure 5.16 shows a maximum deviation from intended course of -1.0 nm. This quantity is well within the four (4) nautical mile airspace requirement established by the RTCA SC-133 MOPS. Inside of five (5) nautical miles the maximum deviation from intended course is -.8 nm, which again is within the required airspace limits of +1.7 nm at the Missed Approach Point (MAP) set by the RTCA SC-133 MOPS.



10.00





5-59

1

Figure 5.19 summarizes in graphical form the Total System Error of all the approaches flown in the single beacon mode at the airport site. The plot shows that between five (5) and ten (10) nautical miles from the beacon the maximum deviation from intended track is -2.5 nm. This deviation is 1.5 miles greater than -1.0 nm deviation shown in the same range interval in Figure 5.16, utilizing the cursor technique. Within five (5) nautical miles the maximum deviation from intended course is -1.3 nm, which is .5 nautical miles greater than the beacon with cursor plot indicates within five (5) miles. All of the approaches in Figure 5.19 indicate a tendency to "home" to the station whereas Figure 5.16 shows no such tendencies.

Figures 5.20 and 5.21 are plots of Flight Technical Error (FTE) vs. distance along the desired track, for all the approaches flown at the airport site in the single beacon and single beacon with cursor modes. Figure 5.20 shows that outside of ten (10) nautical miles the largest FTE beacon with cursor quantity indicated is -2.0 nm. The large -7.5 nm quantity at 22 nm is a result of the <u>intentional</u> offset approach executed during this phase of testing. Figure 5.21 shows numerous large error quantities outside of ten (10) nautical miles. These large FTE single beacon error quantities can be attributed to two things; inadequate initial course acquisition procedures and the pilot having a tendency to "home" to the station. Between five (5) and ten (10) nautical miles the maximum error indicated in Figure 5.20 is -1.7 nm and within five (5) miles the maximum FTE beacon with cursor error shown is -1.5 nm.

Figures 5.22 and 5.23 are plots of ARA Airborne System Error (ASE). Figure 5.22 presents the aggregate ASE quantities for the single beacon with cursor approach testing conducted at the airport site. Figure 5.23 presents the aggregate ASE quantities for the single beacon testing conducted during the period from October 1978 to December 1978 at the airport site (Reference 1). The ASE quantities shown in both figures are relatively small in comparison to the TSCT and FTE quantities shown in earlier figures. Most of the Airborne System Errors lie within a +2 nm region. Large spikes indicated on both plots are due to a rapid change in aircraft heading without the radar display having time to update. As mentioned earlier the ASE quantities for all areas of testing are quite small, proving the radar system offers good accuracy and repeatability.





ARA APPROACHES -- AIRPORT SITE BENDIX RDR-1400A BEACON W/CURSOR MODE

AGGREGATE FTE

17 APPROACHES

ARA Airport Single Beacon With Cursor Approach Flight Technical Error Figure 5.20



ARA Airport Single Beacon Approach Flight Technical Error

Figure 5.21

ARA APPROACHES -- AIRPORT SITE BENDIX RDR-1400A SINGLE BEACON MODE

11 APPROACHES AGGREGATE FTE



ARA Airport Single Beacon With Cursor Approach Airborne System Error Figure 5.22

BENDIX RDR-1400A BEACON W/CURSOR MODE ARA APPROACHES -- AIRPORT SITE

ŀ

17 APPROACHES

ARA APPROACHES -- AIRPORT SITE BENDIX RDR-1400A SINGLE BEACON MODE

11 APPROACHES AGGREGATE ASE . נו



Figure 5.23 ARA Airport Single Beacon Approach Airborne System Error

1

.....

Figure 5.24 presents histograms of ARA TSCT, FTE, ASE and ATE for all of the single beacon with cursor approaches flown at the airport site. Only data within ten (10) nautical miles of the beacon is included. The FTE and TSCT distributions appear almost normal although both show a slight skewness to the right. The ASE quantities appear normal while the ATE quantities appear skewed to the right. Figure 5.25 presents histograms of ARA TSCT, FTE, ASE and ATE quantities in the single beacon mode. Again the TSCT appears normal, but the FTE quantities appear skewed slightly to the right. The ASE distribution appears to be skewed to the left and the ATE distribution appears skewed to the right.

5.3.4 <u>Airport Site: Performance Enhancement In The Beacon Mode Using</u> <u>The Multiple Beacon Technique</u>

The multiple beacon testing was conducted in the NAFEC terminal area utilizing two longitudinally spaced beacons. The beacon spacing was entered as a control variable in the testing with the first or near beacon being placed on the runway threshold. The near beacon was used as the pilot's primary means of navigation while the second beacon was utilized for track orientation purposes. Although not known during the test period it was later discovered that the radar system STC was not properly adjusted. This maladjustment caused increased workload for the pilot because frequent gain adjustments were necessary and quite often the beacon return was splayed over the entire azimuth of the screen making navigation impossible. As mentioned in Section 4.2 numerous flights were flown but only those approaches where the splaying was considered to be at a minimum were recovered. Also, as mentioned earlier on two flights flown on 2 February 1979 a 3dB attenuator was utilized on the near beacon to help reduce the splaying experienced earlier in the testing. Prior to attenuation, when the gain was adjusted to make one beacon a reasonable size the other beacon became either too large in azimuth to use for navigation or disappeared completely.

Table 5.21 summarizes the results of the ARA testing conducted in the multiple beacon mode at the airport site. The statistical summary of error quantities in the table presents the mean values, standard deviations, number of data points and number of approach segments for four error quantities: ATE, ASE, FTE and TSCT. Table 5.21 shows in the ARA ATE case that the mean value is .0975 nm and the one-sigma value is







NOTE: Data Derived From 11 Approach Segments

Figure 5.25 Bendix RDR-1400A Airport Site Single Beacon Mode Histograms

	x (nm)	σ (nm)	Data Points	Approach Segments
ARA ATE				
Long Short Total	.0603 .1163 .0975	.1771 .1154 .1416	391 775 1166	4 11 15
ARA ASE				
Long Short Total	3476 1559 2202	.4180 .3520 .3861	391 775 1166	4 11 15
FTE				
Long Short Total	.1011 0859 0232	.9930 .4774 .6995	391 775 1166	4 11 15
TSCT				
Long Short Total	2466 2418 2434	.8685 .3991 .5986	391 775 1166	4 11 15
IDENTIFIER		RWY	True Heading	Segment
1/19/79 -1 1/19/79 -2 1/19/79 -3 1/19/79 -4		13 13 13 13	118 118 118 118 118	Long Short Short Short
2/5/79 AM-1 2/5/79 AM-2 2/5/79 AM-3 2/5/79 AM-4		13 13 13 13	118 118 118 118 118	Long Short Short Short
2/5/79 PM-1 2/5/79 PM-2 2/5/79 PM-3 2/5/79 PM-4		13 13 13 13	118 118 118 118 118	Long Short Short Short
8/7/79 -1 8/7/79 -2 8/7/79 -3		13 13 13	118 118 118	Long Short Short

Table 5.21	NAFEC ARA Bendix RDR-1400A Multiple Beacon
	Airport Approaches Error Analysis Log And
	Statistical Summary

5-69

ć

.1416 nm for all of the data points collected. These values correlate closely to those ATE values indicated in the airport signle beacon testing (Table 5.17 - Section 5.3.3). Table 5.17 presented an ARA ATE mean of .1168 nm and a one-sigma of .1804 nm. Table 5.21 shows a mean and one-sigma value in the ASE case of -.2202 nm and .3861 nm, respectively. Table 5.17 showed a mean ARA ASE mean value of -.0554 nm and a one-sigma of .4715 nm. The multiple beacon ARA ASE values are somewhat larger than those previously seen in other test areas because of the beacon splaying encountered during the multiple beacon testing.

The Flight Technical Error (FTE) quantities indicated in Table 5.21 showed a total mean value of -.0232 nm and a one-sigma of .6995 nm. These results were obtained from a very large sample size (1166 data points). Table 5.17 showed a mean total FTE quantity of .6279 nm and a one-sigma of 1.4361 nm. These Multiple Beacon Approach quantities indicate a reduction of .6 nm in the mean value and a reduction of .74 nm in the one-sigma value. The TSCT error quantities presented in Table 5.21 showed a mean value of -.2434 nm and a one-sigma of .5986 nm. These quantities indicate a decrease of .33 nm in the mean value for all of the approach segments when compared to the single beacon quantities presented in Table 5.17.

į.

Table 5.22 summarizes in statistical quantities the Airborne Radar Approach Multiple Beacon test data at one nautical mile intervals, starting at ten (10) nautical miles. As before the quantities are presented in both linear and angular terms. The FTE angular quantities at one nautical mile show a mean of -.21 degrees and a one-sigma of 14.2 degrees. The TSCT angular quantities at one nautical mile show a mean value of .25 degrees and a one-sigma value of 12.8 degrees. Table 5.19 (Section 5.3.3) presented the one nautical mile statistical data for the single beacon mode testing. The multiple beacon data showed a marked decrease of 23.5 degrees in the one-sigma quantity and a decrease of 4.1 degrees in the mean value. The multiple beacon linear FTE quantities in Table 5.22 show a mean value of -.0037 nm and a one-sigma of .2532 nm at one nautical mile. The one nautical mile linear TSCT quantities show a mean value of .0043 nm and a one-sigma of .2276 nm. Both ISCT and FIL means in Table 5.22 at one nautical mile are virtually zero as opposed to those TSCT and FTE means shown in Table 5.19 where the quantities indicated

Bendix RDR-1400A Multiple Beacon One Nautical Mile Aggregate Data Table 5.22

	-		LINEAR	ERRORS			AURA LARUE	108
Σ	a tu ta	111		<u></u> 			در) ۲-	
4		1001.	(1999) 1999 - 1999 1999 - 1999	Z = (; ``+ • -	• • •			. 1586 MEA :
		• 634 E.			а 900 е		11, 200	110 STOT 1
74	 • ·						1.200	- COOP MEAN
		- 00 i	0.00 1	. 2378		0.120) () ()	1.4572 610
en.	51			1002		× 205.•C-	921.11-	-1.0054 MEAN
		.0702	うょうり・	.4145		1. 6000	7. OGBO	2.0440 JTD
4	<u>r</u> 1		-1221-	0610		-:00°•:-	1904	-2.3490 MEAN
		0100°.	.4056	0100 ·	- 1 () () - 1	11:00 ° 1	0.00k0	
53	[] 	.1303	9100 -	1164	0 1 1 1 1	-3.4067	-1-9470	HUBW STOP I-
		.1002	1202.	. 4000		9.5468	5.1874	2.6020 STD
Ų	ु ँ। स्त	.1180	10000 ··	6979	1256	-3.5110	100 Tel - 14	-1.1993 MEAN
		.1362	0 4 00.	.4507	다. 이 아 아	3.6710	4.2962	I.2069 STD
r.	1.0	.0540	4275	- 1963	00 00 01 01 1	-0.4951	-1.6266	-1.0717 MEAN
		0000 ·	2000.	5050°	.3500	1) 60 00	0.4965	113 SISE 510
en.	נ"ו 	.0810	4077	5100 1	0000 000 1	-0.4800	-1.6042	-1.8072 MEAN
		.1217	.4106	.4516	0428.	3.0015	3.2300	2.6770 570
.		.2036	3790	[717	1071	+3.6810	-1.0927	-10.6010 MEAN
			00000 00000	.7054	- 4 900	7401	4.4010	0.100×010
਼	<u>)</u> .	.1111		1900	1997	- 2.4007	. 0193	1.1500 MEAN
		.154;	.6043	.4445		G.4581	2.5150	010 0101.1

1

4

ĵ.

.

are .07 nm and .08 nm, respectively. The FTE quantities indicated in Table 5.22 are consistently small at all range intervals out to ten (10) nautical miles. The linear multiple beacon TSCT values are also small but tend to increase as the distance from the beacon increases. For example, the linear TSCT mean value at ten (10) nautical miles is -.63 nm and the one-sigma value is .60 nm. The linear FTE and TSCT quantities indicated in Table 5.19 for the beacon only mode testing are consistently larger than those shown in Table 5.22. The ASE linear quantities presented in Table 5.22 are slightly larger than those shown in Table 5.19. For example at ten (10) nautical miles the multiple beacon ASE mean is -.73 nm and the single beacon ASE mean is -.15 nm. The ATE values indicated in Table 5.22 once again are consistently small.

Table 5.23 presents the Letdown Error (LDE) quantities for the multiple beacon testing conducted at the airport site. The LDE quantities determine the ability of the pilot to utilize the ARA system to define and identify a step-down fix. As shown earlier the LDE quantities are sampled at two approach fixes, that is the 5.0 nautical mile IAF and the 2.0 nautical mile FAF. The mean value at the IAF shows that the pilot initiated his descent .05 nm before passing the fix. The mean value at the FAF shows that the pilot initiated his descent from 500 feet to 200 feet .08 nm after passing the fix. The one-sigma values are .31 nm and .15 nm, respectively. In the multiple beacon case the pilot at the 2.0 mile fix started his descent after passing the fix as was the case in the single beacon testing.

table 5.23 - multiple beacon retubmn cribe Quan	ntitles.
---	----------

	Error Mag	nitudes
Approach	Mean	•] •
Position	nm	nia
IAF (5.0 nm Fix)	0460	.3130
EAF (2.0 nm Fix)	.0750	.1505

Figure 5.26 summarizes in graphical form the Total System Cross Track Error (TSCT) for the multiple beacon testing conducted at the airport site. Outside of ten (10) nautical miles the maximum deviation from intended course is 2.9 nm (at 22 nm along track distance). Between five (5) and ten (10) nautical miles the maximum deviation from the desired track is 1.5 nm and within five (5) nautical miles the maximum error indicated by Figure 5.26 is 1.1 nm. All of the approaches are flown well within the airspace requirements established by the RTCA SC-133 MOPS. Compared to Figure 5.19 (single beacon approach TSCT plot, Section 5.3.3) the approaches in Figure 5.26 at all range intervals show smaller maximum deviations from intended course. The approaches in Figure 5.26 show a marked improvement over the single beacon approaches in terms of track identification and orientation, but because of the large displayed beacon size it was still difficult to determine a course correction angle to eliminate the "homing" tendencies. Figure 5.27 presents a plot of the Flight Technical Error (FTE) quantities vs. along track distance for the multiple beacon testing conducted at the airport site. This plot can be compared to the FTE single beacon approach plot (Figure 5.21) seen in Section 5.3.3. Within five (5) nautical miles Figure 5.27 shows a maximum error quantity of 1.2 nm. In the five (5) to ten (10) nautical mile interval the maximum error shown is 3.1 nm. The larger error spikes are often due to the slow update rate of the radar in conjunction with rapid aircraft heading changes. Outside of ten (10) nautical miles the maximum error indicated by Figure 5.21 is -3.2 nm. With the exception of the one spike at nine (9) nautical miles, the FTE error quantities presented in Figure 5.27 are smaller than those presented in Figure 5.21. Although the workload during the multiple beacon approach could conceivably be higher than the single beacon approach it does seem to offer the pilot some track orientation information, therefore, reducing TSCT and FTE quantities.

Figure 5.28 is a plot of Airborne System Error (ASE) for the multiple beacon testing. As seen in all other areas of testing with the Bendix radar system the system errors are small. The plot indicates that all of the error quantities are within a ± 1.8 nm region outside of five (5) nautical miles and within a $\pm .8$ nm region within five (5) nautical miles.



ARA Airport Multiple Beacon Mode Total System Cross Track Error

Figure 5.26

ARA APPROACHES -- AIRPORT SITE BENDIX RDR-1400A MULTIPLE BEACON MODE

5~74


ľ

ARA APPROACHES -- AIRPORT SITE BENDIX RDR-1400A MULTIPLE BEACON MODE

15 APPROACHES AGGREGATE FTE

5-75



BENDIX RDR-1400A MULTIPLE BEACON MODE ARA APPROACHES -- AIRPORT SITE

Figure 5.29 presents histograms of the ARA TSCT, FTE, ASE and ATE quantities for the multiple beacon testing. The data is presented such that only data within ten (10) nautical miles is included. Figure 5.29 shows that the TSCT distribution appears skewed considerably to the left. The FTE distribution is skewed only slightly to the left. The system errors show that the ATE quantities are again close to zero while being skewed to the left slightly. The ASE quantities appear skewed to the left as in the TSCT and FTE cases.

-

1

5.3.5 Offshore Site: Primus-50 Combined Mode Versus The Beacon-Only Mode

The subsection will compare the results obtained in the offshore (Brandywine Lighthouse) testing utilizing the RCA Primus-50 Radar in the combined and beacon-only modes. The combined mode offers the unique capability of interrogating a ground based transponder while at the same time displaying surrounding skin paint targets. The results are not intended to show that one mode is any better than the other in terms of accuracy, since both modes use the ground based beacon as the primary mode of navigation.

Table 5.24 summarizes the results of the ARA testing conducted at the offshore site in the combined mode. The error analysis log and statistical summary of error quantities in the table presents the mean values, standard deviations, number of data points and number of approach segments for four specific error quantities: ARA ATE, ASE, FTE and TSCT.

Table 5.24 shows in the ARA ATE case that the calculated mean is .3587 nm and the one-sigma is .2816 nm for all of the approach segments. The results for the ARA ASE were a total mean value of .1600 nm and a one-sigma value of .6360 nm. It should be noted that these results were derived for a large sample size (665 data points).

The Flight Technical Error (FTE) quantities indicated in Table 5.24 showed a mean value of -.0039 nm and a one-sigma of .6448 nm. These values were obtained from all of the approach segments flown. The TSCT values presented in Table 5.24 show a total mean value of .1561 nm and a one-sigma value of .4044 nm. The cross track values are very small in relation to the primary airspace requirements of +4 nm specified in the MOPS established by RTCA SC-133. All of the values presented in



	🔀 (nm)	σ (nm)	Data Points	Approach Segments
ARA ATE				
Long Short Total	.4617 .2484 .3587	.3419 .1249 .2816	344 321 665	3 6 9
ARA ASE				
Long Short Total	.4353 1351 .1600	.7651 .2072 .6360	344 321 665	3 6 9
FTE				
Long Short Total	0327 .0270 0039	.8396 .3247 .6448	344 321 665	3 6 9
TSCT				
Long Short Total	.4026 1082 .1561	.3300 .2953 .4044	344 321 665	3 6 9
IDENTIFIER		True Heading	Se	gment
6/26/79 -1		222	S	hort
6/27/79 AM-1 6/27/79 AM-2 6/27/79 AM-3		222 222 222	L S S	ong hort hort
6/27/79/PM-1 6/27/79/PM-2 6/27/79/PM-3		222 222 222	L S S	ong hort hort
6/29/79 AM-1 6/29/79 AM-2		222 222	Ĺ	ong hort

Table 5.24 NAFEC ARA RCA Primus-50 <u>Combined Mode</u> Offshore Approaches Error Analysis Log And Statistical Summary

5-79

1

the table are small with the exception of the ARA ATE. This somewhat larger value is partially attributable to the large target that is displayed by the Primus-50 in the beacon and combined modes. Range interpretation is difficult on the part of the operator because of the large displayed size, but the assumption was made both operationally and in the data reduction that the return edge closest to the radar apex represented the correct range to the target. Section 5.4 will present details of the Primus-50 return size versus range setting.

Table 5.25 summarizes in statistical quantities the ARA test data at one nautical mile intervals, starting at ten (10) nautical miles in the combined mode. These quantities were calculated in both linear and angular terms. Starting at three (3) nautical miles the angular FTE quantities become large. At one nautical mile the FTE angular mean value is 6.6 degrees and the one-sigma value is 12.9 degrees. The TSCT angular quantities at one nautical mile show a mean value of 4.4 degrees and a one-sigma of 11.4 degrees. The angular TSCT two (2) and three (3) nautical mile quantities show a mean value of 3.1 and 2.6 degrees and a one-sigma of 8.8 and 8.4 degrees, respectively. The ASE angular quantities indicated in Table 5.25 are consistently small at all ranges.

The linear errors presented in Table 5.25 are very consistent and small for the entire range of the approach. The linear FTE quantities at the one nautical mile point show a mean of .12 nm and a one-sigma of .23 nm. The TSCT values (at one nautical mile) indicated in Table 5.25 show a mean value of .08 nm and a one-sigma of .20 nm. The TSCT mean value at all ranges out to ten (10) nautical miles consistently remains in the .08 nm to .21 nm interval. The ATE values are not as consistent with a mean value of .12 nm at one mile and a mean value of .34 nm at ten (10) miles.

Figure 5.30 summarizes in graphical form the total system error for all the approaches flown in the combined mode at the offshore site. The plot shows that between five (5) and ten (10) nautical miles from the beacon the maximum deviation from intended track is -1.0 nm. This quantity is well within the four (4) nautical route width established by the RTCA SC-133 MOPS. Within five (5) nautical miles the maximum deviation from the intended course is only .8 nm, which is within the required airspace limits of +1.7 nm. at the MAP set by the RTCA SC-133 MOPS.

5-80

. ...

RCA Primus-50 Combined Mode One Nautical Mile Aggregate Data Table 5.25

ž		L L	 	1				
	••	111			Ц 0 С		13. 1.	ti ti
			1010 ·		· • 0000	4.012		
			1004					
ł	•	.1425	1070	1.52	- 0.24			
		.0054			0550.			医白眼的 医骨骨骨骨骨骨骨骨骨骨骨骨骨骨骨骨骨骨骨骨骨骨骨骨骨骨骨骨骨骨骨骨骨骨骨骨
65	•.	8161 ·	(N)));	. [743				
		1.200	100.	1975.	0810 •	0.4401		
¢	•		と言いで.	1001.	0344	1.0407	(() () () () () ()	
		5 - 5 - 5 - 5 - 5 - 5 - 5 - 5 - 5 - 5 -	on Star	2747.	- 1199	4.1237	6. 39500	
ប់	<i>t</i> .		ALAO.	र्म्स इन्हे इन्हे	1107	1.0051		
		00000 .	4417	0415.	.185∿	5-12C	0.04 0.0	
ν ^j		- NO 44	.101.	.2052.	1040	. 2657	0000 1000 1000	
		1121	.4726	04 00 04 04 04	- 04 10 11	1.5007	0100 N	
` .	Ç.,	មាំ ស្រុះ ស្រុះ	.1040	ものしご。	1446	0000 000 0	2.0405	-1 - 10 - MEAN
		.140.	. 4425		♦	00 7 000 10	1	
1	çî.	.3117.	.0369		-101-	0000 ·	· · · · · · · · ·	
		0000	1904	00000 1	.4014	3.5077		
-	۰.	10.7	.1001.		.1002	.6376	- 0000	
		.1100	00 00 10 10	.036	0. 107.	0,4260	0.120	
ੁ		-2446	.2078	.0371	.1707	1.1004		
		5-1	-202.	.4234	. 5160	2 - 00 20	2.4.17	
							1	1 J J J 1 1 1 1 1

هر.





ARA APPROACHES -- OFFSHORE SITE RCA PRIMUS-50 COMBINED MODE

9 APPROACHES

The plot indicates that the pilot once more had a tendency to "home" towards the station, but in all cases flew directly to the target.

Figures 5.31 and 5.32 are plots of Flight Technical Error (FTE) and Airborne System Error (ASE) vs. distance along the desired track, respectively. The error quantities represent all of the approaches flown at the offshore site. Figure 5.31 shows that between ten (10) and five (5) nautical miles the maximum FTE error shown is -2.1 nm and within five (5) nautical miles the maximum error indicated is .95 nm. The maximum error indicated in Figure 5.31 is 2.5 nm at 17 nm from the beacon. Figure 5.32 shows that between ten (10) and five (5) nautical miles the maximum ASE quantity shown is -1.3 nm and within five (5) nautical miles the maximum error quantity indicated is -1.0 nm. The maximum ASE quantity indicated in Figure 5.32 is -2.8 nm at 17 nm.

Table 5.26 presents the error analysis log and statistical summary for the offshore site testing conducted in the Primus-50 beacon-only mode. Table 5.26 is identical in format to Table 5.24 and presents the same four basic error quantities. The calculated mean in the ARA ATE case shows a mean value of .3955 nm and a one-sigma of .3071 nm for all of the approach segments. The ARA ASE quantities show a mean value of -.2908 nm and a one-sigma of .4218 nm. Both of the mean and one-sigma quantities just mentioned compare favorably with those seen in the combined mode for similar error quantities.

The FTE quantities indicated in Table 5.26 show a total mean value of .5587 nm and a one-sigma of .6935 nm. The TSCT quantities presented show a mean value of .2680 nm and a one-sigma of .4231 nm. These total aggregate values are large because only one long segment approach was flown during this testing and the values indicated in Table 5.26 for the long segments are slightly larger than usual.

Table 5.27 presents the Primus-50 beacon-only mode statistical error quantities aggregated at one nautical mile intervals. As shown in Table 5.25 the quantities are presented in both linear and angular terms. The angular FTE quantities indicated in Table 5.27 at one nautical mile show a mean value of -3.4 degrees and one-sigma value of 8.6 degrees.



ARA Offshore RCA Primus-50 Combined Mode Flight Technical Error

Figure 5.31

ł;

ŕ

. . .

Î

ARA APPROACHES -- OFFSHORE SITE RCA PRIMUS-50 COMBINED MODE

9 APPROACHES AGGREGATE FTE



ARA APPRUACHES -- OFFSHORE SITE

RCA PRIMUS-50 COMBINED MODE

ARA Offshore RCA Primus-50 Combined Mode Airborne System Error Figure 5.32 · ! ·

i,

Ì

	$\overline{\mathbf{x}}$ (nm)	σ (nm)	Data Points	Approach Segments
ARA ATE				
Long Short Total	.5932 .2900 .3955	.4084 .1546 .3071	173 324 497	1 5 6
ARA ASE				
Long Short Total	6089 1209 2908	.5457 .1774 .4218	173 324 497	1 5 6
FTE				
Long Short Total	1.2790 .1741 .5587	.6005 .3468 .6935	173 324 497	1 5 6
TSCT				
Long Short Total	.6701 .0533 .2680	. 3263 . 2922 . 4231	173 324 497	1 5 6
IDENTIFIER		True Headin	<u>e</u>	Segment
6/29/79 AM-3 6/29/79 AM-4		222 222		Short Short
6/29/79 PM-1 6/29/79 PM-2 6/29/79 PM-3 6/29/79 PM-4		222 222 222 222 222		Long Short Short Short

Table 5.26NAFEC ARA RCA Primus-50 Beacon-Only Mode
Offshore Approaches Error Analysis Log
And Statistical Summary

RCA Primus-50 Beacon-Only Mode One Nautical Mile Aggregate Data Table 5.27

+------

ž,	ίλο	ATE		E TE	ACE ACE	TSULT	5 LE	ASE	
-	ć	0001.	0**	1000.	<u>. cr</u>	-2.0705	-3.4374	0 0 0 0 0 0 1	MEAN
			1947 J -	.1551	.0735	4.9490	9.648°	4.2061	5 T D
Ç4	~ ~	STA.	1067.	0543	0539	-3.1109	-1.5691	-1.5142	MEAN
		0400.	1110日。	.2694	.0646	04 	7.5707	1.0010	Ц С Ц С С
en.	<u>,</u> "	1-1-1	- 00 ac	0880.	0849	. 0000	1.4005	-1.6208	MEAN
		,0664	. 4360	.4171	.0566	0. 2680 2680	7.9102	1.0018	11 0
4	۰,	Form.	0000.	.2047	.1100	1.2320	1000	-1.6927	MEAN
		.0541	.4296	0039 .	.1220	6.1307	6.9136	1.7169	0) 11 0)
บ้	ί.	0.887	.1649	-020.	- 1351	1.8801	1.6610	0021.1-	MEAN
		.04.6.	.0978	.4302	.1616	4.5486	4.0174	1.000¢	etto
Ś	Ċ,	2000 1	.2119	.3677	1528	2,0514	0.000 0.000	-1.4536	MEAN
		.0535	.3613	.3619	.1267	3.4459	0.4510	1.2000	្មាច
~	Ŷ	00000.	500 °	.1206	1261	2.4089	3, 4301	-1.0318	MEGN
		0000°.	.3364	4000. 4000	.1000	0.7010	0 0 0 0 0 0 0 0 0 0 0 0 0 0	1) () () () ()	11
0	Ś	0400°	19900 ·	.5744	0000 1	2.4017	1.1070	-1.7105	PLE AN
		1230.	0110 10 10	. 4000	.2038	2.3106			
ņ.	۶-	.7245	0000 ·	1000 1001	(4 () (4))	2.1627	0.0160	-1.3570	
		1040. 1040.	. 1310	.4315	. 0355	2.7420	2.7417		110 0
्	. .4	.0000	. 0000	.000	.0000	.0000	.000	.0000	NC IN
		, 0000	.0000	.0000	.0000	.0000	.0000	.0000	Li Li

Ĺ

5-87

The TSCT angular quantities at one nautical mile show a mean value of -2.1 degrees and a one-sigma of 4.9 degrees. All of the quantities indicated at ten (10) miles are zero because there was only one data point available in this interval. The FTE linear quantities presented in Table 5.27 show a mean value of -.06 nm and a one-sigma of .15 nm at the one nautical mile point. The largest FTE quantities can be seen at the eight (8) nautical mile point where the mean and one-sigma values are .57 nm and .45 nm, respectively. The same can be seen for the linear TSCT values at the eight (8) nautical mile point where the mean is .34 nm and the one-sigma is .32 nm. The ATE and ASE linear quantities increase progressively as the distance from the beacon increases.

Figure 5.33 is a plot of total system error of all the approaches flown at the offshore site in the Primus-50 beacon-only mode. The long approach represented in this plot verifies a point made earlier in this subsection. The approach shows that without the aid of some type of track orientation technique the pilot has a tendency to "home" to the station. Outside of ten (10) nautical miles the maximum deviation from intended course is -1.1 nm. Between ten (10) and five (5) nautical miles the maximum quantity is -1.0 nm and within five (5) nautical miles the largest error shown is .6 nm.

Figures 5.34 and 5.35 are graphical representations of FTE and ASE, respectively, for all the approaches flown in the Primus-50 beacon-only mode. The FTE and ASE values are plotted vs. along track distance to the target. The FTE plot shows a maximum error quantity of -2.5 nm outside of ten (10) nautical miles. Within ten (10) nautical miles the maximum error shown is -1.4 nm (at seven (7) nautical miles). The ASE quantities from Figure 5.35 show a maximum error of .8 nm within ten (10) nautical miles and the largest error shown is -2.1 nm (at 22.5 nm along track distance). All of the FTE and ASE values in Figures 5.34 and 5.35 are very small which reflects good system quality and good approach procedures.

Figures 5.36 and 5.37 are histograms of ARA TSCT, FTE, ASE, and AIE for the RCA Primus-50 combined and beacon-only modes, respectively. The TSCT distribution presented in Figure 5.36 appears skewed to the left, while the FTE distribution appears normal. The radar system errors presented in Figure 5.36 shows that the ATE distribution appears skewed slightly to the



ARA APPROACHES -- OFFSHORE SITE RCA PRIMUS-50 BEACON-ONLY MODE ARA Offshore RCA Primus-50 Beacon-Only Mode Total System Cross Track Error Figure 5.33


ARA Offshore RCA Primus-50 Beacon-Only Mode Flight Technical Error

Figure 5.34

ARA APPROACHES -- OFFSHORE SITE RCA PRIMUS-50 BEACON-ONLY MODE

6 APPROACHES AGGREGATE FTE



ARA Offshore RCA Primus-50 Beacon-Only Mode Airborne System Error Figure 5.35

ARA APPROACHES --- DFFSHORE SITE RCA PRIMUS-50 BEACON-ONLY MODE

6 APPROACHES AGGREGATE ASE



5-92



5-93

....

right and the ASE distribution appears skewed to the left. The TSCT and FTE quantities presented in Figure 5.37 appear skewed to the right while the ATE distribution is also skewed to the right and the ASE quantities appear skewed to the left.

The data presented in this subsection (5.3) indicated that the track orientation techniques utilized enhanced the overall ARA performance. The cursor technique afforded the pilot with better course guidance therfore, reducing the FTE and TSCT quantities. The cursor technique proved effective in both the skin paint and beacon modes. The multiple beacon technique also showed a decrease in the FTE and TSCT quantities in comparison to the single beacon testing. The RCA Primus-50 testing showed that in the combined mode because of the large displayed beacon size close-in surrounding skin paint targets were blocked out. An operational evaluation will be discussed in Section 5.4 and the results will be further discussed in Section 6.0, Summary of ARA Performance.

5.4 OPERATIONAL EVALUATION OF THE ARA CONCEPT

Operationally the Airborne Radar Approach (ARA) concept is a practical solution to navigation where conventional navigation aids are unavailable. There are certain areas though that need careful consideration which relate to the operational feasibility of the Airborne Radar Approach System utilizing various track orientation techniques. First, the ground based beacon transponders must have an effective backup system. At times during this phase of testing and during the single beacon approach testing the beacons were either inoperative or weak and intermittent. This particular problem caused serious problems especially during the multiple beacon testing because without some means of beacon identification, if one beacon is inoperative it is impossible to determine whether or not the displayed beacon is the intended target. Second, the Airborne System needs to offer more advanced features to reduce the pilot's workload. The copilot must constantly monitor the gain (especially in the multiple beacon mode), tilt, range controls and aircraft heading. For purposes of reducing crew workload more advanced Sensitivity Time Constant (STC) circuitry in the Airborne Radar System or variable-gain beacons are required. It should be noted that the STC was not properly adjusted in the Bendix Radar System during the multiple beacon testing.

This fact was verified by the Bendix Corporation (and an adjustment was made) before the start of the skin paint, skin paint with cursor and single beacon with cursor testing. Also, the pilot needs some indication of the aircraft's actual deviation from intended course. This information was supplied to the pilot by utilizing the cursor technique and the multiple beacon technique. It was the purpose of these tests to determine if by using these track orientation techniques the pilot could fly the intended course more accurately with a reduction in mental workload.

The Airborne System Error quantities presented in Section 5.3 indicate that the present "State-of-the-art" system is capable of accurate navigation. The operational problems as related to airspace lie in two areas: pilot's workload and pilot's interpretation of the information presented by the radar system. This presentation of information might include a beacon return, a skin paint return, multiple beacon returns or a beacon with cursor return. Certainly training and workload correlate closely together. If the pilot is trained well and understands the concepts involved in flying the approach, the workload is reduced. It is certainly obvious that even a trained pilot should not have to be constantly changing display controls, since this only distracts from his primary duty of flying the approach safely. Presently the Airborne Radar System return display is adequate, but improvements could be made. For example the wide target azimuth displayed by the Bendix Radar could be improved upon. Also, the large return (both in range and azimuth) displayed in the beacon only and combined modes by the Primus-50 radar certainly could be improved. The cursor technique implemented by NAFEC on the Bendix System certainly is a major improvement in displayed track orientation information. The cursor aiged approaches as shown earlier in Section 5.3 required less airspace than the non-cursor approaches. The marked reduction in the FTE and TSCT values is significant.

5.4.1 Landside ATC Integration

Landside ATC integration could offer some interesting problems because of the items discussed in Section 5.1.3. Although the Airborne System accuracy is good, some difficulties exist in integrating Airborne Radar Approaches into the present ATC System. Comparatively speaking, a standard NDB approach is the present day non-precision approach equivalent

to an ARA. With the implementation of the cursor technique it is feasible to say that the radar offers a weak form of omnidirectional guidance capability. This extra course guidance available with the cursor technique positively impacts airspace requirements. As with any non-precision approach the pilot's proficiency will directly impact the overall accuracy of the approach.

As shown in Section 5.3, the figures indicate that a ±4 nm route width, as established by RTCA SC-133, will be required for those approaches that do not utilize any type of track orientation techniques. Figures presented later in this section relating to airspace requirements will show that those approaches that utilized track orientation techniques used considerably less airspace. If an Airborne Radar Approach to a helipad requires considerable airspace, then the controller will surely need to consider this fact when vectoring other aircraft in his area, because this will surely affect the controller's position in assuring safety to aircraft in this area. The impact of this could potentially be delays in terminal operations. Airborne Radar Approach procedures also need to offer sufficient obstacle clearance, depending on the overall accuracy of the system. If this is the case then ARA altitude minimums might possibly have to be raised above those of the present day nonprecision approaches.

The procedures utilized for ARA should also consider that, when making an Airborne Radar Approach, positive navigation is only available when flying to the target. This positive navigation might also pecessitate the use of beacons instead of depending strictly on skin paint. This could possibly require that some other navigational aid be used to initially acquire the intended inbound track. At NAFEC, during the test program RNAV was used for initial course acquisition. Seconds show that the inaccuracy of the RNAV system used offered some poor course acquisition techniques in the terminal area especially during the single beacon testing (Reference 1). This presents another problem to the controller. The controller must then consider that the pilot will be using two operationally different navigational aids to thy the approach, both having different accuracies and different procedures necessary to make the approach successful. The landside integration of Airborne Radar

Approaches could present difficulties at first, but in time, when standard procedures are developed and airborne system features are improved, airborne radar could possibly become an integral part of our terminal navigation system.

5.4.1.1 Airport Terminal Area

The airport site is without question the most important area in which to consider the ATC integration impact. The airport environment could produce many traffic and obstacle hazards not found at offshore sites. In an environment such as this, many considerations must be recognized so that the procedures utilized offer the utmost safety. A major consideration is the airspace required to repeatably fly an Airborne Radar Approach down to established minimums successfully. This subsection will present plots of mean, mean ±2-sigma values for four specific error quantities: TSCT, FTE, ASE and ATE.^{*} Also plots of TSCT quantities will be presented with the airspace boundaries established by RTCA SC-133 overlayed.

figures 5.38 - 5.40 present a graphical representation of the airport approaches as they relate to obstacle clearance airspace for three specific test areas: single beacon-only, single beacon with cursor and multiple beacon, respectively. The primary and secondary airspace requirements shown are those indicated in the Minimum Operational Performance Standards (MOPS) established by RTCA SC-133. The Initial Approach Fix (IAF) and Final Approach Fix (FAF) shown in Figures 5.38 - 5.40 are not the same IAF and FAF used in the testing procedures. These approach fixes are the recommended approach fixes established by RTCA SC-133. Figure 5.38 shows that for the ARA single beacon approach testing from the IAF to the MAP all of the approaches lie within the primary airspace requirements. Figure 5.39 shows that for the ARA single beacon with cursor approaches, again from the IAF to the MAP, all of the approaches lie within the primary boundaries but a comparison between Figures 5.38 and 5.39 will show that the cursor aided approaches were flown with a great deal more precision. In fact, all of the cursor aided approaches lie within ±4.0 nm requirement. Figure 5.40 presents an airspace requirement plot for the multiple beacon testing conducted at the airport site. These approaches also lie within the primary airspace boundaries from the IAF to the MAP.

/NOTE/ *These plots were generated utilizing aggregate data at one nautical mile intervals.



ARA Airport Single Beacon Approaches Related To Obstacle Clearance Airspace



ARA Airport Single Beacon With Cursor Approaches Related To Obstacle Clearance Airspace 5:3ure 5.39

1

+ · · · ·

ł,

Ì

\$



The approaches indicated in Figure 5.40 were also flown with great precision and all lie within airspace boundaries of ± 2.0 nm between the IAF and MAP. Reference 2, Terminal Instrument Procedures (TERPS) shows that for a standard NDB approach primary airspace requirements at the IAF are ± 3 nm. Figure 5.38 (single-beacon approaches) indicates that the approaches flown also lie inside of a ± 3 nm airspace region, in fact they lie within an airspace region that is ± 2 nm.

Figures 5.41 - 5.43 are graphical representations of Total System Cross Track Error (TSCT), Flight Technical Error (FTE), Airborne System Error (ASE), and Along Track Error (ATE). The mean value and mean ± 2 sigma values are plotted vs. distance to the beacon starting at ten (10) miles. The significance of these figures is strictly with reference to airspace requirements. Figure 5.41 indicates that for the single beacon approaches the TSCT, FTE, ASE and ATE mean +2-sigma values are virtually constant with range from the beacon, converging only slightly in the ASE case as the target is approached. It is suprising to note that the TSCT and FTE mean ±2-sigma values are constant when in fact they should converge as the target is approached. Figure 5.42 indicates that for the beacon with cursor approaches the TSCT, FTE and ASE mean +2-sigma values converge as the target is approached, while the ATE values remain constant. The same facts are evident for the multiple beacon TSCT, FTE, ASE and ATE values indicated in Figure 5.43. Figures 5.43 does show that in the TSCT case the mean +2-sigma error quantities stay virtually constant between eight (8) and three (3) nautical miles. The FTE and ASE quantities as mentioned earlier converge as the target is approached.

With the above facts in mind two conclusions can be drawn. First, the approaches flown utilizing the track orientation techniques require less airspace especially as the target is approached. Second, testing to date on Airborne Radar Approaches indicates they would work well in the non-precision approach environment. The procedures would necessitate little ATC structural change to integrate the ARA approaches at airport sites into the present system. But as mentioned earlier in Section 5.4.1, unless the approach is straight in, i.e., no holding patterns are required, etc., some other means of navigation is required to acquire the IAF. If two sets of procedures are required then complications set in that involve both the pilot and controller. ARA is a viable concept that with careful consideration can integrate with the present ATC system.

18 18 0 œ FTE (CROSS-TRACK) DISTANCE (NM) DISTANCE (NM) ω ω 111 ASE ທ ທ + Data Perived From 11 Augmonts Segments DASHED LINE: MEAN VALUE SOLID LINE: MEAN PLU3 DR MIMUS E-SIGMA Ø n ผ e) ø ٩J 60 ų ရှ + Ø, ŋ Ť 7 ရ BIUJAV AMBIB-9 . ИАЗМ NEAN . 2-SIGMA VALUES 18 18 თ Ø **^-**DISTANCE (NM) DISTANCE (NM) 80 ω TSCT ATE S ທ Ø, n e, n 1 t ېه ຝ Ð សុ en I ŧ 7 MEAN & E-SIGMA VALUES EBULAU ANDIE-S . NASH

Airport Site Firshe Beacon Mode: Mean, Mean Plus Or Minus 2-Sigma Values

711, re 5.41

ŀ

i.

ł.

14

3



5-103

1

. •



5.4.2 Offshore ATC Procedures

Offshore ATC procedures utilized will be quite different from the landside procedures. These offshore targets typically consist of prominent surface objects such as oil rigs, lighthouses, and buoys, or beacons placed on these targets. Since the targets are so far offshore, conventional navigation aids are virtually non-existent with the exception of NDB's located on some oil rigs. The airborne radar system is a low cost, dependable answer to the situation.

The ATC procedures utilized would basically consider two items: first, obstruction clearance, which is very important to the operator who flies to a cluster of oil rigs and wants to execute an approach, and second, what weather minimums are necessary to safely conduct the approach in the presence of surrounding oil rigs, ships or lighthouses. Figures 5.44 - 5.48 present in graphical form the ARA offshore approaches as they relate to obstacle clearance. The test areas presented in Figures 5.44 - 4.48 are as follows: single beacon-only, skin paint, skin paint with cursor, Primus-50 combined mode and Primus-50 beacon-only mode, respectively. Figure 5.44 presents the airspace plot for the single beacon approach testing. The Figure illustrates that within ten (10) nautical miles all of the approaches lie inside a \cdot 2 nm region with the exception of two. Even these two approaches still lie inside the airspace requirements established by RTCA SC-133 as represented in Figure 5.44. Figure 5.45 presents the airspace requirements for the offshore skin paint tests. As indicated in Figure 5.45 all of the approaches that were flown to the correct target lie within a +1 nm region, but on two occasions the pilot flew to the wrong target taking the aircraft outside of the airspace boundaries. Figure 5.46 presents the airspace plot for the Skin Paint with Cursor Approach testing. Between the IAF and MAP all of the approaches flown lie inside of a +2 nm route wide, well within established RTCA SC-133 limits. Figures 5.47 and 5.48 presents the airspace plots for the RCA Primus-50 combined and beacon-only mode testing, respectively. All of the approaches flown during the Primus-50 test period lie within required airspace boundaries. All of the approaches flown during the offshore testing were within specified limits with the exception of a few skin paint approaches. This only verifies the fact that it is essential to have some means of positive target identification before the approach



ARD Cffsnore Single Beacon Approaches Related To Obstacle Clearance Airspace rigure 5.44

(]







ARA Offshore Skin Paint With Cursor Approaches Related To Obstacle Clearance Airspace Figure 5.46







ARA Offshore Primus-50 Beacon-Only Approaches Related To Obstacle Clearance Airspace 5:3ure 5.48

1

1.1.1

i

t:

Ĭ
can be conducted down to minimums safely. Figures 5.49 - 5.53 are graphical representations of the mean and mean ± 2 -sigma values vs. distance to the beacon for four specific error quantities. They are as follows: TSCT, FTE, ASE and ATE. Figure 5.49 indicates that the offshore single beacon mean ± 2 sigma values for the TSCT and FTE quantities converge as the beacon is approached, while the ATE and ASE mean ±2-sigma values are virtually independent of along track distance. Figure 5.50 shows some mean ± 2 sigma results that are different than those seen in previous figures of this nature. Figure 5.50 indicates that the skin paint TSCT mean ± 2 sigma values are virtually independent of range with some values converging at one (1) nautical mile and also converging at ten (10) nautical miles because of the lack of data points available. The FTE and ASE mean ± 2 sigma values indicated in Figure 5.50 converge as the target is approached while the ATE values start to diverge at four (4) nautical miles from the target. Figure 5.5] presents the mean ± 2 -sigma results for the skin paint with cursor mode testing. The TSCT mean ±2-sigma values converge as the distance to the beacon decreases and the FTE values converge and diverge at varying distances from the target. The ATE values behave similiarly to the FTE values while the ASE values in Figure 5.51 converge as the target is approached. Note, the values plotted in the skin paint mean, mean ±2-sigma plots do not include those approaches flown to the wrong target. Figures 5.52 and 5.53 presents the mean values and mean ± 2 -sigma values for the RCA Primus-50 test period. On both figures the TSCT and FTE mean ±2-sigma values behave similarly, that is, both sets of quantities remain virtually constant as the beacon is approached, converging only slightly close-in. Figure 5.52 shows that the combined mode mean ± 2 -sigma values for the ASE case converge as distance to the beacon decreases and the ATE quantities appear to be independent of range. Figure 5.53 shows that the beacon-only mode ASE and ATE mean ± 2 -sigma quantities are virtually independent of range with only a slight convergence in certain areas.

Table 5.28 presents the RCA Primus-50 beacon return area statistics. This table is presented to show the relative area covered on the radar screen by the RCA Primus-50 return. The quantities show that at the 75 mautical mile range setting a mean area of 20.25 nm² is covered by the beacon return. At the ten (10) nautical mile setting a mean area of 7.7 nm² covered and at the two (2) nautical mile setting an area of .06 nm² is

8 18 0 (CROSS-TRACK) ¢D ~ DISTANCE (NM) DISTANCE (NM) ω ω FTE ASE ທ ທ Data Derived From 12 Approach Segments DASHED LINE: MEAN VALUE Solid Line: Mean Plus or Minus R-Sigma თ മ ស QJ 60 Q, ស្ព 80 7 ရှ 1 QJ 0 7 မှ o; t BBULAV AMBIB-S * NASM иазм BBUJAV AMDIC-S # 18 18 60 Ø ~ : 3101 DISTANCE (NM) DISTANCE (NM) G ω TSCT АТЕ ທ ល + + თ n 11111 61 Q, e, 80 n I ۹ + 94 ~ 60 ရ ကို 7 7 Ť EBULAV AMBIE-S + NASM EBULAV AMBIE-S . NABM









Bendix RDR-1400A Skin Paint With Cursor: Mean, Mean Plus Or Minus 2-Sigma Values

Figure 5.51

••

*

5



1

19 18 Φ თ FTE (CROSS-TRACK) 10 ഹ RCA Primus-50 Beacon-Only Mode: Mean, Mean Plus Or Minus 2-Sigma Values 3 ~ DISTANCE (NM) DISTANCE (NM) 60 ω 11111 ASE ທ ທ + NOTE: Data Derived From 6 Approach Segments DASHED LINE: MEAN VALUE Solid Line: Mean Plus or Minus 2-Sigma ŋ Ø ØJ. **N** ۹ ŋ * **N** Ø លុ n ***** 7 7 REAN . 2-SIGMA VALUES BBUJAV AMBIE-S . NABM 10 18 თ ŋ, Ø ø 5 ~ DISTANCE (NM) DISTANCE (NM) Q ω TSCT ATE S ທ + ٠ m თ N, ណ

5-116

ຸ

60

EBUJAV AMBIE-S . NABM

ដុ n 1 *

1

7

+

٩ ĥ

Figure 5.53

¢,

69

MEAN + 2-SIGMA VALUES

blocked out by the Primus-50 return. These quantities are quite significant because in the combined mode, with the range selector set on 2.0 nm, skin paint targets within a 840 foot radius circle are not visible on the radar screen because of the large beacon return displayed. Since such a large area is covered by the return, obstacles close to the target are not displayed on the radar screen, therefore, offering no obstacle clearance information.

Range Selector Setting	Mean (nm ²)	One-Sigma (nm ²)	Number Of Points
75	20.25	16.66	43
25	7.70	2.74	448
10	.43	.15	498
4	.14	.05	277
2	.06	.03	149

Table 5.28 RCA Primus-50 Beacon Return Area Statistics

When flying to an offshore site the most critical stage of the approach is near the missed approach point. The missed approach procedures utilized will need to be executed in a direction that will direct the aircraft well clear of any obstacles. Since a cluster of oil rigs are usually situated so that 6-8 rigs lie very close to each other, operating in the skin paint mode might offer confusion to the pilot as to which oil rig to select for landing as shown by earlier examples. If operating in the beacon mode, a discrete return would be displayed at the landing site. This discrete target concept offers the pilot reassurance and establishes confidence during the final stages of the approach. For reliable missed approach procedures the pilot must be familiar with the formation of the cluster so he can choose the safest missed approach procedure according to his direction of flight. ARA for offshore use is very practical. It offers navigation where other navigational aids are unavailable and it offers the ability to display surface objects as well as discrete beacons.

SUMMARY OF ARA PERFORMANCE

This section summarizes the technical and operational performance of the Airborne Radar Approach (ARA) System. This summary includes the testing performed in the skin paint, skin paint with cursor, single beacon with cursor, multiple beacon, combined and beacon-only approach modes.

6.1 TECHNICAL PERFORMANCE

The technical performance objectives stated below are in response to the RTCA SC-133 "Minimum Operational Performance Standards" (MOPS) requirements. This is a summary of the quantitative data presented in detail in Section 5.3.

6.1.1 Range Performance

The beacon testing conducted utilizing various track orientation techniques showed basically the same range performance results obtained during the single beacon testing (Reference 1). That is, testing utilizing the Bendix Radar System showed a maximum acquisition range of 21 nm at an altitude of 1000 feet with the beacon at ground level. During the offshore RCA Primus-50 testing the beacon was situated approximately thirty feet from the water's surface, which allowed a consistent maximum acquisition of 35 nm at 1000 feet altitude. The minimum range at which the beacon could be tracked and displayed for both systems was .7 nm at 200 feet. These results were both qualitatively and quantitatively determined from the data collected during the track orientation techniques and RCA Primus-50 testing. No attempt was made during these tests to specifically determine the range performance during adverse weather conditions such as precipitation.

The offshore skin paint tests showed that since the lighthouse presented a target of such large radar cross section (approximately $100,000 \text{ m}^2$) against a relatively low clutter background that it was almost always displayed at a range of 20 nm and an altitude of 1000 feet. However, the lighthouse target was virtually indistinguishable from the targets presented by nearby ships and was often unidentifiable until visual contact was established.

6.0

6.1.2 Bearing Accuracy

Table 6.1 presents the Airborne Radar Approach System bearing accuracy data for both the Bendix RDR-1400A and the RCA Primus-50 radar systems. The Bendix Airborne Radar Approach System was determined to have a mean accuracy in bearing with which a beacon return can be displayed of .56 degrees and a one-sigma of 3.1 degrees at five (5) nautical miles from the beacon. The bearing accuracy data presented in Table 6.1 for the Bendix ARA System in the skin paint mode showed a mean value of 3.7 degrees and a one-sigma of 5.4 degrees at the five (5) nautical mile point. The RCA Primus-50 radar system was determined to have a mean accuracy in bearing with which a beacon return can be displayed of -.15 degrees and a one-sigma value of 2.0 degrees at five (5) nautical miles. Data is also presented in Table 6.1 at ten nautical miles for the three above mentioned areas. SC-133 requires an accuracy of $\pm 3^\circ$ at all ranges.

Distance To Beacon	Bendix RDR-1400A Beacon Mode		Bendix RDR-1400A Skin Paint Mode		RCA Primus-50 Beacon & Combined Mode	
	x(Deg)	±lσ (Deg)	x(Deg)	+1₀ (Deg)	x(Deg)	'lo (Deg)
5 nm	.56	3.1	3.7	5.4	- .15	2.0
10 nm	-3.3	6.1	3.3	3.8	.9 8	3.5

Table 6.1	Airborne	Radar	Approach	System	Bearing	Accuracy	Data
-----------	----------	-------	----------	--------	---------	----------	------

6.1.3 Display Readability

No specific tests were performed in this area. Qualitative observations showed that in direct sunlight the display was washed out, making it totally unreadable. In other forms of intense lighting the display readability was degraded but still could be resolved.

6.1.4 Display Resolution

Display resolution is inherently a system design characteristic, therefore, data obtained was to the same degree subjective in nature. As shown in Section 5.5 of Reference 1, a target width analysis of the Bendix Radar System showed a mean value of 13.2 degrees and a one-sigma

of 4.1 degrees. These values were obtained for all of the approaches flown during the single beacon testing. The RCA Primus-50 target area analysis showed at mean value of 7.7 nm² and a one-sigma of 2.74 nm² at the 25 nm range selector setting. At 2 nm range selector setting the target area analysis showed a mean value of .06 nm² and a one-sigma of .03 nm². It determined that even though the target displayed would have perferably been smaller, qualitatively it appears that the displayed size did not greatly affect the pilot's interpretation of the radar display for either system tested. It was, however, difficult for the pilot to identify laterally separated multiple beacons, while longitudinally separated multiple beacons could be distinguished at a 5000 foot spacing.

-

6.2 OPERATIONAL PERFORMANCE

This subsection will describe the operational performance of the ARA system using various track orientation techniques. The purpose of this subsection is to respond to the operational performance objectives stated in Section 3.0.

6.2.1 Beacon/Ground Clutter Discrimination

The combined beacon/ground mapping mode of operation offers two very important features; a) positive target identification, and b) obstacle clearance. The RCA Primus-50 was utilized for the combined mode testing phase. During the test period it was found that a major operational problem exists in the concept of the combined mode. The problem stems from the large beacon return that is displayed. The area covered by the return does not allow for any obstacle clearance within the immediate area of the intended target.

6.2.2 Offshore Target Discrimination

It was discovered during the offshure skin paint testing that while executing approaches to the lighthouse it was often times difficult to distinguish between ships and the lighthouse. On two occasions the wrong target was identified and tracked down to minimums. This lack of positive target identification could pose some very serious problems while executing an approach to an offshore site.

6.2.3 Cursor Interpretation

The use of an electronically generated cursor which obtains its input directly from HSI course selection proved very effective during the flight test program. Results showed that the cursor helped to decrease the Total System Cross Track (TSCT) and Flight Technical Error (FTE) quantities markedly. Results also indicated that the cursor aided approaches were conducted under conditions of decreased workload. The cursor technique provides the operator with orientation guidance relating to aircraft position and heading, desired course, and target location.

6.2.4 Performance In The Skin Paint And Skin Paint With Cursor Modes

The Bendix RDR-1400A radar system performed well in the skin paint and skin paint with cursor modes. As mentioned earlier the only serious problem encountered was that of positive target identification, which in turn creates a very high workload situation. The cursor aided approaches proved quite effective by reducing the overall TSCT and FTE values. The displayed target size was large, as was the case in previous testing, but the size did not present any operational problems.

6.2.5 Performance In The Single Beacon With Cursor Mode

Results in Section 5.3.3 indicated that the beacon with cursor tests proved very sucessful. The track orientation guidance provided by the generation of the cursor on the radar screen afforded the pilot a decreased workload situation and better track guidance. Plots presented in Section 5.3.3 show that the tendency to "home" to the station is eliminated with the use of the cursor. A comparison between the single beacon testing conducted earlier showed a large decrease in the TSCT and FTE quantities, while the ASE and ATE values continued to remain small. The ASE and ATE values reflect a system accuracy that is quite good, which also correlates to the qualitative assessment of the system.

6.2.6 Performance In The Multiple Beacon Mode

This mode of operation allowed the pilot to determine course error by creating a mental image of the track angle error utilizing two longitudinally separated beacons. Results indicate that this particular mode of operation does offer an increase in approach accuracy. Quantities indicated in Section 5.3.4 also show a decrease in the TSCL and ill quantities over

 ℓ_{1-4}

those obtained during the airport single beacon testing. It was discovered though that because the course error is not directly displayed on the radar screen the mental workload involved in flying the approaches was greater. Gain control also caused a major problem during this portion of testing. Later after the completion of the tests it was discovered by the Bendix Corporation that the STC was improperly adjusted. This maladjustment caused serious display problems close-in to the target because proper gain adjustment for one beacon meant that the other beacon was either splayed across the entire screen or lost completely. Because there is no means for beacon identification on occasions when one beacon was inoperative, positive intended target identification was impossible.

6.2.7 RCA Primus-50 Combined And Beacon-Only Mode Performance

The RCA Primus-50 Radar System performed well in the combined and beacon-only modes. Results presented in 5.3.5 show that the TSCT and FTE values are reasonably small. Operationally the combined mode performed well. The test results indicated only one major operational problem, which stems from the size of the displayed beacon. In the combined mode the Primus-50 Radar System flashes the beacon return on and off at one second intervals. This allows the operator positive target identification, but due to the size of the displayed beacon skin paint targets in the immediate area are blocked out by the return. The pilots also indicated during this phase of testing that the large displayed beacon made navigation difficult at times, namely in determining the azimuth of the beacon relative to zero azimuth. The beacon-only results indicated the same orders of magnitudes as the combined mode results for all of the error quantities computed.

CONCLUSIONS

The major conclusions from the operational flight test evaluation of the Airborne Radar Approach (ARA) system using various track orientation techniques and operational modes are summarized in this section. These conclusions are, by intent, qualitative in nature. The quantitative results from which these conclusions were reached are summarized in Section 6.0 and discussed in depth in Section 5.0. These conclusions are organized to represent a qualitative summary of the detailed evaluation objectives presented in Section 3.0.

- Skin Paint Mode
 - The Airborne Radar Approach Systems's ability to distinguish offshore targets such as lighthouses from ships and other surrounding surface objects is quite limited.
 - Since there is no positive intended target identification capability, obstacle clearance can not be assured at approach minimums.
- Combined Mode
 - The combined mode offers the unique capability of providing obstacle clearance while at the same time offering positive target identification.
 - The combined mode does offer one serious problem. Due to the large size of the displayed beacon on the radar system tested, skin paint targets in the inmediate area are blocked out.
- Cursor Interpretation
 - The cursor technique proved very effective in terms of track orientation in both the beacon and skin paint modes.
 - 2) The cursor technique affords the pilot the ability to fly a predetermined inbound course instead of "housed" to the station. this fact is evident from the large reduction in the 1200 and FIL errors for the cursor aided approaches.

7.0

/-!

- 3) The cursor offers a very quick and easy course orientation reference, therefore reducing the level of mental workload involved in flying the approach.
- 4) The cursor technique is a very low cost solution to track guidance that can be easily implemented in the present "state-of-the-art" radar systems.
- Multiple Beacon Mode
 - The multiple beacon technique proved effective in track orientation while utilizing various longitudinal beacon spacings.
 - The multiple beacon techniques also showed a decrease in the TSCT and FTE error quantities over those quantities calculated in the single beacon mode.
 - Since the beacon targets are not uniquely identified when one of the ground based beacons becomes inoperative, positive identification of the intended target is rendered impossible.
 - More advanced Sensitivity Time Constant (STC) circuitry is required so that multiple beacons are displayed at equal sizes with one gain setting.
 - 5) The mental workload involved in flying a multiple beacon approach is higher because the track angle error is not directly displayed on the radar screen.
- Beacon Mode vs. Skin Paint Mode Comparison Operationally both modes of operation have their own specific problems. The Airborne Radar Approach System is an operationally viable navigation aid to landing. The skin paint mode offers the ability to identify obstacles that surround the landing site, but offers no positive identification of the site itself. The beacon mode on the other hand offers positive site identification, but no obstacle discrimination.



- General Conclusions
 - The Airborne Radar Systems tested performed accurately at the airport and offshore sites in all of the various operational modes.
 - 2) The Airborne Radar Approach System range performance is adequate at minimum ranges, but maximum range performance at 1000 feet altitude should be improved as regards the requirements of RTCA SC-133.
 - For both systems tested the Airborne Radar Approach System bearing accuracy proved to be very acceptable.
 - 4) The display readability of the Airborne Radar Approach System is poor under high ambient light conditions.
 - 5) While large target widths and target sizes were usually encountered, this did not affect the operational performance or display interpretability of the Airborne Radar Approach System.
 - Unidentified beacon returns caused some operational problems.
 - The airborne system cross track error was found to be quite good.
 - 8) Data showed that without the use of some type of track orientation device the TSCT and FTE errors reflected the tendency to use ARA as a homing device rather than a cross track error "nulling" device.
 - 9) The Airborne System along track error was very good, again confirming the quality of the airborne system itself.
 - 10) The pilots utilized for the Airborne Radar Approach flight test performed very well. The workload involved in flying the approach necessitates two crew members, not one.

1

REFERENCES

- King, L.D. and Adams, R.J. "Airborne Radar Approach System Flight Test Experiment", Systems Control, Inc. (Vt), Champlain Technology Industries Division for the Federal Aviation Administration Systems Research And Development Service, FAA-RD-79-99, October 1979.
- 2. Anonymous, "United States Standard for Terminal Instrument Procedures (TERPS)", Federal Aviation Administration, July 1976.

Auto

BIBLIOGRAPHY

- Anonymous, "Operator's Manual Bendix RDR-1300/1400 Multi-Mode Radar", Bendix Avionics Division.
- King, L.D. and Ace, R.E., "Airborne Radar Approach Testing in CH53A Helicopter N-39 RCA Primus-50 Radar", Systems Control, Inc. (Vt), Champlain Technology Industries Division, June 1979.
- King, L.D. and Ace, R.E. "Airborne Radar Approach Testing in CH53A Helicopter N-39 Modified Bendix RDR-1400A ", Systems Control, Inc. (Vt), Champlain Technology Industries Pivision, June 1979.
- Perez, J., Draft Report, "Flight Evaluation of a Radar Cursor Technique as an Aid to Airborne Radar Approaches", National Aviation Facilities Experimental Center (NAFEC), 26 November 1979.
- Special Committee 133, "Minimum Operational Performance Standard for Airborne Radar Approach and Beacon Systems for Helicopters", Radio Technical Commission for Aeronautics (RTCA), RTCA Paper No. 161-79/SC133-80, 15 June, 1979.